

TABLE OF CONTENTS

Contents	Tab Nı	umber
Claim Deferral Form	······	1
End Claim Deferral Form	•••••	2
First Amended and Restated United States Mineral Products Company Asbestos Personal Injury Settlement Trust Distribution Procedures		3
Proof of Claim Form	•••••	4
Proof of Claim Instructions		5
Unacceptable Physicians List		6
United States Mineral Products Company Asbestos PI Settlement Trust Product Listing		7

United States Mineral Products Company Asbestos PI Settlement Trust Worksite Listing....... 8

Claim Deferral Form

United States Mineral Products Company Asbestos Personal Injury Settlement Trust

Claimant Name:		
Claim ID:		
Law Firm (If any):		
Section 6.8 of the Tr	ust Distribution Procedures a	llows claimants to defer the processing
of their claim for a p	eriod not to exceed three (3)	years without affecting the status of the
Claim for statute of l	imitations purposes. By elec	ting to defer processing, no further
action will take place	e with the claim until you not	ify CRMC that you are ready to resume
processing. Please n	ote that you may only reques	t a deferral ONE time per claim.
Please defer process	ing of the claim shown above	until you are notified otherwise.
1		•
Claimant,Perso	onal Representative or	Date
Law firm contact	Signature (check one)	
Print or type name if	not Claimant	
Time of type name if	not Ciannant	

End Claim Deferral Form

United States Mineral Products Company Asbestos Personal Injury Settlement Trust

Claimant Name:		
Claim ID:		
Law Firm (If any):		
Please end the deferm	ral of the claim shown above and	resume processing of it.
	sonal Representative or Signature (check one)	Date
Print or type name if	not Claimant	

FIRST AMENDED AND RESTATED UNITED STATES MINERAL PRODUCTS COMPANY ASBESTOS PERSONAL INJURY SETTLEMENT TRUST DISTRIBUTION PROCEDURES

These United States Mineral Products Company Asbestos Personal Injury Settlement Trust Distribution Procedures ("TDP") provide for resolving all Asbestos-Related Personal Injury Claims and Demands (as those terms are defined in the United States Mineral Products Company Plan of Reorganization ("Plan")) caused by exposure to asbestos-containing products for which United States Mineral Products Company ("USM") and its predecessors, successors, and assigns have legal responsibility (hereinafter for all purposes of this TDP referred to as "Trust Claims"), as provided in and required by the Plan and by the United States Mineral Products Company Asbestos Personal Injury Settlement Trust Agreement ("Trust Agreement"). The Plan and Trust Agreement establish the United States Mineral Products Company Asbestos Personal Injury Settlement Trust ("Trust"). The Trustees of the Trust ("Trustees") shall implement and administer this TDP in accordance with the Trust Agreement. Capitalized terms used herein and not otherwise defined shall have the meanings assigned to them in the Plan and the Trust Agreement.

SECTION I Introduction

- **1.1 Purpose.** This TDP has been adopted pursuant to the Trust Agreement. It is designed to provide fair and equitable treatment for all Trust Claims that may presently exist or may arise in the future in substantially the same manner.
- **1.2 Interpretation.** Nothing in this TDP shall be deemed to create a substantive right for any claimant.

SECTION II Overview

2.1 Trust Goals. The goal of the Trust is to treat all claimants equitably in accordance with the requirements of Section 524(g) of the Bankruptcy Code. This TDP furthers that goal by setting forth procedures for processing and paying claims generally on an impartial, first-in-first-out ("FIFO") basis, with the intention of paying all claimants over time as equivalent a share as possible of the value of their claims based on historical values for substantially similar claims in the tort system. To this end, the TDP establishes a schedule of three asbestos-related diseases ("Disease Levels"), with medical and exposure requirements ("Medical/Exposure Criteria") and liquidated values ("Scheduled Values"). The Disease Levels, Medical/Exposure Criteria, and Scheduled Values, which are set forth in Section 5.3, have all been selected and derived with the intention of achieving a fair allocation of the small amount of money that the Trust has available to distribute among claimants suffering from different disease processes in light of the best available information considering the settlement history of USM and the rights claimants would have in the tort system absent the bankruptcy.

2.2 Claims Liquidation Procedures. Trust Claims shall be processed based on their place in the FIFO Processing Queue to be established pursuant to Section 5.1(a) below. The Trust shall take all reasonable steps to resolve Trust Claims as efficiently and expeditiously as possible at each stage of claims processing and the alternative dispute resolution procedures. To this end, the Trust, in its sole discretion, may conduct settlement discussions with persons representing more than one claimant, provided that the claimants' respective positions in the FIFO Processing Queue are maintained and each claim is individually evaluated pursuant to the valuation factors set forth in Section 5.3(a)(3) below. The Trust shall liquidate all Trust Claims that meet the Medical/Exposure Criteria under the procedures described below. Claims that do not meet the Medical/Exposure Criteria for the relevant Disease Level may utilize the Trust's Alternate Review Process described in Section 5.3(b) below. In such a case, notwithstanding that the claim does not meet the presumptive Medical/Exposure Criteria for the relevant Disease Level, the Trust may offer the claimant an amount up to the Scheduled Value of that Disease Level if the Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system.

All unresolved disputes over a claimant's medical condition or exposure history shall, at the election of the claimant, be subject to the alternative dispute resolution procedures described in Section 5.10 below. Trust Claims that are the subject of a dispute with the Trust that cannot be resolved under the alternative dispute resolution procedures may enter the tort system as provided in Sections 5.11 and 7.6 below. However, if and when a claimant obtains a judgment in the tort system, the judgment will be payable only as provided in Section 7.7 below.

2.3 Application of the Payment Percentage. After the liquidated value of a Trust Claim is determined pursuant to the procedures set forth herein for Standard Review, Alternate Review, the alternative dispute resolution procedures described in Section 5.10 below, or litigation in the tort system, the claimant (other than the holder of a Settled Asbestos Claim) will be entitled to receive a pro-rata share of that value based on a Payment Percentage described in Section 4.2 below. The Payment Percentage may be adjusted upwards or downwards from time to time by the Trust, with the consent of the TAC and the Legal Representative, to reflect then-current estimates of the Trust's assets and its liabilities, as well as the then-estimated value of pending and future claims. If the Payment Percentage is increased over time, the Trustees shall have the discretion, in consultation with the TAC and the Legal Representative, to determine that claimants whose claims were liquidated and paid in prior periods under the TDP will receive additional payments. Because there is uncertainty in the prediction of both the number and severity of future claims, and the amount of the Trust's assets, no guarantee can be made of any Payment Percentage of a Trust Claim's liquidated value. A "Settled Asbestos Claim" is any Asbestos-Related Personal Injury Claim existing on or before the Petition Date that was scheduled by the Debtor in its schedules filed with the Bankruptcy Court on September 5, 2001 as being fixed and liquidated in the amount scheduled but which was then unpaid. The holder of a Settled Asbestos Claim shall be paid as provided herein.

2.4 Trust's Determination of the Maximum Annual Payment. The Trust shall estimate or model the amount of cash flow anticipated to be necessary over its entire life to ensure that funds will be available to treat all present and future claimants as similarly as possible. In each year, the Trust will be empowered to pay out all of the interest earned during the year, together with a portion of its principal, calculated so that the application of Trust funds over its life shall correspond with the needs created by the anticipated flow of claims (the "Maximum Annual Payment"), taking into account the Payment Percentage provisions set forth in Sections 2.3 above and 4.2 below and the amount needed to satisfy the Settled Asbestos Claims. Except for the payment of Settled Asbestos

Claims, the Trust's distributions to all claimants for any year shall not exceed the Maximum Annual Payment determined for that year.

2.5 Distribution of Maximum Annual Payment.

- **2.5(a)** By FIFO Payment Queue. If the Trustees determine that the Maximum Annual Payment is or will be equal to or greater than the amount needed to pay all claims in the FIFO Payment Queue, the Maximum Annual Payment shall be paid to claimants based on their place in the FIFO Payment Queue.
- **2.5(b) Priority to Disease Level II and III Claims.** If the Trustees determine that the Maximum Annual Payment is or will be less than the amount needed to pay all claims in the FIFO Payment Queue, the Maximum Annual Payment shall be first applied to pay holders of Disease Level II and III Claims based on their place in the FIFO Payment Queue. The balance of the Maximum Annual Payment shall be paid to holders of Disease Level I Claims based on their place in the FIFO Payment Queue.
- **2.5(c)** Carryover of Unpaid Claims. Claims that were in the FIFO Payment Queue but were not paid, whether because of the limitation of the Maximum Annual Payment or because of limited liquidity, shall be carried over to the next year retaining their place in the FIFO Payment Queue.
- **2.6 Indirect Asbestos-Related Personal Injury Claims.** An Indirect Asbestos-Related Personal Injury Claim, whether based on indemnity, contribution, or other theory of reimbursement, if any, shall be subject to the provisions of Section 5.6 below.

SECTION III TDP Administration

- **3.1 Trust Advisory Committee and Legal Representative.** Pursuant to the Plan and the Trust Agreement, the Trust Agreement and this TDP shall be administered by the Trustees in consultation with the TAC, which represents the interests of holders of present Asbestos-Related Personal Injury Claims, and the Legal Representative, who represents the interests of holders of PI Demands that will be asserted in the future. Pursuant to Section 8.1 below, the Trustees shall obtain the consent of the TAC and the Legal Representative to any amendments to these Procedures and on such other matters as are otherwise required below and in Section 2.2(f) of the Trust Agreement. The Trustees shall also consult with the TAC and the Legal Representative on such matters as are provided below and in Section 2.2(e) of the Trust Agreement. The initial members of the TAC and the initial Legal Representative are identified in the Trust Agreement.
- **3.2 Consent and Consultation Procedures.** In those circumstances in which consultation or consent is required, the Trustees will provide written notice to the TAC and the Legal Representative of the specific amendment or other action that is proposed. The Trustees will not implement such amendment nor take such action unless and until the parties have engaged in the Consultation Process described in Sections 5.9(a) and 6.7(a), or the Consent Process described in Sections 5.9(b) and 6.7(b), of the Trust Agreement, respectively.

SECTION IV

Payment Percentage; Periodic Estimates

- **4.1** Uncertainty of Amount of Payment to Holders of Trust Claims. There is inherent uncertainty regarding USM's total asbestos-related tort liabilities, as well as the total value of the assets available to the Trust to pay Trust Claims. Consequently, there is inherent uncertainty regarding the amounts that holders of Trust Claims will receive. To seek to ensure substantially equivalent treatment of all Trust Claims present and future, the Trustees must determine from time to time the percentage of the Scheduled Value that holders of Trust Claims, present and future, will be likely to receive, i.e., the "Payment Percentage" described in Sections 2.3 above and 4.2 below.
- **4.2 Computation of Payment Percentage.** The Trustees shall base their determination of the Payment Percentage on current estimates of the number, types, and values of present and future Trust Claims, the value of the Trust assets (less the amount needed to pay the Settled Asbestos Claims) then available to the Trust for their payment, all anticipated administrative and legal expenses, and any other material matters that are reasonably likely to affect the sufficiency of funds to pay a comparable percentage of full value to all holders of Trust Claims. When making these determinations, the Trustees shall exercise common sense and flexibly evaluate all relevant factors. The Payment Percentage shall be subject to change pursuant to the terms of this TDP and the Trust Agreement if the Trustees determine that an adjustment is required. No less frequently than once every two years, commencing with the Initial Claims Filing Date, the Trustees shall reconsider the then-applicable Payment Percentage to assure that it is based on accurate, current information and may, after such reconsideration, change the Payment Percentage if necessary, with the consent of the TAC and the Legal Representative. The Trustees shall also reconsider the then-applicable Payment Percentage at shorter intervals if they deem such reconsideration to be appropriate or if requested to do so by the TAC or the Legal Representative. "Initial Claims Filing Date" means the date that the Trust begins accepting proof of claim forms.
- **4.3 Proposed Change in the Payment Percentage.** No holder of a Trust Claim, except as provided in Section 5.3(a) for holders of Settled Asbestos Claims, shall receive a payment that exceeds the liquidated value of the claim times the Payment Percentage in effect at the time of payment. If a re-determination of the Payment Percentage has been proposed in writing by the Trustees to the TAC and the Legal Representative, but has not yet been adopted, the claimant shall receive the lower of the current Payment Percentage or the proposed Payment Percentage. If the proposed Payment Percentage was the lower amount but was not subsequently adopted, the claimant shall thereafter receive the difference between the lower proposed amount and the higher current amount. If the proposed Payment Percentage was the higher amount and was subsequently adopted, the claimant shall thereafter receive the difference between the lower current amount and the higher adopted amount. Claimants who received an offer of settlement from the Trust prior to a proposed redetermination of the Payment Percentage shall receive the higher of the current Payment Percentage or the proposed Payment Percentage, provided that they accept the offer.

SECTION V Resolution of Trust Claims

5.1 Ordering, Processing and Payment of Claims.

5.1(a) Ordering of Claims.

5.1(a)(1) Establishment of the FIFO Processing Queue. The Trust will order claims that are sufficiently complete to be reviewed for processing purposes on a FIFO basis except as otherwise provided herein (the "FIFO Processing Queue"). The Trust will pay Settled Asbestos Claims as soon as practical after the Trust has been organized and independently from the FIFO Processing Queue. For all claims filed on or before the Initial Claims Filing Date, a claimant's position in the FIFO Processing Queue shall be determined as of the earlier of (i) the date prior to the Petition Date that the specific claim was filed against USM in the tort system; (ii) the date after the Petition Date but before the Effective Date that the claim was filed against another defendant in the tort system; (iii) the date after the Petition Date but before the Effective Date that a proof of claim was filed against USM in USM's Chapter 11 case; (iv) the date a ballot was submitted in USM's Chapter 11 case for purposes of voting on the Plan in accordance with the voting procedures adopted by the Bankruptcy Court; or (v) the date a claim was filed with the Trust. Following the Initial Claims Filing Date, the claimant's position in the FIFO Processing Queue shall be determined by the date the claim was filed with the Trust. If any claims are filed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by the date of the diagnosis of the claimant's asbestos-related disease. If any claims are filed and diagnosed on the same date, the claimant's position in the FIFO Processing Queue shall be determined by the date of the claimant's birth, with older claimants given priority over younger claimants.

5.1(a)(2) Effect of Statutes of Limitations and Repose. To be eligible for a place in the FIFO Processing Queue, a claim must meet either (i) for claims first filed in the tort system against USM prior to the Petition Date, the applicable federal, state, and foreign statutes of limitations and repose that were in effect at the time of the filing of the claim in the tort system, or (ii) for claims that were not filed against USM in the tort system prior to the Petition Date, the applicable statute of limitations and repose in effect at the time of the filing with the Trust. However, the running of the relevant statute of limitations shall be tolled as of the earliest of (A) the date prior to the Petition Date that the specific claim was filed against USM in the tort system; (B) the date after the Petition Date but before the Effective Date that a proof of claim was filed against USM in USM's Chapter 11 case; (D) the date a ballot was submitted in USM's Chapter 11 case for purposes of voting on the Plan in accordance with the voting procedures adopted by the Bankruptcy Court; or (E) the date after the Initial Claims Filing Date that a proof of claim with the requisite supporting documentation was filed with the Trust.

If a Trust Claim of a claimant who was first diagnosed before the Petition Date meets any of the tolling provisions described in the preceding sentence and was not barred by the applicable statute of limitations at the time of the tolling event, the claim will be treated as timely filed if it is actually filed with the Trust within three years of the Initial Claims Filing Date. In addition, any claims that were first diagnosed after the Petition Date, irrespective of the application of any relevant statute of limitations or repose, may be filed with the Trust within three years after the date of diagnosis or within three years after the Initial Claims Filing Date, whichever occurs later.

The processing of any Trust Claim by the Trust may be deferred, however, at the election of the claimant pursuant to Section 6.8 below.

5.1(a)(3) Last Date to Submit Settled Claims to the Trust. All Settled Asbestos Claims, as defined in Section 2.3 above, must be filed no later than one year after the Initial Claims Filing Date or they shall be deemed barred and shall not be paid by the Trust as a Settled Asbestos Claim.

5.1(b) Processing of Claims. As a general practice, the Trust will review claims on a regular basis.

5.1(c) Payment of Claims. Trust Claims that have been liquidated by the review process as provided in Section 5.3(a)-(b) below, by alternative dispute resolution procedures as provided in Section 5.10 below, or by litigation in the tort system as provided in Section 5.11 below, shall be paid in FIFO order based on the date their liquidation became final (the "FIFO Payment Queue"), all such payments being subject to the applicable Payment Percentage and the Maximum Available Payment, except as otherwise provided herein. Where the claimant is deceased or incompetent, and the settlement and payment of his or her claim must be approved by a court of competent jurisdiction or through a probate process prior to acceptance of the payment by the claimant's representative, an offer made by the Trust on the claim shall remain open so long as proceedings before that court or in that probate process remain pending, provided that the Trust has been furnished with evidence that the payment offer has been submitted to such court or probate process for approval. If the offer is ultimately approved by the court or through the probate process and accepted by the claimant's representative, the Trust shall pay the claim in the amount so offered, multiplied by the Payment Percentage in effect at the time the offer was first made. If any claims are liquidated on the same date, the claimant's position in the FIFO Payment Queue shall be determined by the date of the diagnosis of the claimant's asbestos-related disease. If any claims are liquidated on the same date and the respective claimants' asbestos-related diseases were diagnosed on the same date, the position of those claimants in the FIFO Payment Queue shall be determined by the Trust based on the dates of the claimants' birth, with older claimants given priority over younger claimants.

5.2 [Omitted.]

5.3 Resolution of Unliquidated Trust Claims. The Trustees, with the consent of the TAC and the Legal Representative, may adopt a more detailed set of procedures for reviewing and liquidating all unliquidated Trust Claims. The proof of claim form shall require the claimant to assert his or her claim for the highest Disease Level for which the claim qualifies at the time of filing. Irrespective of the Disease Level alleged on the proof of claim form, all claims shall be deemed to be a claim for the highest Disease Level for which the claim qualifies at the time of filing. All lower Disease Levels for which the claim may also qualify at the time of filing or in the future shall be treated as subsumed in the higher Disease Level for both processing and payment purposes. Upon filing of a valid proof of claim form with the required supporting documentation, the claimant shall be placed in the FIFO Processing Queue in accordance with the ordering criteria described in Section 5.1(a) above.

5.3(a) Standard Review Process.

5.3(a)(1) In General. The Trust's Standard Review Process is designed to provide an expeditious, efficient, and inexpensive method for liquidating all claims that meet the

Medical/Exposure Criteria for the relevant Disease Level. Claims that meet the Medical/Exposure Criteria for a Disease Level shall be paid based on the Scheduled Value for such Disease Level set forth in Section 5.3(a)(3) below. Claimants holding claims that cannot be liquidated by Standard Review because they do not meet the Medical/Exposure Criteria for the relevant Disease Level may elect the Trust's Alternate Review Process set forth in Section 5.3(b) below.

5.3(a)(2) Claims Processing Under Standard Review. All claimants seeking liquidation of their claims pursuant to Standard Review shall file either an electronic proof of claim form or a paper-based proof of claim form. As a proof of claim form is reached in the FIFO Processing Queue, the Trust shall determine whether the claim described therein meets the Medical/Exposure Criteria for one of the three Disease Levels, and shall advise the claimant of its determination. After a Disease Level is determined, the claim shall be placed in the FIFO Payment Queue. The Trust, on receipt of an executed release, shall pay the claimant, subject to the limitations of the Maximum Available Payment, the Scheduled Value for the relevant Disease Level multiplied by the applicable Payment Percentage. Payment of a Claim shall act as and constitute a release of the Trust, except to the extent the Claimant has a second asbestos-related claim that qualifies for payment under Disease Level II or III.

5.3(a)(3) Disease Levels, Scheduled Values, and Medical/Exposure Criteria. The three Disease Levels covered by this TDP, the Medical/Exposure Criteria, and the Scheduled Values for each of the three Disease Levels are set forth below. With the consent of the TAC and the Legal Representative, the Trustees may add to, change or eliminate Disease Levels, Scheduled Values, or Medical/Exposure Criteria; develop subcategories of Disease Levels, Scheduled Values, or Medical/Exposure Criteria; or determine that a novel or exceptional asbestos-related personal injury claim is compensable even though it does not meet the Medical/Exposure Criteria for any of the then-current Disease Levels.

Settled Asbestos Claims shall be classified by Disease Level, and shall be paid a fixed distribution of \$3,750 for Disease Level III, \$2,200 for Disease Level II, and \$100 for Disease Level I.

Disease Level	Scheduled Value	Medical/Exposure Criteria
Mesothelioma (Level III)	\$10,000	(1) Diagnosis of malignant mesothelioma, and (2) credible evidence of USM Exposure (as defined in Section 5.7(b)(3) below).

Lung Cancer (Level II)	\$4,000	(1) Diagnosis of a primary lung cancer, plus evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease, (2) credible evidence of four months USM Exposure prior to the Exposure Deadline (as defined in Section 5.7(b)(5) below), (3) Significant Occupational Exposure to asbestos (as defined in Section 5.7(b)(2) below), and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the lung cancer in question.
Other Asbestos Disease (Level I)	\$500	(1) Diagnosis of a primary colo-rectal, laryngeal, esophageal, pharyngeal, or stomach cancer (plus evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease), asbestosis, or Bilateral Asbestos-Related Nonmalignant Disease, (2) credible evidence of four months USM Exposure prior to the Exposure Deadline (as defined in Section 5.7(b)(5) below), (3) Significant Occupational Exposure to asbestos (as defined in Section 5.7(b)(2) below), and (4) supporting medical documentation establishing asbestos exposure as a contributing factor in causing the other asbestos disease in question.

5.3(b) Alternate Review Process: Review of Medical/Exposure Criteria. The Trust's Alternate Review Process provides a claimant with an opportunity for alternate consideration and evaluation of a Trust Claim that fails to meet the presumptive Medical/Exposure Criteria for Disease Levels I–III. If a claim is submitted for Alternate Review, the Trust shall either deny the claim, or, if

¹ Evidence of "Bilateral Asbestos-Related Nonmalignant Disease" for purposes of meeting the criteria for establishing Disease Levels I and II means either (i) a chest X-ray read by a qualified B reader of 1/0 or higher on the ILO scale, or (ii) (x) a chest X-ray read by a qualified B reader or other Qualified Physician, (y) a CT scan read by a Qualified Physician, or (z) pathology, in each case showing either bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification. Evidence submitted to demonstrate (i) or (ii) above must be in the form of a written report stating the results (e.g., an ILO report, a written radiology report, or a pathology report). Solely for claims filed against USM or another asbestos defendant in the tort system prior to the Petition Date, if an ILO reading is not available, either (1) a chest X-ray or a CT scan read by a Qualified Physician, or (2) pathology, in each case showing bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification consistent with, or compatible with, a diagnosis of asbestos-related disease, shall be evidence of Bilateral Asbestos-Related Nonmalignant Disease for purposes of meeting the presumptive medical requirements of Disease Levels I and II. Pathological proof of asbestosis may be based on the pathological grading system for asbestosis described in the Special Issue of the Archives of Pathology and Laboratory Medicine, "Asbestos-associated Diseases," Vol. 106, No. 11, App. 3 (October 8, 1982). For all purposes of this TDP, a "Qualified Physician" is a physician who is board-certified (or in the case of Canadian claims or foreign claims, a physician who is certified or qualified under comparable medical standards or criteria of the jurisdiction in question) in one or more relevant specialized fields of medicine such as pulmonology, radiology, internal medicine, or occupational medicine; provided, however, subject to the provisions of Section 5.7(a), that the requirement for board certification in this provision shall not apply to otherwise-qualified physicians whose X-ray and/or CT scan readings are submitted for deceased holders of Trust Claims.

the Trust is satisfied that the claimant has presented a claim that would be cognizable and valid in the tort system, the Trust may pay the claimant based on the Scheduled Value for that Disease Level.

5.4 [Omitted.]

5.5 Secondary Exposure Claims. If a claimant alleges an asbestos-related disease resulting solely from exposure to an occupationally exposed person, such as a family member, the claimant may seek review of his or her claim pursuant to Section 5.3(a)(2) above. In such a case, the claimant must establish that the occupationally exposed person would have met the exposure requirements under this TDP that would have been applicable had that person filed a direct claim against the Trust. In addition, the claimant with secondary exposure must establish that he or she is suffering from one of the three Disease Levels described in Section 5.3(a)(3) above, that his or her own exposure to the occupationally exposed person occurred within the same time frame as the occupationally exposed person was exposed to asbestos products produced by USM, and that such secondary exposure was a cause of the claimed disease. All other liquidation and payment rights and limitations under this TDP shall be applicable to such claims.

5.6 Indirect Asbestos-Related Personal Injury Claims. An Indirect Asbestos-Related Personal Injury Claim (as defined in the Plan) may not be processed or paid by the Trust unless (a) such claim satisfied the requirements of the Bar Date established by the Bankruptcy Court for such claims, if applicable, and is not otherwise discharged by Section 502(e) of the Code, and (b) the holder of such claim (the "Indirect Claimant") establishes to the satisfaction of the Trustees that (i) the Indirect Claimant has paid in full the liability and obligations of the Trust to the individual claimant to whom the Trust would otherwise have had a liability or obligation under these Procedures (the "Direct Claimant"), (ii) the Direct Claimant and the Indirect Claimant have in writing forever released the Trust from any and all liability to the Direct Claimant, and (iii) the claim is not otherwise barred by a statute of limitations or repose or by other applicable law. No Indirect Claimant shall have any rights against the Trust superior to the rights of the related Direct Claimant against the Trust, including any rights with respect to the timing, amount, or manner of payment. In addition, no Indirect Claimant may be paid an amount that exceeds what the Indirect Claimant has actually paid the related Direct Claimant. The Trust shall not pay any Indirect Claimant unless and until the Indirect Claimant's aggregate liability for the Direct Claimant's claim has been fixed, liquidated, and paid fully by the Indirect Claimant by settlement (with an appropriate full release in favor of the Trust) or a Final Order (as defined in the Plan) provided that such claim is valid under the applicable state law. In any case where the Indirect Claimant has satisfied the claim of a Direct Claimant against the Trust under applicable law by way of a settlement, the Indirect Claimant shall obtain for the benefit of the Trust a release in form and substance satisfactory to the Trustees in favor of USM, the Trust, the Trustees, the TAC, the Legal Representative, the chapter 11 Trustee for USM, and Reorganized USM, and their respective officers, directors, members, employees, agents, attorneys, successors, trustees, administrators, and assignees. The Trustees may develop and approve a separate proof of claim form for Indirect Trust Claims. Indirect Trust Claims that have not been disallowed, discharged, or otherwise resolved by prior order of the Bankruptcy Court shall be processed in accordance with procedures to be developed and implemented by the Trustees, which procedures (a) shall determine the validity, allowability, and enforceability of such claims, and (b) shall otherwise provide the same liquidation and payment procedures and rights to the holders of such claims as the Trust would have afforded the holders of the underlying valid Trust Claims.

5.7 Evidentiary Requirements. A Claimant filing a proof of claim shall satisfy the evidentiary requirements set forth herein by execution and submission of the proof of claim form and

supporting documentation. No evidence shall be required to support a claim that otherwise qualifies as a Settled Asbestos Claim. The Trust requires claimants to submit the documentary evidence that satisfies the evidentiary criteria, and may reject a claim if such information and documents are not provided within such period as the Trustees may fix. Pursuant to Section 5.8 below, all claims are subject to audit by the Trust to determine that the claim is supported by the evidentiary standards set forth herein and any other requirements for acceptance and payment of a claim. Any Claimant may satisfy the medical evidence criteria by proof (or certification, if permitted) of acceptance of a claim for asbestos disease by such other asbestos trusts that the Trust may designate as acceptable alternative evidence of disease.

5.7(a) Medical Evidence.

5.7(a)(1) In General. All diagnoses of a Disease Level shall be accompanied by either (i) a statement by the physician providing the diagnosis that at least ten years have elapsed between the date of first exposure to asbestos or asbestos-containing products and the diagnosis, or (ii) a history of the claimant's exposure sufficient to establish a ten-year latency period. A finding by a physician after the Effective Date that a claimant's disease is "consistent with" or "compatible with" asbestosis shall not alone be treated by the Trust as such a diagnosis.

5.7(a)(1)(A) Disease Level I. A diagnosis of a nonmalignant asbestos-related disease or of a primary colo-rectal, laryngeal, esophageal, pharyngeal, or stomach cancer shall be based (i) in the case of a claimant who was living at the time the claim was filed, upon a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease, and evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in footnote 1 above), and (ii) in the case of a claimant who was deceased at the time the claim was filed, upon (a) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease, (b) a physical examination of the claimant by another doctor whose physical examination and findings are reliable, (c) a diagnosis of the nonmalignant or malignant asbestos-related disease by a board-certified pathologist or by a pathology report prepared at or on behalf of a hospital accredited by the Joint Commission on Accreditation of Healthcare Organizations ("JCAHO"), or (d) evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in footnote 1 above).

5.7(a)(1)(B) Disease Levels II–III. A diagnosis of an asbestos-related Lung Cancer (Disease Level II) or Mesothelioma (Disease Level III) shall be based upon either (i) a physical examination of the claimant by the physician providing the diagnosis of the asbestos-related disease, or (ii) a diagnosis of such a malignant Disease Level by a board-certified pathologist or by a pathology report prepared at or on behalf of a hospital accredited by JCAHO.

5.7(a)(2) Credibility of Medical Evidence. Before making any payment to a claimant, the Trust must have reasonable confidence that the medical evidence provided in support of the claim is credible and consistent with recognized medical standards. The Trust may require the submission of X-rays, CT scans, detailed results of pulmonary function tests, laboratory tests, tissue samples, results of medical examination, or reviews of other medical evidence, and may require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods, and procedures to assure that such evidence is reliable. Medical evidence that is (i) of a kind shown to have been received in evidence by a state or federal judge at trial, (ii) consistent with evidence submitted to USM to settle for payment similar disease cases prior to USM's bankruptcy, or (iii) a diagnosis by a physician shown to have previously qualified as a medical expert with respect to

the asbestos-related disease in question before a state or federal judge, is presumptively reliable, although the Trust may seek to rebut the presumption. In addition, claimants who otherwise meet the requirements of this TDP for payment of a Trust Claim shall be paid irrespective of the results in any litigation at any time between the claimant and any other defendant in the tort system. However, any relevant evidence submitted in a proceeding in the tort system involving another defendant, other than any findings of fact, a verdict, or a judgment, may be introduced by either the claimant or the Trust in any Alternate Review Proceeding conducted pursuant to Section 5.3(b).

5.7(a)(3) Reliance by Trust on Finding of Another Asbestos Trust. The Trustees may review the governing documents of another asbestos trust or claims facility and, with the consent of the TAC and the Legal Representative, determine to accept the disease level classifications as found by such other asbestos trust or claims facility in lieu of the medical evidence claimants are required to submit under this TDP.

5.7(a)(4) Exemption from Filing Medical Evidence. The Trustees, with the consent of the TAC and the Legal Representative, may exempt claimants from the obligation to submit medical evidence or certain types of medical evidence.

5.7(b) Exposure Evidence.

5.7(b)(1) In General. As set forth in Section 5.3(a)(3) above, to qualify for any Disease Level, the claimant must provide evidence of USM Exposure (as defined in Section 5.7(b)(3) below). Claims based on conspiracy theories that involve no USM Exposure are not compensable under this TDP. To meet the exposure requirements of Standard Review set forth in Section 5.3(a)(3) above, the claimant must show (i) for all Disease Levels, USM Exposure, as defined in Section 5.7(b)(3) below, prior to the Exposure Deadline, as defined in Section 5.7(b)(5) below; and (ii) for Disease Levels I and II, four months USM Exposure prior to the Exposure Deadline, plus Significant Occupational Exposure to asbestos as defined in Section 5.7(b)(2) below. If the claimant cannot meet the relevant exposure requirements for a Disease Level, the claimant may seek Alternate Review of his or her claim pursuant to Section 5.3(b) above.

5.7(b)(2) Significant Occupational Exposure. "Significant Occupational Exposure" means employment for a cumulative period of at least five years, with a minimum of two years prior to the Exposure Deadline, as defined in Section 5.7(b)(5) below, in an industry and occupation in which the claimant (a) handled raw asbestos fibers on a regular basis, (b) fabricated asbestos-containing products so that the claimant in the fabrication process was exposed on a regular basis to raw asbestos fibers, (c) altered, repaired, or otherwise worked with an asbestos-containing product such that the claimant was exposed on a regular basis to asbestos fibers, or (d) worked on a regular basis in close proximity to workers engaged in the activities described in (a), (b), and/or (c).

5.7(b)(3) USM Exposure.

(A) Standard. A claimant has USM Exposure if, prior to the Exposure Deadline as defined in Section 5.7(b)(5) below, the claimant was exposed to asbestos or asbestos-containing products supplied, specified, manufactured, produced, distributed, sold, fabricated, installed, released, maintained, repaired, replaced, removed, or handled by USM and/or any entity for which USM is responsible.

(B) Qualified Worksites and Industries and Occupations. A claimant establishes USM Exposure by demonstrating that before the Exposure Deadline

(i) the claimant worked at a site that either (I) the claimant establishes to the Trust's satisfaction is a qualified USM worksite or (II) the Trust designates as a qualified USM worksite, and

(ii) the claimant worked at the qualified USM worksite either (I) during the installation, rip-out, or repair of USM product or (II) during or after the time the USM product was first installed at the qualified worksite, provided the claimant then was working in an occupation and industry specified by the Trust.

(C) **Designation.** The Trust shall designate sites that are qualified USM worksites and provide a list of those qualified USM worksites to claimants. The Trust, in consultation with the TAC and Legal Representative, shall also designate occupations and industries in which claimants were likely to have been exposed to USM product. A claimant who claims USM exposure at a site that is not on the Trust's list of qualified USM worksites, or who indicates employment in an occupation or industry that is not specified on the Trust's list of occupation and industry, must provide additional exposure evidence.

(D) Meaningful and Credible Evidence. USM exposure must be established by meaningful and credible evidence. Evidence may consist of an affidavit or sworn statement of the claimant; an affidavit or sworn statement of a co-worker or, in the case of a deceased claimant, an affidavit or sworn statement of a family member (provided that the Trust finds such evidence reasonably reliable); invoices, employment, construction, or similar records; interrogatory answers, sworn work histories, or depositions; or other credible evidence. The Trust may also require submission of other or additional evidence of exposure when it deems such to be necessary.

5.7(b)(4) Exemption from Filing Exposure Evidence. The Trustees, with the consent of the TAC and the Legal Representative, may exempt claimants from the obligation to submit exposure evidence or certain types of exposure evidence.

5.7(b)(5) Exposure Deadline. The Exposure Deadline is December 31, 1982.

5.8 Claims Audit Program. The Trust, with the consent of the TAC and the Legal Representative, may develop methods for auditing the reliability of medical evidence, including additional reading of X-rays, CT scans, and verification of pulmonary function tests, as well as the reliability of evidence of exposure to asbestos, including USM Exposure prior to December 31, 1982. In the event that the Trust reasonably determines that any individual or entity has engaged in a pattern or practice of providing unreliable medical evidence to the Trust, it may decline to accept additional evidence from such provider in the future. Further, in the event that an audit reveals that fraudulent information has been provided to the Trust, the Trust may penalize any claimant or claimant's attorney by disallowing the Trust Claim or by other means, including, but not limited to, requiring the source of the fraudulent information to pay the costs associated with the audit and any future audit or audits, reordering the priority of payment of all affected claimants' Trust Claims, raising the level of scrutiny of additional information submitted from the same source or sources, refusing to accept additional evidence from the same source or sources, seeking the prosecution of the claimant or claimant's attorney for presenting a fraudulent claim in violation of 18 U.S.C. § 152, and seeking sanctions from the Bankruptcy Court.

5.9 Second Disease (Malignancy) Claims.

5.9(a) Second Disease (Malignancy) Trust Claims. The holder of a Disease Level I Trust Claim that is not a Settled Asbestos Claim may assert a new Trust Claim against the Trust for lung cancer or mesothelioma (Disease Levels II–III) that is first diagnosed subsequent to the filing with the Trust of the Level I claim. Any payments to which such claimant may be entitled with respect to a Disease Level II or III claim shall not be reduced by the amount paid for the Disease Level I claim.

5.9(b) Second Disease (Malignancy) Settled Asbestos Claims. The holder of a Trust Claim that is a Settled Asbestos Claim under Section 2.3 above in the category of Disease Level I or II may assert a new Trust Claim against the Trust for lung cancer or mesothelioma (Disease Levels II–III) that is a higher level Disease Claim than that set out on the Debtors' Schedules filed with the Bankruptcy Court on September 5, 2001, pursuant to which the claim was categorized as a Settled Asbestos Claim. Any payments to which such claimant may be entitled with respect to a Disease Level II or III claim shall not be reduced by the amount paid for the Settled Asbestos Claim. Any additional claim filed by such claimant shall in all respects be governed by the provisions of the TDP.

Notwithstanding Section 5.1(a)(3) above, the holder of a Trust Claim that is a Settled Asbestos Claim under Section 2.3 above may decline to file a Settled Asbestos Claim and may instead file an asbestos claim under the procedures applicable to all non-Settled Asbestos Claims.

5.10 Alternative Dispute Resolution Procedures.

5.10(a) Establishment of ADR Procedures.

5.10(a)(1) Arbitration. The Trust, with the consent of the TAC and the Legal Representative, shall adopt binding and nonbinding arbitration procedures and rules for resolving disputes concerning whether the Trust's outright rejection or denial of a claim was proper, or whether the claimant's medical condition or exposure history meets the requirements of this TDP for purposes of categorizing a claim in Disease Levels I–III. In all arbitrations, the arbitrator shall consider the same medical and exposure evidentiary requirements that are set forth in Section 5.7 above. With respect to all claims eligible for arbitration, the claimant, but not the Trust, may elect either binding or nonbinding arbitration.

5.10(a)(2) Trustee Conference. The Trust may establish a Trustee conference process and, with the consent of the TAC and the Legal Representative, adopt procedures and rules for resolving disputes concerning whether the Trust's outright rejection or denial of a claim was proper, or whether the claimant's medical condition or exposure history meets the requirements of this TDP for purposes of categorizing a claim in Disease Levels I–III. The Trustee conference process may supplement the binding and nonbinding arbitration procedures adopted under Section 5.10(a)(1) above.

5.10(b) Claims Eligible for Alternative Dispute Resolution. In order to be eligible for alternative dispute resolution, the claimant must first complete the Standard or Alternate Review Process as well as mediation (if the Trustees adopt a mediation process) with respect to the disputed issue.

- **5.10(c)** Limitations on and Payment of Arbitration Awards. The arbitrator shall not return an award in excess of the Scheduled Value for the appropriate Disease Level as set forth in Section 5.3(a)(3) above. A claimant who submits to arbitration and who accepts the arbitral award will receive payment in the same manner as one who accepts the Trust's original classification of the claim.
- **5.11 Litigation.** Claimants who elect nonbinding arbitration and then reject their arbitral awards retain the right to institute a lawsuit in the tort system against the Trust pursuant to Section 7.6 below. A claimant shall be eligible for payment of a judgment for monetary damages obtained in the tort system from the Trust's available cash only as provided in Section 7.7 below.

SECTION VI Claims Materials

6.1 Release of Information from Other Trusts. This TDP has been prepared with the goal of simplifying the filing of claims, reducing paperwork by claimants, and reducing the cost of reviewing claims in an effort to maximize distribution to claimants. Accordingly, with the consent of the claimant or his or her attorney, the Trust may obtain information regarding such claimant from any other trust created in connection with a case under the Bankruptcy Code that is administering asbestos claims, and such information may be used to support claims filed with this Trust.

6.2 [Omitted.]

- **6.3** [Omitted.]
- **6.4 Claims Materials.** The Trust shall prepare suitable and efficient claims materials ("Claims Materials") for all Trust Claims that may be filed by claimants or their counsel, and shall provide such Claims Materials upon request.
- **6.5** Claim Submitted Must Be for Highest Claim Existing; Compliance with Rule 11(b). The proof of claim form to be submitted to the Trust shall require the claimant to assert the highest Disease Level for which the claim qualifies at the time of filing. The proof of claim form shall also include a certification by the claimant or his or her attorney sufficient to meet the requirements of Rule 11(b) of the Federal Rules of Civil Procedure.
- **6.6 Content of Claims Materials.** The Claims Materials shall include a copy of this TDP, such instructions as the Trustees shall approve, and a detailed proof of claim form. If feasible, the forms used by the Trust to obtain claims information shall be the same or substantially similar to those used by other asbestos claims resolution organizations. Instead of collecting some or all of the claims information from a claimant or the claimant's attorney, the Trust may, with the consent of the claimant, obtain such information from electronic databases maintained by other asbestos claims resolution organizations.

6.7 [Omitted.]

- **6.8 Withdrawal or Deferral of Claims.** A claimant may withdraw a Trust Claim at any time upon written notice to the Trust, and file another such claim subsequently, without affecting the status of the claim for statute of limitations purposes, but any such claim filed after withdrawal shall be given a place in the FIFO Processing Queue based on the date of such subsequent filing. A claimant may also request that the processing of his or her Trust Claim by the Trust be deferred for a period not to exceed three years without affecting the status of the claim for statute of limitations purposes, in which case the claimant shall retain his or her original place in the FIFO Processing Queue. Except for Trust Claims held by representatives of deceased or incompetent claimants for which court or probate approval of the Trust's offer of payment is required, or a Trust Claim for which deferral status has been granted, a claim will be deemed to have been withdrawn if the claimant neither accepts, rejects, nor initiates arbitration within six months of the Trust's offer of payment or rejection of the claim. Upon written request and good cause, the Trust may extend this period or the deferral period for an additional six months.
- **6.9 Filing Fees.** The Trustees shall have the discretion to determine, with the consent of the TAC and the Legal Representative, whether a filing fee should be required for any Trust claims.
- **6.10 Confidentiality of Claimants' Submissions.** All submissions to the Trust by a holder of a Trust Claim, or a proof of claim form and materials related thereto, shall be treated (a) as made in the course of settlement discussions between the claimant and the Trust and (b) as intended by the parties, to be confidential and protected by all applicable state and federal privileges, including, but not limited to, those directly applicable to settlement discussions. The Trust will preserve the confidentiality of such claimant submissions, and shall disclose the contents thereof, only with the permission of the claimant, to another trust established for the benefit of asbestos personal injury claimants pursuant to section 524(g) and/or section 105 of the Bankruptcy Code or other applicable law, to such other persons as authorized by the claimant, or in response to a valid subpoena for such materials. Furthermore, the Trust shall provide counsel for the claimant a copy of any such subpoena immediately upon being served.

SECTION VII

General Guidelines for Liquidating and Paying Claims

- **7.1 Showing Required.** To establish a valid Trust Claim, a claimant must meet the requirements set forth in this TDP. The Trust may require the submission of X-rays, CT scans, laboratory tests, medical examinations or reviews, other medical evidence, or any other evidence to support or verify the claim, and may further require that medical evidence submitted comply with recognized medical standards regarding equipment, testing methods, and procedures to assure that such evidence is reliable.
- **7.2 Costs Considered.** Notwithstanding any provisions of this TDP to the contrary, the Trustees shall always give appropriate consideration to the cost of investigating and uncovering invalid Trust Claims so that the payment of valid Trust Claims is not impaired by such processes with respect to issues related to the validity of the medical evidence supporting a Trust Claim. The Trustees shall also have the latitude to make judgments regarding the amount of transaction costs to be expended by the Trust so that valid Trust Claims are not unduly impaired by the costs of additional investigation. Nothing herein shall prevent the Trustees, however, from contesting the validity of a Trust Claim in appropriate circumstances, whatever the cost, or from declining to accept

medical evidence from sources that the Trustees have determined to be unreliable pursuant to the claims audit program described in Section 5.8 above.

7.3 Discretion to Vary Order and Amounts of Payments in Event of Limited Liquidity. The Trustees shall adopt a target first distribution date that shall not be more than 24 months after the Initial Claims Filing Date. Thereafter, the Trustees shall proceed promptly to liquidate valid Trust Claims, and shall make payments to holders of such claims in accordance with this TDP at least once a year, while maintaining sufficient resources to pay future valid claims in substantially the same manner. Because the Trust's income over time remains uncertain and decisions about payments must be based on estimates that cannot be done precisely, they may have to be revised in light of experiences over time, and there can be no guarantee of any specific level of payment to claimants. The Trustees shall use their best efforts, however, to treat similar claims in substantially the same manner, consistent with their duties as Trustees, the purposes of the Trust, the FIFO Queue procedures, and the practical limitations imposed by the inability to predict the future with precision. In the event that the Trust faces temporary periods of limited liquidity, the Trustees may, in addition to implementing Section 2.5(b), and with the consent of the TAC and the Legal Representative, suspend the normal order of payment, and temporarily limit or suspend payments altogether.

7.4 Punitive Damages. In determining the value of any liquidated or unliquidated Trust Claim, punitive or exemplary damages (i.e., damages other than compensatory damages) shall not be considered or allowed, notwithstanding their availability in the tort system.

7.5 [Omitted.]

7.6 Suits in the Tort System. If the holder of a disputed claim disagrees with the Trust's determination regarding the Disease Level of the claim or the claimant's satisfaction of the exposure requirements of Section 5.3(a)(3), and if the claimant has first submitted the claim to nonbinding arbitration, the claimant may file a lawsuit in the Claimant's Jurisdiction, as defined below. Any such lawsuit must be filed by the claimant in his or her own right and name, and not as a member or representative of a class, and no such lawsuit may be consolidated with any other lawsuit. All defenses (including, with respect to the Trust, all defenses which could have been asserted by USM) shall be available to both sides at trial; the Trust may, however, waive any defense and/or concede any issue of fact or law. If the claimant was alive at the time that the initial pre-petition complaint was filed or on the date the proof of claim was filed with the Trust, the case will be treated as a personal injury case with all personal injury damages to be considered even if the claimant has died during the pendency of the claim.

For these purposes, the "Claimant's Jurisdiction" is the jurisdiction in which the claim was filed (if at all) against USM in the tort system prior to the Petition Date. If the claim was not filed against USM in the tort system prior to the Petition Date, the claimant may elect as the Claimant's Jurisdiction either (i) the jurisdiction in which the claimant resided at the time of diagnosis or when the claim was filed with the Trust, or (ii) a jurisdiction in which the claimant was exposed to an asbestos-containing product for which USM has legal responsibility.

7.7 Payment of Judgments for Money Damages. If and when a claimant obtains a judgment in the tort system, the claim shall be placed in the FIFO Payment Queue based on the date on which the judgment became final. Thereafter, the claimant shall receive from the Trust an initial payment (subject to the applicable Payment Percentage and the Maximum Available Payment provisions set forth above) of an amount equal to one hundred percent (100%) of the greater of (i) the

Trust's offer of payment to the claimant, or (ii) the award that the claimant declined in nonbinding arbitration. After the payment of all other Trust Claims, to the extent funds are available, the claimant shall receive the balance of the judgment, if any, in five, equal yearly installments that are subject to the applicable Payment Percentage. The total amounts paid with respect to such claims shall not exceed two hundred percent (200%) of the Scheduled Value for the applicable Disease Level.

- **7.8 Releases.** The Trustees shall have the discretion to determine the form and substance of the releases to be provided to the Trust in order to maximize recovery for claimants against other tortfeasors without increasing the risk or amount of claims for indemnification or contribution from the Trust. As a condition to making any payment to a claimant, the Trust shall obtain a general, partial, or limited release as appropriate in accordance with the applicable state or other law. If allowed by state law, the endorsing of a check or draft for payment by or on behalf of a claimant shall constitute such a release.
- **7.9 Third-Party Services.** Nothing in this TDP shall preclude the Trust from contracting with another asbestos claims resolution organization to provide services to the Trust so long as decisions about the categorization and liquidated value of Trust Claims are based on the relevant provisions of this TDP, including the Disease Levels, Scheduled Values, and Medical/Exposure Criteria set forth above.

7.10 Trust Disclosure of Information.

7.10(a) General. Periodically, but not less often than once a year, the Trust shall make available to the TAC and the Legal Representative a report that provides the number of claims by Disease Level that have been resolved by both the Standard and Alternate Review Processes, by Trustee Conference or alternative dispute resolution, and by litigation in the tort system.

7.10(b) Disclosure to TAC and Legal Representative. Subject to entry into an appropriate and reasonable confidentiality agreement, the Trustees, in connection with regularly scheduled Trust meetings or otherwise, shall report to and provide information to the TAC and Legal Representative relating to the financial condition of the Trust, the number of claims filed, the number of claims paid, and the number of unpaid claims, as well as accounting or statistical information concerning the operation of USM and the possible sale of USM.

SECTION VIII Miscellaneous

- **8.1 Amendments.** Except as otherwise provided herein, the Trustees may amend, modify, delete, or add to any provisions of this TDP (including, without limitation, amendments to conform this TDP to advances in scientific or medical knowledge or other changes in circumstances), provided they first obtain the consent of the TAC and the Legal Representative pursuant to the Consent Process set forth in Sections 5.7(b) and 6.6(b) of the Trust Agreement.
- **8.2** Severability. Should any provision contained in this TDP be determined to be unenforceable, such determination shall in no way limit or affect the enforceability and operative effect of any and all other provisions of this TDP. Should any provision contained in this TDP be determined to be inconsistent with or contrary to USM obligations to any insurance company

providing insurance coverage to USM in respect of claims for personal injury based on exposure to asbestos-containing products manufactured or produced by USM, the Trust, with the consent of the TAC and the Legal Representative, may amend this TDP and/or the Trust Agreement to make the provisions of either or both documents consistent with the duties and obligations of USM to said insurance company.

8.3 Governing Law. Administration of this TDP shall be governed by, and construed in accordance with, the laws of the State of New York. The law governing the liquidation of Trust Claims in the case of Standard and Alternate Review, alternative dispute resolution, or litigation in the tort system shall be the law of the Claimant's Jurisdiction as described in Section 7.6 above.

UNITED STATES MINERAL PRODUCTS COMPANY ASBESTOS PERSONAL INJURY SETTLEMENT TRUST

Proof of Claim Form for Asbestos-Related Personal Injury Claims

Submit completed claims to:

Claims Resolution Management Corporation P.O. Box 12003 Falls Church, VA 22042-0683

(703) 204-9300 (800) 536-2722

For information on filing electronically, please call Claims Resolution Management Corporation ("CRMC") Customer Service at the above telephone numbers, or e-mail us at inquiry@claimsres.com.

Please review the claim filing instructions prior to completing this proof of claim form. By checking this box, the Injured Party or the person filing the claim on his/her behalf (such as the Injured Party's personal representative or counsel) consents to the United States Mineral Products Company Asbestos Personal Injury Settlement Trust ("USM Trust") obtaining prior settlement information from the Manville Personal Settlement Injury Trust ("Manville Trust") for purposes of verifying his/her claim filed with the USM Trust.

6/1/2010 Form No: POC-USM Version 1

PART A: INJURED PARTY

INJURED PARTY INFORMATION:

Name:		
First	Middle	Last
Social Security Number:		or International ID:(Required for Foreign Claims)
Date of Birth: (MM/DD/YYYY)		
If you are relying on a prior sett	lement with th	ne Manville Trust, please provide the Claim I
Manville Trust Claim ID: _		
nformation. Address:		
Daytime Telephone: (_)	
E-mail:		
If the Injured Party is <u>deceased</u> ,	provide the fo	ollowing information.
Injured Party's Last State of	of Residence:	
Date of Death:		

Name:First	Middle	Last	
Address:			
Daytime Telephone: ()			
E-mail:			

Provide the information requested below if someone other than the Injured Party is filing the claim on the Injured Party's behalf (such as the Injured Party's personal representative).

PART B: LAW FIRM INFORMATION

CLAIMANT'S COUNSEL INFORMATION:

Provide the information	requested below if the Claimant is represented by legal counsel.
CRMC Law Firm	Code, if known:
Attorney Code:	(if previously supplied by CRMC)
-OR-	
Attorney Name: _	
Law Firm:	
Telephone: (Fax: ()
E-mail:	
Contact Name:	
Tax ID Number: _	Website:
Mailing Address for Claim-Related Correspondence: (if different from	Street Address
above)	City, State (Province), Zip Code (Postal Code) Country
("USM") on behalf of the	RMATION: d lawsuit been filed against United States Mineral Products Company he Injured Party (check applicable box below)? iest date filed, and State or Country in which filed.)
Date (MM/DD/YYYY) Name of Court	Y) State Country

PART C: ASBESTOS-RELATED INJURY

DIAGNOSED INJURY:

for w	k the box below for the injury being claimed a hich medical documentation is attached (referedures ("TDP")). Also provide the date of dia	to Section 5.7(a) of the Trust Distribution
	Mesothelioma (Disease Level III)	Diagnosis Date:(MM/DD/YYYY)
	Diagnosis of a malignant mesothelioma.	
	Lung Cancer* (Disease Level II)	Diagnosis Date:(MM/DD/YYYY)
	Diagnosis of a primary lung cancer, plus evi Related Nonmalignant Disease, and supporti asbestos exposure as a contributing factor in	ng medical documentation establishing
	Other Asbestos Disease* (Disease Level I)	Diagnosis Date:(MM/DD/YYYY)
	(plus evidence of an underlying Bilateral As asbestosis, or Bilateral Asbestos-Related No	nmalignant Disease, and supporting medical re as a contributing factor in causing the other
	☐ Pleural Disease ☐ Interstitial Lung Disease ☐ Colorectal Cancer ☐ Esophageal Cancer	Laryngeal Cancer Pharyngeal Cancer Stomach Cancer

^{*} Requires a showing of Significant Occupational Exposure, which is defined in the USM Trust Distribution Procedures (the "TDP") to mean employment for a cumulative period of at least five years, with a minimum of two years prior to December 31, 1982), in an industry and an occupation in which the claimant (a) handled raw asbestos fibers on a regular basis, (b) fabricated asbestos-containing products so that the claimant in the fabrication process was exposed on a regular basis to raw asbestos fibers, (c) altered, repaired, or otherwise worked with an asbestos-containing product such that the claimant was exposed on a regular basis to asbestos fibers, or (d) was employed in an industry and occupation such that the claimant worked on a regular basis in close proximity to workers engaged in the activities described in (a), (b), and/or (c).

PART D: EXPOSURE

To qualify for any Disease Level under the TDP, the claimant must demonstrate meaningful and credible exposure prior to December 31, 1982 to asbestos or asbestos-containing products supplied, specified, manufactured, produced, distributed, sold, fabricated, installed, released, maintained, repaired, replaced, removed, or handled by USM and/or any entity for which USM is responsible.

The completion of **Section 1** of this **Part D** is required for all claims.

The completion of **Section 2** of this **Part D** is required only for claims alleging Lung Cancer (Disease Level II) or Other Asbestos Disease (Disease Level I).

The completion of **Section 3** of this **Part D** is required only for claims alleging an asbestos-related disease resulting from secondary exposure to an occupationally exposed person.

Section 1: Occupational Exposure to USM Asbestos Products

Provide all periods of occupational exposure to USM asbestos products.
Site Code: (A list of previously qualified Site Codes is available on the CRMC website under the Documents tab for the U.S. Mineral Trust; use the code "NQ" to indicate an exposure site that is not on this list.)
If the site/plant is not on the approved list, please complete the following:
Name of Site/Plant of Exposure:
City:
State/Province:
Country:
Industry in which Exposure Occurred: (Industry Codes are provided on page 14 below.)
If Code 21 – Other, please describe:
Date Exposure Began: Month Year
Date Exposure Ended: Month Year

Occupation Code (if applicable):provided on pages 12-13 below.)	(Occupation Codes are
If no Occupation Code exists, provide the occupat asbestos-containing products.	ion at the time of exposure to USM
Occupation:	
Describe the nature or circumstances of exposure	to USM asbestos-containing product(s):

If needed, please photocopy this Section 1 to indicate each company site, industry, and occupation upon which you rely to meet the USM exposure requirements of the TDP.

Section 2: Significant Occupational Exposure (Lung Cancer or Other Asbestos Claims)

Provide all periods of the occupationally exposed person's asbestos exposure sufficient to meet the five-year requirement regardless of whether USM products were involved. Occupation Codes are provided on pages 12-13 below.

If the site/plant is not on the approved list, please complete the following:

Nan	me of Si	te/Plant of	Exposu	re:					
City	y:								
Stat	te/Provi	nce:							
Cou	untry:								
Industry page 14 be		ch Exposu	re Occu	rred:	(In	dustry	Codes aı	re provided or	1
If Code 2	21 – Oth	er, please o	lescribe:						
From:	Month	Year	То:	Month	Year	_			
Occupation	ion Cod	e:							
		upation Co escribe his/			ne occupati	onally	exposed	person's wor	k
Occupati	ion:								
(Continue	ed on ne	ext page.)							

Check the cate	egory that best describes the exposure of the Injured Party at this site:
(a)	Handled raw asbestos fibers on a regular basis.
(b)	Fabricated asbestos-containing products so that he/she was exposed on a regular basis to raw asbestos fibers.
(c)	Altered, repaired, or otherwise worked with an asbestos-containing product such that he/she was exposed on a regular basis to asbestos fibers.
(d)	Was employed in an industry and occupation such that he/she worked on a regular basis in close proximity to workers engaged in the activities described in (a), (b), and/or (c).
(e)	None of the above. Please provide a description of how the Injured Party was exposed to asbestos:

If needed, please photocopy this Section 2 to indicate each company site, industry, and occupation upon which you rely to meet the exposure requirements of the TDP.

Section 3: Bystander/Secondary Exposure (including by family members)

Enter the dates that you	were exposed to the o	ccupationally	exposed person.			
From: Year	To: Month	Year				
Provide the following information about the occupationally exposed person and his/her relationship to you:						
First Name:		Last Name: _				
Relationship: (check bo	x) Family	,	Tenant, boarder, roommate			
	Laundı	ry Customer	Other			
Social Security Numbe	r:		ational ID:ed for Foreign Claims)			
Manville Trust Claim ID (if applicable):						

PART E: SIGNATURE PAGE

All proof of claim forms must be signed by the Injured Party or the person filing the claim on his/her behalf (such as the Injured Party's personal representative). If the Injured Party or the person filing the claim on his/her behalf is represented by counsel, then counsel may sign this proof of claim form in lieu of the Injured Party or the person filing the claim on his/her behalf.

By signing this proof of claim form, you are certifying, under penalty of perjury, that all representations you have made are true and accurate to the best of your knowledge, and that neither the Injured Party nor any representative of the Injured Party previously has relinquished the right to file this personal injury claim against USM or the USM Trust.

Signature of Injured Party or Representative	
Printed name of the signatory above and his/her relation	ship to the Injured Party
Executed on:(MM/DD/YYYY)	
By signing below, counsel for the Injured Party or the behalf is certifying that the information and material submitted pursuant to and subject to the provisions of Procedure.	ls with respect to this claim are being
Signature of Counsel	

ATTACHMENT 1 - OCCUPATION CODES

- 01. Abatement Worker
- 02. Acoustical Worker
- 03. Air Conditioning Installer/ Repairer
- 04. Asbestos Worker
- 05. Beamer
- 06. Boiler Coverer
- 07. Boiler Inspector
- 08. Boiler Insulator
- 09. Boiler Operator
- 10. Boiler Repairer
- 11. Boiler Tender
- 12. Boiler Worker
- 13. Boilermaker
- 14. Boilerman
- 15. Brick Gang
- 16. Brick Mason
- 17. Brick Room Helper
- 18. Bricklayer
- 19. Brickman
- 20. Building Remodeler
- 21. Burner
- 22. Carpenter
- 23. Caulker
- 24. Ceiling Tile Installer
- 25. Cement Finisher
- 26. Cement Worker
- 27. Chipper
- 28. Chipper & Grinder
- 29. Concrete Laborer
- 30. Concrete Worker
- 31. Construction Laborer
- 32. Construction Mechanic
- 33. Construction Superintendent
- 34. Construction Worker
- 35. Contractor
- 36. Coppersmith
- 37. Crane Operator
- 38. Demolition Worker
- 39. Drywall Applicator
- 40. Drywall Taper
- 41. Electrical Technician
- 42. Electrician
- 43. Electrician Helper

- 44. Elevator Construction Worker
- 45. Erector
- 46. Final Construction Inspector
- 47. Fireman
- 48. Floor Tile/Linoleum Layer
- 49. Floor Tile Installer
- 50. Furnace Installer/Repairer
- 51. Furnace Worker
- 52. Glazer
- 53. Hammer Driver
- 54. Heat System Installer/Repairer
- 55. Heating Insulator
- 56. Heating Worker
- 57. Hod Carrier
- 58. HVAC Installer/Repairer
- 59. Industrial Electrician
- 60. Insulation Apprentice/Helper
- 61. Insulation Machinist
- 62. Insulation Worker
- 63. Insulator
- 64. Ironworker
- 65. Laborer
- 66. Lather
- 67. Lagger
- 68. Locksmith
- 69. Machinist
- 70. Maintenance Mechanic
- 71. Mason
- 72. Mason Tender
- 73. Millwright
- 74. Mortar Mixer
- 75. Paint Mixer
- 76. Painter
- 77. Pile Driver
- 78. Pipe Carrier
- 79. Pipe Coverer
- 80. Pipe Cutter
- 81. Pipe Foreman
- 82. Pipe Grinder
- 83. Pipe Hanger
- 84. Pipe Insulator
- 85. Pipe Layer
- 86. Pipe Racker
- 87. Pipe Repairer

- 88. Pipe Stripper
- 89. Pipe Welder
- 90. Pipefitter
- 91. Pipefitter Helper
- 92. Plasterer
- 93. Plumber
- 94. Refrigeration/HVAC Equipment Installer/Repairer
- 95. Rigger
- 96. Rivet Bulker
- 97. Riveter
- 98. Roofer
- 99. Sheetmetal Mechanic
- 100. Sheetmetal Worker
- 101. Sheetrock Hanger
- 102. Shingle Catcher
- 103. Siding Erector
- 104. Soundproofing Installer
- 105. Spray Insulator
- 106. Steamfitter

- 107. Structural Worker
- 108. Telephone Cable Insulator and Installer
- 109. Tile Grinder
- 110. Tile Helper
- 111. Tile Installer
- 112. Tile Layer
- 113. Tile Mechanic
- 114. Tile Operator
- 115. Tile Worker
- 116. Vinyl Asbestos Floor Tile
 - Worker
- 117. Weld Checker
- 118. Welder
- 119. Welding Assistant
- 120. Welding Helper
- 121. Welding Inspector
- 122. Welding Instructor
- 123. Welding Foreman

ATTACHMENT 2 - INDUSTRY CODES

- 01. Aerospace/Aviation
- 02. Asbestos Abatement
- 03. Asbestos Mining
- 04. Asbestos Products Manufacturing
- 05. Automobile/Mechanical Friction
- 06. Chemical
- 07. Construction Trades
- 08. Insulation
- 09. Iron/Steel
- 10. Longshore
- 11. Maritime

- 12. Military
- 13. Non-Asbestos Products Manufacturing
- 14. Petrochemical
- 15. Railroad
- 16. Shipyard Construction/Repair
- 17. Textile
- 18. Tire/Rubber
- 19. Utilities
- 20. Building Occupant/Bystander
- 21. Other

Instructions for Completing the United States Mineral Products Company Asbestos Personal Injury Settlement Trust Proof of Claim Form

This document has been designed to assist you with the completion and submission of your proof of claim ("POC") form to the United States Mineral Products Company Asbestos Personal Injury Settlement Trust (the "USM Trust").

Most important to the successful processing of your claim is to ensure that it is accurate and complete. The USM Trust encourages you to review your claim carefully before submitting it.

The USM Trust has provided detailed instructions for each section of the POC form, as well as tips for filing your claim. Please read through these instructions carefully.

The USM Trust Distribution Procedures ("TDP") allow for the reliance on claims settled by the Manville Personal Injury Settlement Trust (the "Manville Trust"), as identified in the matrix on page 5 of these instructions. Unless you are relying on a claim settled by the Manville Trust, you must attach at least one medical report to your POC form.

Claims Resolution Management Corporation ("CRMC"), the USM Trust's claims processing facility, has the capability to accept POCs in electronic format. If you wish to file claims electronically, please contact Customer Service at (800) 536-2722 or inquiry@claimsres.com for further information and instructions. If you are a first-time filer, we recommend that you contact CRMC Customer Service to register your law firm prior to submitting your claim. This will help expedite the claim-filing process.

If submitting a paper-based POC form, please send it to the following address:

Claims Resolution Management Corporation P.O. Box 12003 Falls Church, VA 22042-0683 (703) 204-9300 (800) 536-2722

6/1/2010 Form No: Instructions POC-USM Version 1

COMPONENTS OF PART A: INJURED PARTY

INJURED PARTY INFORMATION:

If you are relying on a prior settlement with the Manville Trust, you must provide the Manville Claim ID.

If the Injured Party is living and not represented by counsel, you must provide his/her contact information. If the Injured Party is deceased, you must provide the date of death and injured party's last state of residence. If this information is not provided, the claim cannot be processed.

INJURED PARTY'S REPRESENTATIVE INFORMATION:

If the Injured Party is deceased or has a personal representative or heir other than, or in addition to the injured party's attorney, please indicate the following for the representative. Certificate of Official Capacity or other estate documentation must be enclosed if applicable per state law.

COMPONENTS OF PART B: LAW FIRM INFORMATION

CLAIMANT'S COUNSEL INFORMATION:

Please enter the Law Firm Code and Attorney Code (if previously supplied to you by CRMC), or complete the law firm information. (Fax numbers, firm website addresses, and e-mail addresses are not required, but they can facilitate future communication with the firm.)

LITIGATION INFORMATION:

Please answer "Yes" or "No" (by checking the appropriate box on the POC form) to the question regarding whether an asbestos-related lawsuit has been filed against United States Mineral Products Company ("USM") on behalf of the Injured Party. If you answer "Yes," provide the date of the earliest filing against USM (if more than one occurred), as well as the State or Country (for a foreign claim) in which the lawsuit was filed.

COMPONENTS OF PART C: ASBESTOS-RELATED INJURY

(Unless the Claimant is relying on a claim previously settled by the Manville Trust, Part C of the POC form must be completed.)

DIAGNOSED INJURY:

You must select one of the three Disease Levels identified below. The Scheduled Value of each Disease Level is set forth below, as well as the medical proof required for each Disease Level.

Claims for all Disease Levels must be accompanied by either (1) a statement by the physician providing the diagnosis that at least ten years have elapsed between the date of first exposure to asbestos or asbestos-containing products and the diagnosis, or (2) a history of the Claimant's exposure sufficient to establish a ten-year latency period. A finding by a physician after the Effective Date (1/3/2006) that a Claimant's disease is "consistent with" or "compatible with" asbestosis shall not alone be treated by the USM Trust as such a diagnosis.

For nonmalignant asbestos-related disease or primary colo-rectal, laryngeal, esophageal, pharyngeal, or stomach cancer (**Disease Level I**), Claimants must demonstrate the following:

- 1. In the case of an Injured Party who was <u>living</u> at the time the claim was filed:
 - A. A physical examination of the Injured Party by the physician providing the diagnosis; and
 - B. Evidence of Bilateral Asbestos-Related Nonmalignant Disease (as defined in footnote 1 to section 5.3(a)(3) of the TDP).
- 2. In the case of an Injured Party who was deceased at the time the claim was filed:
 - A. A physical examination of the Injured Party by the physician providing the diagnosis;
 - B. A physical examination of the Injured Party by another doctor whose physical examination and findings are reliable;
 - C. Pathological evidence of the nonmalignant asbestos-related disease; or
 - D. Evidence of Bilateral Asbestos-Related Nonmalignant Disease.

Level I: Other Asbestos Disease – \$500 (Scheduled Value)

Diagnosis of a primary colo-rectal, laryngeal, esophageal, pharyngeal, or stomach cancer (plus evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease), asbestosis, or Bilateral Asbestos-Related Nonmalignant Disease, requires the following:

1. Credible evidence of four months USM Exposure prior to December 31, 1982;

- 2. Credible evidence of Significant Occupational Exposure ("SOE"); and
- 3. Either (i) a chest X-ray read by a qualified B reader of 1/0 or higher on the ILO scale, or (ii) (a) a chest X-ray read by a qualified B reader or other Qualified Physician, (b) a CT scan read by a Qualified Physician, or (c) pathology, in each case showing either bilateral interstitial fibrosis, bilateral pleural plaques, bilateral pleural thickening, or bilateral pleural calcification.

An acceptable ILO must include the date of the X-ray, the date of reading, the name of the physician or B reader, and the name of the Injured Party name and/or the Injured Party's Social Security Number. If only the B reader's or physician's initials appear on the ILO form, please write the full name on the form.

The USM Trust does not consider death certificates to be medical reports, and they will not be accepted in the place of medical reports.

For lung cancer and mesothelioma (**<u>Disease Levels II and III</u>**), Claimants must demonstrate the following:

- 1. A physical examination of the Injured Party by the physician providing the diagnosis; or
- 2. A report by a board-certified pathologist.

Level II: Lung Cancer – \$4,000 (Scheduled Value)

Diagnosis of a primary lung cancer (plus evidence of an underlying Bilateral Asbestos-Related Nonmalignant Disease), requires the following:

- 1. Credible evidence of four months USM Exposure prior to December 31, 1982; and
- 2. Credible evidence of SOE.

Level III: Mesothelioma – \$10,000 (Scheduled Value)

Diagnosis of malignant mesothelioma requires credible evidence of USM Exposure prior to December 31, 1982.

¹ SOE means employment for a cumulative period of at least five years, with a minimum of two years prior to December 31, 1982, in an industry and an occupation in which the Claimant (a) handled raw asbestos fibers on a regular basis, (b) fabricated asbestos-containing products so that the Claimant in the fabrication process was exposed on a regular basis to raw asbestos fibers, (c) altered, repaired, or otherwise worked with an asbestos-containing product such that the Claimant was exposed on a regular basis to asbestos fibers, or (d) was employed in an industry and occupation such that the Claimant worked on a regular basis in close proximity to workers engaged in the activities described in (a), (b), and/or (c).

If the Claimant has settled a claim with the Manville Trust, he/she may rely on the disease for that settlement and forgo supplying medical information to the USM Trust. Use the conversion matrix below to determine which USM Trust Disease Level corresponds to the Claimant's settled Manville Trust claim:

MANVILLE TRUST-USM TRUST DISEASE CONVERSION MATRIX							
Manville TDP	Manville Disease	USM Disease	USM Disease	USM Disease			
Version	Category/Level	Level I	Level II	Level III			
1995 TDP	CAT 1	*X					
1995 TDP	CAT 2	*X					
1995 TDP	CAT 3	*X					
1995 TDP	CAT 4	*X					
1995 TDP	CAT 7			X			
2002 TDP	LVL 1	*X (Non-					
		Malignant					
		claims only)					
2002 TDP	LVL 2	X					
2002 TDP	LVL 3	X					
2002 TDP	LVL 4	X					
2002 TDP	LVL 5	X					
2002 TDP	LVL 7		X				
2002 TDP	LVL 8			X			

^{*} Claimant must have at least 5 years of exposure.

COMPONENTS OF PART D: <u>EXPOSURE</u>

(All Claimants must complete Part D of the POC form.)

To qualify for any Disease Level under the TDP, the Claimant must demonstrate meaningful and credible exposure to asbestos or asbestos-containing products supplied, specified, manufactured, installed, maintained, or repaired by USM and/or any entity for which USM has legal responsibility. All Disease Levels except Disease Level III require at least four months USM Exposure. USM Exposure can be established in the following ways:

- 1. By use of an approved Site Code available on the CRMC website under the Documents tab, U. S. Mineral Trust;
- 2. By affidavit or sworn statement of the Claimant;
- 3. By affidavit or sworn statement of a co-worker, or affidavit of a family member in the case of a deceased Claimant (provided that the USM Trust finds such evidence reasonably reliable);
- 4. By invoices, employment, construction, or similar records; or
- 5. By interrogatory answers, sworn work histories, or depositions.

USM Worksites are to be entered using Site Codes. A list of previously qualified worksites is available on the CRMC website under the Documents tab, U. S. Mineral Trust. You may use the

worksite code "NM" (Non-Mineral) for entering worksites to qualify for SOE that do not qualify for USM Exposure. If you have a site that has not yet been qualified use the "NQ" (Not Qualified) code. An exposure affidavit will be required in this instance.

Describe all employment periods sufficient to meet the five-year SOE requirement and provide the Industry Code, Occupation Code, and Site Code for the exposure of the Claimant or occupationally exposed person. Use the Industry and Occupation Codes listed on pages 7–9 below. If you have a question regarding which Industry or Occupation Code should be used, please contact CRMC Customer Service at (800) 536-2722.

Be sure to provide the earliest month and year of exposure to asbestos because latency and length of exposure will be calculated based on this information.

COMPONENTS OF PART E: SIGNATURE PAGE

All paper-based POC forms must be signed by the Injured Party or by the person filing the claim on his/her behalf (such as the personal representative); if the Injured Party or the person filing the claim on his/her behalf is represented by counsel, then counsel may sign the POC form in lieu of the Injured Party or the person filing the claim on his/her behalf.

By signing the POC form, you are certifying that all representations you have made are true and accurate to the best of your knowledge, and that neither the Injured Party nor any representative of the Injured Party previously has relinquished the right to file this personal injury claim against USM or the USM Trust.

OCCUPATION CODES

- 01. Abatement Worker
- 02. Acoustical Worker
- 03. Air Conditioning Installer/ Repairer
- 04. Asbestos Worker
- 05. Beamer
- 06. Boiler Coverer
- 07. Boiler Inspector
- 08. Boiler Insulator
- 09. Boiler Operator
- 10. Boiler Repairer
- 11. Boiler Tender
- 12. Boiler Worker
- 13. Boilermaker
- 14. Boilerman
- 15. Brick Gang
- 16. Brick Mason
- 17. Brick Room Helper
- 18. Bricklayer
- 19. Brickman
- 20. Building Remodeler
- 21. Burner
- 22. Carpenter
- 23. Caulker
- 24. Ceiling Tile Installer
- 25. Cement Finisher
- 26. Cement Worker
- 27. Chipper
- 28. Chipper & Grinder
- 29. Concrete Laborer
- 30. Concrete Worker
- 31. Construction Laborer
- 32. Construction Mechanic
- 33. Construction Superintendent
- 34. Construction Worker
- 35. Contractor
- 36. Coppersmith
- 37. Crane Operator
- 38. Demolition Worker
- 39. Drywall Applicator
- 40. Drywall Taper
- 41. Electrical Technician
- 42. Electrician
- 43. Electrician Helper

- 44. Elevator Construction Worker
- 45. Erector
- 46. Final Construction Inspector
- 47. Fireman
- 48. Floor Tile/Linoleum Layer
- 49. Floor Tile Installer
- 50. Furnace Installer/Repairer
- 51. Furnace Worker
- 52. Glazer
- 53. Hammer Driver
- 54. Heat System Installer/Repairer
- 55. Heating Insulator
- 56. Heating Worker
- 57. Hod Carrier
- 58. HVAC Installer/Repairer
- 59. Industrial Electrician
- 60. Insulation Apprentice/Helper
- 61. Insulation Machinist
- 62. Insulation Worker
- 63. Insulator
- 64. Ironworker
- 65. Laborer
- 66. Lather
- 67. Lagger
- 68. Locksmith
- 69. Machinist
- 70. Maintenance Mechanic
- 71. Mason
- 72. Mason Tender
- 73. Millwright
- 74. Mortar Mixer
- 75. Paint Mixer
- 76. Painter
- 77. Pile Driver
- 78. Pipe Carrier
- 79. Pipe Coverer
- 80. Pipe Cutter
- 81. Pipe Foreman
- 82. Pipe Grinder
- 83. Pipe Hanger
- 84. Pipe Insulator
- 85. Pipe Layer
- 86. Pipe Racker
- 87. Pipe Repairer

- 88. Pipe Stripper
- 89. Pipe Welder
- 90. Pipefitter
- 91. Pipefitter Helper
- 92. Plasterer
- 93. Plumber
- 94. Refrigeration/HVAC Equipment Installer/Repairer
- 95. Rigger
- 96. Rivet Bulker
- 97. Riveter
- 98. Roofer
- 99. Sheetmetal Mechanic
- 100. Sheetmetal Worker
- 101. Sheetrock Hanger
- 102. Shingle Catcher
- 103. Siding Erector
- 104. Soundproofing Installer
- 105. Spray Insulator
- 106. Steamfitter

- 107. Structural Worker
- 108. Telephone Cable Insulator and Installer
- 109. Tile Grinder
- 110. Tile Helper
- 111. Tile Installer
- 112. Tile Layer
- 113. Tile Mechanic
- 114. Tile Operator
- 115. Tile Worker
- 116. Vinyl Asbestos Floor Tile
 - Worker
- 117. Weld Checker
- 118. Welder
- 119. Welding Assistant
- 120. Welding Helper
- 121. Welding Inspector
- 122. Welding Instructor
- 123. Welding Foreman

INDUSTRY CODES

- 01. Aerospace/Aviation
- 02. Asbestos Abatement
- 03. Asbestos Mining
- 04. Asbestos Products Manufacturing
- 05. Automobile/Mechanical Friction
- 06. Chemical
- 07. Construction Trades
- 08. Insulation
- 09. Iron/Steel
- 10. Longshore
- 11. Maritime

- 12. Military
- 13. Non-Asbestos Products Manufacturing
- 14. Petrochemical
- 15. Railroad
- 16. Shipyard Construction/Repair
- 17. Textile
- 18. Tire/Rubber
- 19. Utilities
- 20. Building Occupant/Bystander
- 21. Other

CHECKLIST

Please review the checklist below prior to submitting your claim to ensure that all required information and documentation are provided:

An alleged injury has been indicated.
At least one medical report has been provided (required only when the Claimant is <u>not</u> relying on a prior settlement with the Manville Trust).
If the exposure site of NQ was used, an exposure affidavit is required.
If a certificate of official capacity is required in the claimants state one has been provided.
For a paper-based POC form, it has been signed by the Injured Party or the person filing the form on his/her behalf (such as the Injured Party's personal representative), as well as by counsel if the Injured Party or his/her representative is represented by counsel.
If someone other than the Injured Party is filing the claim on his/her behalf (such as the Injured Party's personal representative), you must provide the representative's contact information and attach a Certificate of Official Capacity, if required by state law.

United States Mineral Products Company

List of Unacceptable Physicians

Physicians

- 1. Dr. James Ballard, Birmingham, AL
- 2. Dr. Kevin Cooper, Pascagoula, MS
- 3. Dr. Todd Coulter, Ocean Springs, MS
- 4. Dr. Andrew Harron, Kenosha, WI
- 5. Dr. Ray Harron, Bridgeport, WV
- 6. Dr. Glynn Hilburn, Moss Point, MS
- 7. Dr. Barry Levy, Sherborn, MA
- 8. Dr. Gregory Martindale, Mobile, AL
- 9. Dr. Gregory Nayden, Mobile, AL
- 10. Dr. W. Allen Oaks, Mobile, AL

United States Mineral Products Company Asbestos Trus	st Approved Produ	ıct List		
Product	Manufacturer	Asbestos	BEGIN YR	END YR
Cafco Spray	US Mineral	Yes	1954	1958
Cafco Blaze-Shield	US Mineral	Yes	1958	1971
Cafco Blaze-Shield Type D	US Mineral	Yes	1964	1972
Cafco Blaze-Shield Patching Fiber	US Mineral	Yes	1954	1971
Cafco Blaze-Shield Patching Fiber Type D	US Mineral	Yes	1965	1971
Cafco Spray Type 1	US Mineral	Yes	1954	1958
Cafco Sound-Shield	US Mineral	Yes	1958	1969
Cafco Blaze-Shield Type H	US Mineral	Yes	1969	1971
Cafco Heat-Shield	US Mineral	Yes	1958	1971
Cafco Power-Shield	US Mineral	Yes	1964	1971
J Spray	US Mineral	Yes	1964	1967
Cominco	US Mineral	Yes	pre-1958	1971
Cafco Heat-Shield Type 2	US Mineral	Yes	1961	1962
All-Purpose Cement (alternate name for Cominco)	US Mineral	Yes	pre-1958	1971
Mark II Coating	US Mineral	Yes	1966	1972
Ace-Tite Cement (alternate name for Cominco)	US Mineral	Yes	pre-1958	1971
Cafco Blaze-Shield Type M	US Mineral	Yes	1964	1965
Cafco Weather-Shield	US Mineral	Yes	1970	1972
Sound-Shield "85"	US Mineral	No		
Cominco Monoply	US Mineral	Yes	1963	1971

United States Mineral Products Company Asbestos Trust Approved Site List as of December 7, 2009

CITE		C.T.	CDATE
SITE	CITY	<u>ST.</u>	<u> </u>
AQUATRAIN TEAM TRACK SIDING	ANCHORAGE	AK	9/21/1962
NORTHERN CONSOLIDATED HANGAR	BETHEL	AK	8/11/1958
MALLOW CONSTRUCTION CO	CLEAR	AK	1/1/1961
ALASKA STEAMSHIP COMPANY	SPENARD	AK	6/9/1960
MALLOW CONSTRUCTION CO, ALASKA FREIGHT LINES, 111 26th AVE.	SPENARD	AK	1/1/1958
SHOOK & FLETCHER INSULATION COMPANY, 2431 1ST AVE.	BIRMINGHAM	AL	12/30/1957
GULF STATE PAPER COMPANY	DEMOPOLIS	AL	11/23/1966
ROBERTS PLASTERING	GADSDEN	AL	2/7/1966
BOEING AIRCRAFT PLANT, INDUSTRIAL RESEARCH PARK	HUNTSVILLE	AL	2/25/1966
ALLIED PAPER CORPORATION	JACKSON	AL	7/8/1964
INSULATION ENGINEERS, INCORPORATED	MOBILE	AL	9/6/1968
JOHNS BARGAIN HOUSE CORPORATION	MOBILE	AL	3/9/1965
MEDOR, WESTERN AVE.	MOBILE	AL	12/5/1968
MEARTHON SOUTHERN PAPER COMPANY	NAHELOA	AL	6/24/1964
GEORGIA PACIFIC PAPER CORPORATION	CROSSETT	AR	6/30/1965
ARKANSAS POWER & LIGHT, MOSES STATION	FOREST CITY	AR	9/17/1968
BREEDING INSULATION	LITTLE ROCK	AR	2/24/1967
JIFFY INSULATION COMPANY	LITTLE ROCK	AR	4/20/1967
KNOX HILL SIDING	LITTLE ROCK	AR	1/24/1968
ARKANSAS POWER & LIGHT, LYNCH STATION	NORTH LITTLE ROCK	AR	3/22/1971
FEDERAL OFFICE BUILDING, WARRIOR CONSTRUCTION COMPANY	PINE BLUFF	AR	8/8/1966
ARKANSAS POWER & LIGHT, COUCH STATION	STAMPS	AR	1/4/1967
LITCHFIELD TEAM TRACK	LITCHFIELD	ΑZ	7/18/1964
UNITED WHOLESALE, S. PACIFIC TEAM TRACK	PHOENIX	ΑZ	2/20/1965
TEXAS AND PACIFIC	TUSCON	ΑZ	4/11/1960
OWEN HASKEL TSPS BUILDING	ALAMEDA	CA	1/14/1971
RICHFIELD COKE BUILDING	ALAMEDA CITY	CA	6/26/1967
SOUTHERN PACIFIC TEAM TRACK	ANAHEIM	CA	11/16/1961
AIRCODE TEMESCAL FACILITIES	BERKELEY	CA	4/21/1969
SANTA FE TEAM TRACK	BERKELEY	CA	3/11/1970
SIERRA VISTA SCHOOL, REAR	BEVERLY HILLS	CA	3/17/1967
STATE FARM INSURANCE BUILDING	CAMARILLO	CA	7/8/1969
MARINE BASE THEATRE	CAMP PENDLETON	CA	5/1/1969
TEAM TRACK #876	CAMPBELL/SAN JOSE	CA	6/13/1969
COLMA TEAM TRACK	COLMA	CA	6/20/1964
ADVANCED INSTALLATION, INCORPORATED	CULVER CITY	CA	6/5/1965
SERRAMONTE SHOPPING CENTER, DUAL THEATERS	DALY CITY	CA	9/23/1971
S.D.S. BUILDING, 711 AVIATION BLVD.	EL SIGUNDO	CA	3/13/1968
DOYLE & ASSOCIATES	FULLERTON	CA	10/22/1963
ROSARY HIGH SCHOOL	FULLERTON	CA	4/14/1967
SANTA FE TEAM TRACK	FULLERTON	CA	1/13/1964
GLENDALE PUBLIC TEAM TRACK	GLENDALE		8/11/1959
GLENDALE SIDING, CERRITOS & RAILROAD	GLENDALE	CA	8/31/1965
LESSER TOWER PROJECT, GLENDALE SIDING	GLENDALE	CA	9/27/1965
HAYWARD STATE COLLEGE	HAYWARD	CA	12/23/1970
	· · · · · · · · · · · · · · · · · · ·		,,,

HAMMADD STATE LIDDADY	1147/14/48		10/10/1070
HAYWARD STATE LIBRARY	HAYWARD	CA	12/18/1970
HUNTSWOOD, S.P. TEAM TRACKS	HAYWARD	CA	4/17/1970
LAIRPORT TEAM TRACK	INGLEWOOD	CA	
UNIVERSAL FIREPROOFING CORPORATION	INGLEWOOD	CA	
GENERAL TELEPHONE COMPANY, 210 W. OCEAN BLVD.	LONG BEACH	CA	1/4/1967
KARPEN PLYWOOD JOB	LONG BEACH	CA	2/17/1967
LONG BEACH TEAM TRACK	LONG BEACH	CA	5/3/1968
3RD AND SANTE FE TEAM TRACK	LOS ANGELES	CA	1/13/1964
BRINKMAN STEEL & SUPPLY COMPANY	LOS ANGELES	CA	10/20/1969
CENTURY CITY JOB	LOS ANGELES	CA	10/30/1965
INTERNATIONAL AIRPORT	LOS ANGELES	CA	3/13/1970
L.D. REEDER COMPANY, 2900 ROWENA AVE.	LOS ANGELES	CA	12/12/1958
LAX SUPER BAY HANGER, NORTHRIDGE MALL	LOS ANGELES	CA	4/29/1971
LOS ANGELES LATHING COMPANY	LOS ANGELES	CA	7/2/1969
MADISON APARTMENTS	LOS ANGELES	CA	8/24/1964
OCCIDENTAL CENTER	LOS ANGELES	CA	10/2/1962
OCCIDENTAL COLLEGE, A.G. COONS CENTER	LOS ANGELES	CA	11/5/1967
PACIFIC COAST FIREPROOFING	LOS ANGELES	CA	6/10/1969
PACIFIC COMMERCIAL	LOS ANGELES	CA	5/25/1966
PRUDENTIAL SAVINGS & LOAN	LOS ANGELES	CA	4/23/1966
S.F. TEAM TRACK	LOS ANGELES	CA	
SAN CARLOS CIRCLE STAR THEATRE	LOS ANGELES	CA	
SUPER BAY HANGAR, AMERICAN AIRLINES	LOS ANGELES		8/27/1970
VALLEY PLACE METHODIST CHURCH	LOS ANGELES	CA	8/10/1966
WILDASIN TEAM TRACK	LOS ANGELES	CA	1/21/1966
SANTA FE TEAM TRACK	MARCH FIELD	CA	1/13/1964
U.S. FIREPROOFING(311 S. GREENWOOD AVE.	MONTEBELLO	CA	2/1/1968
HOMART JOB	NORTH HOLLYWOOD	CA	5/5/1966
PETE KING CORPORATION	NORTH HOLLYWOOD	CA	6/13/1966
PRUDENTIAL SAVINGS & LOAN	NORTH HOLLYWOOD		5/5/1966
UNIVERSAL FIREPROOFING CORPORATION	NORTH HOLLYWOOD	CA	• •
SOUTHERN PACIFIC TEAM TRACK	NORTHRIDGE	CA	9/22/1964
ADMINISTRATION BUILDING, 8TH AND MADISON	OAKLAND	CA	
B.A.R.T. ADMINISTRATION BUILDING	OAKLAND	CA	
HARRISON STREET OFFICE BUILDING	OAKLAND	CA	
L.D. REEDER COMPANY, 92 CHARTER ST.	OAKLAND	CA	1/7/1958
,			
ARMY TERMINAL, KWAJALEIN NIKE-ZEUS PROJECT	OAKLAND/KWAJALEIN	CA	• •
UNION BANK, GLENDALE SIDING	ORANGE	CA	• •
UNION BANK, GLENDALE TEAM TRACK	ORANGE	CA	10/24/1964
DESERT HOSPITAL	PALM SPRINGS	CA	
SYNTEX, CALIFORNIA AVE. TEAM TRACK	PALO ALTO	CA	5/22/1970
PASEDENA TEAM TRACK	PASEDENA	CA	
POMONA TEAM TRACK	POMONA	CA	
SAN MATEO HALL OF JUSTICE BUILDING	REDWOOD CITY	CA	
BROADWAY DEPARTMENT STORE	RIVERSIDE	CA	
C.C.T. TEAM TRACK	SACRAMENTO	CA	9/7/1961
HIRAM JOHNSON SCHOOL	SACRAMENTO	CA	2/28/1958

L.D. REEDER COMPANY, V ST.	SACRAMENTO	CA	9/25/1959
SOUTHERN PACIFIC TEAM TRACK	SACRAMENTO	CA	4/7/1970
GEORGIA PACIFIC, PAPER MILL	SAMOA	CA	4/21/1965
SEARS JOB, MILL & E STS. TEAM TRACK	SAN BERNADINO	CA	1/21/1966
STATE COLLEGE, CENTRAL HEAT & AIR CONDITION BUILDING	SAN BERNADINO	CA	3/23/1967
EXECUTIVE HOUSE, 1ST AND C STS.	SAN DIEGO	CA	6/28/1968
LAIRPORT & SANTA FE TEAM TRACKS	SAN DIEGO	CA	4/26/1968
PASADENA & SANTA FE TEAM TRACKS	SAN DIEGO	CA	4/4/1968
SANTA FE TEAM TRACK	SAN DIEGO	CA	8/10/1961
WESTGATE EXECUTIVE HOUSE	SAN DIEGO	CA	10/4/1968
660 MINNESOTA ST.	SAN FRANCISCO	CA	8/12/1960
ASSOCIATED LATHING & PLASTERING COMPANY	SAN FRANCISCO	CA	2/25/1966
BEALE ST. TEAM TRACK	SAN FRANCISCO	CA	4/29/1965
BETHLEHEM STEEL COMPANY	SAN FRANCISCO	CA	3/24/1969
BURNS DRAYAGE COMPANY, 516 TOWNSEND ST.	SAN FRANCISCO	CA	5/21/1965
FEDERAL OFFICE BUILDING, 507 POLK ST.	SAN FRANCISCO	CA	2/18/1963
FEDERAL OFFICE BUILDING, 660 MINNESOTA ST.	SAN FRANCISCO	CA	8/10/1962
FRANK D. SMITH COMPANY	SAN FRANCISCO	CA	1/1/1966
HUNTSWOOD, S.P. TEAM	SAN FRANCISCO	CA	7/22/1970
JAMES L. WHITTAKER	SAN FRANCISCO	CA	12/29/1967
L.D. REEDER COMPANY, 1258 SANSOME ST.	SAN FRANCISCO	CA	12/12/1958
	SAN FRANCISCO	CA	1/9/1963
L.D. REEDER COMPANY, 660 MINNESOTA AVE.	SAN FRANCISCO		
ROSCOE-AJAX-KNICKERBOCKER, 507 POLK ST.		CA	4/8/1964
SANTA FE TEAM TRACK	SAN FRANCISCO	CA	8/23/1965
SENIOR CITIZEN BUILDING, 1880 PINE ST.	SAN FRANCISCO	CA	10/15/1971
SEQUOIA	SAN FRANCISCO	CA	2/6/1968
SEQUOIA, PASADENA TEAM TRACK	SAN FRANCISCO	CA	3/19/1968
SEQUOIA, SANTA FE TEAM TRACK	SAN FRANCISCO	CA	5/15/1968
SOUTHERN PACIFIC TEAM TRACK	SAN FRANCISCO	CA	2/20/1965
STATE COLLEGE, PHYSICAL SCIENCE BUILDING	SAN FRANCISCO	CA	8/26/1971
UNIVERSAL FIREPROOFING CORPORATION, TRACK #876 AT 16TH AND HARRISON	SAN FRANCISCO	CA	1/23/1968
WATERLOO SPUR, WESTERN PACIFIC RAILWAY	SAN FRANCISCO	CA	11/7/1969
WELLS FARGO BUILDING, 44 MONTGOMERY ST.	SAN FRANCISCO	CA	2/28/1967
WELLS FARGO, SANTA FE TEAM TRACK	SAN FRANCISCO	CA	8/27/1965
SANTA ANA POST OFFICE	SANTA ANA	CA	7/26/1967
SANTA ANA TEAM TRACK	SANTA ANA	CA	5/15/1967
J.C. PENNEY	SIERRA MADRE	CA	6/10/1971
PACIFIC COAST FIREPROOFING	SIERRA MADRE	CA	5/26/1969
VAL DIEHL, 12952 DICKENS ST.	STUDIO CITY	CA	8/28/1964
J.C. PENNEY	TORRANCE	CA	1/20/1970
MT. SAN ANTONIO JUNIOR COLLEGE	WALNUT	CA	1/27/1967
UNIVERSAL FIREPROOFING CORPORATION	WEST PALM SPRINGS	CA	10/20/1967
CLETUS G. COOPER, 420 S. CORONA AVE.	COLORADO SPRINGS	СО	8/11/1961
CLETUS G. COOPER, 6 1/2 W. CIMARRON ST.	COLORADO SPRINGS	СО	10/2/1959
EMPIRE ACOUSTICS & FIREPROOF	COLORADO SPRINGS	СО	2/13/1958
C.B. & Q. TEAM TRACK	DENVER	СО	10/13/1961
CELEBRITY LANE, 320 S. COLORADO BLVD.	DENVER	СО	5/23/1960
CELESTIT E TIVE, 320 S. COLOTVIDO DEVD.	D L I V L I V		3/23/1300

LALIDEN DUDT INCORDODATED	DENVER	со	E /2E /10E9
LAUREN BURT, INCORPORATED M.A. MACKAY COMPANY	DENVER	co	5/25/1958 11/10/1961
MEADE & MOUNT CONSTRUCTION COMPANY	DENVER	co	10/2/1961
NIELSON PLASTERING	DENVER	co	2/9/1965
ASBESTOS DISTRUBUTORS CORPORATION		CT	
CATHOLIC HIGH SCHOOL FOR GIRLS	BRIDGEPORT		12/19/1963
	BRIDGEPORT	CT	7/1/1963
CENTRAL POLICE HEADQUARTERS FEDERAL COURT HOUSE BUILDING	BRIDGEPORT BRIDGEPORT	CT	10/11/1965
		CT	1/25/1968
FEDERAL OFFICE BUILDING, LAFAYETTE ST.	BRIDGEPORT	CT	10/30/1967
JEWISH COMMUNITY CENTER	BRIDGEPORT	CT	8/14/1964
PEOPLES SAVINGS BANK, MAIN ST.	BRIDGEPORT	CT	12/11/1964
POST PUBLISHING COMPANY	BRIDGEPORT	CT	5/31/1966
173 PARK AVE.	EAST HARTFORD	CT	7/12/1963
EMPIRE-ACE INSULATION MANUFACTURING COMPANY, 175 PARK AVE.	EAST HARTFORD	CT	1/1/1963
OWENS-CORNING SUPPLY CORPORATION	EAST HARTFORD	CT	12/18/1963
US TOBACCO COMPANY, WEST PUTNAM AVE.	GREENWICH	CT	6/11/1969
CARPENTER STEEL COMPANY, 277 NEW PARK AVE.	HARTFORD	CT	9/28/1965
FRED WISH ELEMENTARY SCHOOL	HARTFORD	CT	4/4/1961
GAM-BECK BUILDING	HARTFORD	CT	10/14/1968
HARTFORD INSURANCE COMPANY, SUMNER ST.	HARTFORD	CT	6/1/1967
NORTHWEST CATHOLIC HIGH SCHOOL	HARTFORD	CT	1/31/1961
68 GARDEN ST.	MANCHESTER	CT	9/20/1972
CUMMINGS INSULATION COMPANY, STATE ST.	MERIDEN	CT	9/7/1966
NEW BRITIAN TRUST COMPANY	NEW BRITIAN	CT	4/26/1965
GRACE NEW HAVEN HOSPITAL	NEW HAVEN	CT	7/6/1964
SOUTHERN CONNECTICUT STATE COLLEGE, AUDITORIUM	NEW HAVEN	CT	6/21/1967
STERLING SCHOOL OF MEDICINE, B WING	NEW HAVEN	CT	1/17/1966
YALE ART & ARCHITECTURE BUILDING	NEW HAVEN	CT	4/16/1963
U.S. COAST GUARD ACADAMY, BILLIARD HALL	NEW LONDON	CT	10/12/1966
PERRAMAR HOTEL	OLD SAYBROOK	CT	5/9/1958
943 HOPE ST.	SPRINGDALE	СТ	
RICHLEE ACOUSTICS & FIREPROOF	SPRINGDALE	СТ	1/1/1959
CALDOR DEPARTMENT STORE	STAMFORD	СТ	1/1/1965
HIGH RIDGE PARK	STAMFORD	CT	7/28/1969
TELEREGISTER CORPORATION	STAMFORD	CT	12/10/1959
WOODSIDE JUNIOR HIGH SCHOOL	STAMFORD	СТ	9/26/1966
TORRINGTON MANUFACTURING COMPANY	TORRINGTON	СТ	5/16/1963
THERMAL ACOUSTICS, INC., SCOTLAND SCHOOL	WEST HAVEN	СТ	1/1/1967
THERMAL ACOUSTICS, INCORPORATED, 81 FAREWELL ST.	WEST HAVEN	CT	1/3/1963
THERMAL ACOUSTICS, INCORPORATED, MASON LAB	WEST HAVEN	CT	11/23/1966
ARMSTRONG CONSTRUCTION & SUPPLY	WETHERSFIELD	СТ	7/26/1963
201 KENNEDY ST.	WASHINGTON	DC	3/28/1968
5050 ADDISON RD.	WASHINGTON	DC	2/27/1964
5050 MINNESOTA AVE.	WASHINGTON	DC	
AMERICAN UNIVERSITY, DORM "K"	WASHINGTON	DC	
AMERICAN UNIVERSITY, MARY GRAYDON HALL	WASHINGTON	DC	2/5/1968
EQUITABLE LIFE INSURANCE COMPANY	WASHINGTON	DC	6/19/1963

HOD CARRIERS BUILDING, 16TH AND EYE STS.	WASHINGTON	DC	12/22/1958
JOHN F. KENNEDY CENTER, 23RD AND F STS.	WASHINGTON	DC	•
LORD & TAYLOR BUILDING	WASHINGTON		4/30/1959
NATIONAL PRESBYTERIAN CHURCH	WASHINGTON	DC	
OFFICE BUILDING, 19TH AND JEFFERSON STS.	WASHINGTON	DC	11/7/1963
PARK OPERATIONS BUILDING	WASHINGTON	DC	•
RUBINO & FERRIS, INCORPORATED	WASHINGTON	DC	12/19/1961
SHRINE OF IMMACULATE CONCEPCION, CATHOLIC UNIVERSITY	WASHINGTON	DC	2/5/1959
WALTER E. CAMPBELL COMPANY	WASHINGTON	DC	5/9/1966
WASHINGTON CATHEDRAL	WASHINGTON	DC	7/20/1970
WASHINGTONIAN TOWERS	WASHINGTON	DC	10/28/1965
DOVER A.F.B., OIC AIR FREIGHT TERMINAL	DOVER	DE	11/6/1967
DOVER BANK	DOVER	DE	4/30/1965
STATE BOARD OF HEALTH BUILDING	DOVER	DE	1/6/1960
UNIVERSITY OF DELAWARE, COLLEGE & EDUCATION BUILDING	NEWARK	DE	3/27/1967
ATLAS CHEMICAL JOB	WILMINGTON	DE	11/30/1965
INDUSTRIAL TRUST BUILDING	WILMINGTON	DE	• •
TURNER CONSTRUCTION, 100 W. 10TH ST. BUILDING	WILMINGTON	DE	
WILMINGTON TRUST COMPANY	WILMINGTON	DE	1/13/1960
FLORIDA SUGAR CORPORATION	BELLE GLADE	FL	9/9/1963
CIDCO PARK, CITY POINT SIDING	COCOA	FL	1967
UNIVERSITY OF MIAMI, ENGINEERING SCHOOL	CORAL GABLES	FL	5/18/1959
JOHNS-MANVILLE SALES CORPORATION	FLORIDA CITY	FL	11/8/1966
POWER HOUSE, TURKEY POINT	FLORIDA CITY	FL	2/10/1967
FLORIDA INSULATION & FIREPROOF, 2617 S.W. 2ND AVE.	FT. LAUDERDALE	FL	12/9/1957
FLORIDA POWER & LIGHT, FORT EVERGLADES PLANT	FT. LAUDERDALE	FL	11/28/1966
NORTH BROTHERS, INCORPORATED, 5641 N.W. 6TH AVE.	FT. LAUDERDALE	FL	4/5/1967
NEW V.A. HOSPITAL	GAINESVILLE	FL	6/1/1965
AIR ROUTE TRAFFIC CONTROL CENTER	HILLIARD	FL	7/25/1959
1ST FEDERAL JOB, SEABOARD-HOLLYWOOD FREIGHT	HOLLYWOOD	FL	5/15/1970
ACOUSTI-ENGINEERING COMPANY	JACKSONVILLE	FL	
EARL CORPORATION - INTERPACE	JACKSONVILLE	FL	7/10/1964
NAVAL AIR STATION, COMMUNICATIONS BUILDING	JACKSONVILLE	FL	5/4/1967
NORTH BROTHERS, INCORPORATED, 621 E. FORSYTH ST.	JACKSONVILLE	FL	8/29/1966
NORTHSIDE GENERATING STATION	JACKSONVILLE	FL	10/12/1971
12703 14TH AVE.	MIAMI	FL	8/8/1958
DADELAND SHOPPING CENTER	MIAMI	FL	5/20/1966
DOW BUILDING, PRAEHL JOB	MIAMI	FL	4/29/1970
E.A.L. BUILDING	MIAMI	FL	2/24/1971
FLORIDA INSULATION & FIREPROOF	MIAMI	FL	2/26/1959
JORDAN MARSH ADDITION	MIAMI	FL	7/16/1958
LEBANON HOSPITAL	MIAMI	FL	5/2/1961
LOTSPEICH COMPANY	MIAMI	FL	10/31/1964
MIAMI AIRPORT, TERMINAL BUILDING, SECTION 7	MIAMI	FL	12/9/1957
TECHNICAL HIGH SCHOOL, 95TH ST.	MIAMI	FL	11/7/1959
NEW SENIOR HIGH SCHOOL, D CAUSEWAY	NEW SMYRNA BEACH	FL	9/24/1962
ACOUSTI-ENGINEERING COMPANY, 360 W. GRANT ST.	ORLANDO	FL	9/14/1967

FIRST NATIONAL BANK	ORLANDO	FL	2/26/1959
JUNIORS RESTAURANT JOB	ORLANDO	FL	2/5/1968
WALT DISNEY WORLD PROJECT	ORLANDO	FL	2/25/1970
LINCOLN MEMORIAL SCHOOL	PALMETTO	FL	5/2/1961
TYNDALL A.F.B, GROUND POWERED EQUIPMENT SHOP	PANAMA	FL	5/21/1962
TYNDALL A.F.B., JET TEST CELL	PANAMA	FL	8/1/1961
SHOOK & FLETCHER INSULATION COMPANY	PENSACOLA	FL	10/11/1966
SPECIALTY CONTRACTORS, INCORPORATED, 100 W. HERMAN ST.	PENSACOLA	FL	8/27/1962
BURDINES DEPARTMENT STORE	POMPANO BEACH	FL	3/12/1969
POMPANO FASHON SQUARE	POMPANO BEACH	FL	6/11/1969
FLORIDA POWER & LIGHT GENERATING STATION	PORT EVERGLADE	FL	3/16/1966
BIZARRE INTERNATIONAL, RT. #1	RIVIERA BEACH	FL	11/11/1959
ELECTRONIC MECHANICAL RESEARCH BUILDING	SARASOTA	FL	5/16/1958
FIRST FEDERAL SAVINGS & LOAN ASSOCIATION	ST. PETERSBURG	FL	1/3/1957
BLACKBURN INSULATION SERVICE	TALLAHASSEE	FL	2/28/1966
2711 ADAMO DR.	TAMPA	FL	3/12/1965
ACOUSTI-ENGINEERING COMPANY, 5134 W. HANNA AVE.	TAMPA	FL	3/16/1966
ALAFIA LUTZ JOB	TAMPA	FL	9/6/1972
JOHNS-MANVILLE SALES CORPORATION	TAMPA	FL	1/29/1968
LOW RENT HOUSING	TAMPA	FL	1/29/1908
FLORIDA POWER & LIGHT, TICE SIDING	TICE	FL	5/8/1969
·			
UNIVERSAL INSUL MACH COMPANY	WINTER HAVEN	FL	8/13/1962
TONY POSTERO INSULATION	ATHENS		8/30/1961
230 PEACHTREE ST. BUILDING	ATLANTA	GA	8/31/1964
ACOUSTI-ENGINEERING COMPANY	ATLANTA	GA	11/5/1957
ATLANTA FAB. DIVISION, DOCK #9	ATLANTA	GA	2/6/1969
AUSTIN COMPANY, WHITE COLUMNS ON PEACHTREE	ATLANTA	GA	5/10/1966
COCOA OFFICE BUILDING, MARIETTA AND NORTH AVE.	ATLANTA	GA	5/28/1968
IRA HARDIN COMPANY, OFFICE BUILDING	ATLANTA	GΑ	7/31/1968
MUNICIPAL AUDITORIUM	ATLANTA		8/20/1959
NEW COCA COLA GARAGE	ATLANTA	GΑ	12/11/1964
NORTH BROTHERS, INCORPORATED, 3250 WOODSTOCK RD. S.E.	ATLANTA	GΑ	4/17/1969
NORTH BROTHERS, INCORPORATED, 911 MONROE DR. N.E.	ATLANTA	GΑ	10/28/1964
SHOOK & FLETCHER INSULATION COMPANY, 95 MILTON AVE. S.E.	ATLANTA	GΑ	5/2/1968
TOWER MOTEL, ATLANTA AIRPORT	ATLANTA	GΑ	5/2/1961
EAST FANNIN SCHOOL	BLUE RIDGE	GA	10/26/1959
MUNDET CORK COMPANY, 3589 BROAD ST.	CHAMBLEE	GΑ	12/17/1962
NORTH BROTHERS, INCORPORATED, 560 INDIAN ST.	SAVANNAH	GΑ	12/7/1966
UNION BAG CAMP, OLD BARGE BUILDING	SAVANNAH	GΑ	6/1/1965
UNION BAG CAMP, PAPER CORPORATION	SAVANNAH	GA	1/1/1965
CRAIG & COMPANY, 350 WARD AVE.	HONOLULU	н	10/14/1963
WEIDENHAF CORPORATION, BUILDING #1037	ALGONA	IA	 7/3/1959
ALLIED CONSTRUCTION SERVICE, 35TH AND STATE ST.	BETTENDORF	IA	3/27/1959
MERCHANTS NATIONAL BANK	CEDAR RAPIDS	IA	1/26/1962
ST. LUKE'S METHODIST HOSPITAL	CEDAR RAPIDS	IA	4/15/1967
MACHINE PRODUCTS COMPANY	CORYDON	IA	8/30/1967
ART GALLERY & MUSEUM	DAVENPORT	IA	7/10/1962
S. LEEN & MOSEON	27.172.11. 011.1	., \	., 10, 1002

FIRST FEDERAL SAVINGS & LOAN BUILDING	DAVENPORT	IA	5/10/1965
NORTHWEST BELL TELEPHONE BUILDING	DAVENPORT	IA	1/28/1958
YMCA	DAVENPORT	IA	5/16/1963
ALLIED CONSTRUCTION SERVICE, 2122 FLEUR DR.	DES MOINES	IA	5/16/1963
ALLIED CONSTRUCTION SERVICE, 412 EAST GRAND AVE.	DES MOINES	IA	2/27/1959
FARMERS ELEVATOR INSURANCE COMPANY	DES MOINES	IA	9/7/1962
IOWA SHEET METAL CONSTRUCTION, 2601 BELL AVE.	DES MOINES	IA	11/8/1968
NATIONAL TRAVELERS LIFE INSURANCE BUILDING	DES MOINES	IA	10/5/1959
REGISTER & TRIBUNE BUILDING	DES MOINES	IA	6/21/1967
ROOSEVELT HIGH SCHOOL ADDITION	DES MOINES	IA	4/6/1962
STORM LAKE HIGH SCHOOL	DES MOINES	IA	10/31/1961
DORMAN FIELD HOUSE	FAYETTE	IA	9/18/1963
UNIVERSITY OF IOWA, BASIC SCIENCE BUILDING	IOWA CITY	IA	10/12/1971
SHRIVER VAN HORN COMPANY	JEFFERSON	IA	6/5/1961
NEW HIGH SCHOOL	MAQUOKETA	IA	9/18/1963
STORM LAKE HIGH SCHOOL	STORM LAKE	IA	10/30/1961
POTLATCH FOREST, INCORPORATED	LEWISTON	ID	4/12/1968
HIGHLAND HOSPITAL	BELVIDERE	IL	10/21/1968
INTERNATIONAL HARVESTER, GERMACK AND 17TH	BROADVIEW	IL	5/31/1962
RIVER OAKS SHOPPING CENTER	CALUMET CITY	IL	7/8/1966
HERMAN M. ADLER CLINIC	CHAMPAIGN	IL	3/2/1966
230 W. MONROE ST.	CHICAGO	IL	11/22/1969
ALBERT EINSTEIN SCHOOL	CHICAGO	IL	6/14/1960
AMERICAN DENTAL ASSOCIATION BUILDING	CHICAGO	IL	10/19/1964
ASBESTOS & MAGNESIA MATERIALS	CHICAGO	IL	11/14/1957
BENEFIT TRUST LIFE INSURANCE COMPANY	CHICAGO	IL	2/3/1964
BODINE ELECTRIC COMPANY	CHICAGO	IL	9/5/1967
CHILDREN'S MEMORIAL HOSPITAL	CHICAGO	IL	6/19/1961
COMMONWEALTH EDISON STATE GENERATING STATION	CHICAGO	IL	6/3/1968
CONN MUTUAL LIFE INSURANCE COMPANY	CHICAGO	IL	9/27/1965
FOSTER KLEISER CORPORATION, 40TH AND MORGAN	CHICAGO	IL	5/27/1965
HARTFORD INSURANCE BUILDING	CHICAGO	IL	10/4/1960
HOME FEDERAL SAVINGS & LOAN ASSOCIATION	CHICAGO	IL	4/21/1962
INSTITUTE OF TECHNOLOGY, STUDENT UNION	CHICAGO	IL	10/3/1960
JOSEPH T. RYERSON, 2608 W. 16TH ST.	CHICAGO	IL	11/2/1959
M. & W. AUTO PARTS	CHICAGO	IL	8/23/1965
MERCANTILE BANK BUILDING	CHICAGO	IL	3/22/1960
NATIONAL CASH REGISTER BUILDING	CHICAGO	IL	11/18/1958
PARKING GARAGE, 333 S. SHERMAN ST.	CHICAGO	IL	8/6/1962
S. LA SALLE BUILDING	CHICAGO	IL	7/15/1964
SCHUESSLER KNITTING MILLS	CHICAGO	IL	6/23/1964
ST. JOSEPH'S HOSPITAL	CHICAGO	IL	5/23/1962
ST. MIL'S HIGH SCHOOL	CHICAGO	IL	8/20/1959
STEVENS ANNEX	CHICAGO	IL	5/2/1960
TURNER CONSTR COMPANY, 230 W. MONROE BUILDING	CHICAGO	IL	1/1/1969
TWO FIRST NATIONAL PLAZA BUILDING	CHICAGO	IL	8/8/1969
UNITED INSURANCE BUILDING, STATE AND WACKER ST.	CHICAGO	IL	1/17/1961
			, ,

UNIVERSITY OF ILLINOIS, NURSING SCHOOL	CHICAGO	ш	7/10/1069
US STEEL, FERROSLAG PROCESSING COMPANY	CHICAGO	IL IL	7/10/1968 12/29/1959
WEST SIDE JUNIOR COLLEGE	CHICAGO	IL	11/25/1969
FOREMAN MANUFACTURING COMPANY	CHICAGO RIDGE	IL	10/22/1964
HONEYWELL PLANT	COOK COUNTY	IL	8/7/1969
LUMBERMAN'S MUTUAL CASUALTY COMPANY	DECATUR	IL	3/31/1967
BEHLEN MANUFACTURING COMPANY	EL PASO	IL	6/20/1962
ST. FRANCIS HOSPITAL	EVANSTON	IL	4/24/1963
WEST MINISTER PRESBYTERIAN HOME	EVANSTON	IL	9/26/1962
LITTLE COMPANY OF MARY HOSPITAL	EVERGREEN PARK	IL	6/26/1967
MAYWOOD PARK TROTTING ASSOCIATION	MAYWOOD	IL	4/1/1965
ZENITH RADIO CORPORATION	NORTH LAKE	IL	6/18/1969
BABSON BROTHERS COMPANY	OAKBROOK	IL	10/1/1964
FREMD HIGH SCHOOL, COOK COUNTY	PALATINE	IL	4/15/1969
KREIGSMAN TRANSFER STORAGE COMPANY	PEKIN	IL	8/6/1964
PEKIN MEMORIAL HOSPITAL	PEKIN	IL	7/6/1964
CENTRAL ILLINOIS LIGHT COMPANY	PEORIA	IL	9/13/1966
GENERAL OFFICE BUILDING	PEORIA	IL	7/8/1966
LE TOURNEAU WESTINGHOUSE, OFFICE BUILDING	PEORIA	IL	6/17/1966
PEORIA SAVINGS & LOAN ASSOCIATION	PEORIA	ΙL	4/6/1964
SECURITY SAVINGS & LOAN	PEORIA	IL	7/30/1965
POWERTON GENERATING STATION	POWERTON	ΙL	1/9/1967
COMMONWEALTH EDISON COMPANY, JOLIET POWER STATION	ROCKDALE	ΙL	8/3/1964
ROCKFORD COLLEGE, FINE ARTS COMPLEX	ROCKFORD	IL	7/23/1969
SWEDISH-AMERICAN HOSPITAL	ROCKFORD	IL	5/8/1969
CHEMPLEX COMPANY, GULF RD. AND HWY 53	ROLLING MEADOWS	IL	5/29/1967
CONTAINER CORPORATION OF AMERICA	TAROL STREAM	IL	8/31/1964
ANDERSON COLLEGE, WARNER AUDITORIUM	ANDERSON	IN	9/21/1964
INDIANA UNIVERSITY, ALUMNI HALL	BLOOMINGTON	IN	12/16/1963
INDIANA UNIVERSITY, GEOLOGICAL SURVEY BUILDING	BLOOMINGTON	IN	5/29/1963
CITY COUNTY JAIL	COLUMBUS	IN	2/28/1962
FEDERAL SAVINGS & LOAN BUILDING	COLUMBUS	IN	 12/21/1959
H.H. ROBERTSON COMPANY	CONNERSVILLE	IN	6/8/1965
WABASH COLLEGE, BAXTER HALL	CRAWFORDSVILLE	IN	10/1/1963
COMMERCIAL INSULATION & ACOUST	ECKHART	IN	12/9/1965
115 E. JACKSON ST.	ELKHART	IN	11/12/1959
ALUMINUM CORPORATION OF AMERICA	ELKHART	IN	8/25/1958
CITY ROOFING COMPANY, 115 E. JACKSON ST.	ELKHART	IN	2/12/1958
GOOD SAMARITAN HOSPITAL	ELKHART	IN	11/20/1969
LIMA HOSPITAL JOB	ELKHART	IN	3/15/1968
PENTA COUNTY VOCATIONAL HIGH SCHOOL	ELKHART	IN	10/26/1965
PURDUE UNIVERSITY, REGIONAL CAMPUS	ELKHART	IN	9/24/1971
KENTUCKY UTILITIES, GREEN RIVER, TURBINES	EVANSVILLE	IN	4/9/1970
LOUISVILLE INSULATION & SUPPLY, 509 N. 9TH AVE.	EVANSVILLE	IN	1/27/1971
TRITON CENTRAL SCHOOL	FAIRLAND	IN	3/9/1962
CITY COUNTY COURT HOUSE BUILDING	FORT WAYNE	IN	6/25/1969
COMMERCIAL INSULATION & ACOUST	FORT WAYNE	IN	7/8/1969

MIDWEST UNITED INCLIDANCE COMPANY DUILDING	FORT MANAGE		40/5/4063
MIDWEST-UNITED INSURANCE COMPANY BUILDING	FORT WAYNE	IN	10/5/1962
ATLAS MUTUAL LIFE INSURANCE COMPANY	INDIANAPOLIS	IN	8/31/1959
BUTLER UNIVERSITY , WOMENS RESIDENCE HALL	INDIANAPOLIS	IN	7/1/1963
CHRISTIAN CHURCH, 38TH ST.	INDIANAPOLIS	IN	5/19/1966
COMMERCIAL FLOOR COVERING, CHRYSLER CORPORATION	INDIANAPOLIS	IN	8/13/1962
ENGLEWOOD CHRISTIAN CHURCH	INDIANAPOLIS	IN	10/9/1961
ESSEX HOUSE	INDIANAPOLIS	IN	8/6/1965
FARMERS MUTUAL INSURANCE COMPANY	INDIANAPOLIS	IN	5/21/1970
GREYHOUND BUS TERMINAL	INDIANAPOLIS	IN	7/17/1965
INDIANA TRUST RUIL DING	INDIANAPOLIS	IN	8/8/1969
INDIANA TRUST BUILDING	INDIANAPOLIS	IN	8/31/1967
MARION COUNTY GENERAL HOSPITAL	INDIANAPOLIS	IN	4/18/1966
MARION COUNTY JAIL	INDIANAPOLIS	IN	2/1/1965
OWENS-CORNING SUPPLY CORPORATION	INDIANAPOLIS	IN	2/23/1965
REILLEY TAR & CHEMICAL, LAB BUILDING WOMEN'S RESIDENTIAL HALL "B"	INDIANAPOLIS	IN	3/23/1966
SEAGRAMS BOTTLING PLANT	LANADENICEDIDO	IN	7/8/1963
	LAWRENCEBURG	IN	8/7/1964
LOGANSPORT HIGH SCHOOL	LOGANSPORT MADISON	IN	10/19/1971
WILLIAMS CONSTRUCTION COMPANY, 337 CLIFFTY DR. ROGER HIGH SCHOOL		IN	10/13/1961
EDUCATION & GENERAL CLASSROOM BUILDING	MICHIGAN CITY	IN	2/19/1971 8/31/1967
STUDEBAKER RESIDENCE HALL	MUNCIE MUNCIE	IN	4/8/1964
NORTH CENTRAL HIGH SCHOOL	NORA	IN IN	
	SOUTH BEND	IN	10/1/1962 7/2/1962
ASSOCIATED INVESTMENT ADDITION, RUSKIN ST. NOTRE DAME UNIVERSITY	SOUTH BEND	IN	8/19/1970
STATE TEACHERS COLLEGE, W. RESIDENT HALL	TERRE HAUTE	IN	1/18/1965
SAMARITAN HOSPITAL	VINCENNES	IN	6/3/1969
AMERICAN OIL COMPANY, GATE #4	WHITING	IN	11/20/1968
RETARDED CHILDRENS CENTER	EMPORIA	KS	12/9/1964
NATIONAL BANK BUILDING	GREAT BEND	KS	8/24/1962
FINE ARTS CENTER	IOLA	KS	11/26/1963
AIR TRAFFIC CONTROL CENTER, RT. 150	OLATHE	KS	12/20/1960
CIVIC & CULTURE CENTER	TOPEKA	KS	7/29/1967
HALSEY-TEVIS, INCORPORATED, 725 HANCOCK	TOPEKA	KS	10/28/1968
300 GARVEY CENTER BUILDING	WICHITA	KS	7/19/1965
C.F. & Y. BUILDING, KELLOG & WEBB RD.	WICHITA	KS	4/4/1960
CIVIC & CULTURE CENTER	WICHITA	KS	7/29/1967
HALSEY-TEVIS, INCORPORATED, 306 S. MOSLEY	WICHITA	KS	9/15/1965
K.G. & E. BUILDING, 1ST AND MARKET STS.	WICHITA	KS	7/18/1969
EAST KENTUCKY RECC, COOPER STATION	BURNSIDE	KY	7/7/1969
KENTUCKY STATE OFFICE BUILDING	FRANKFORT	KY	7/2/1964
1ST FEDERAL SAVINGS & LOAN BRANCH OFFICE	FRANKLIN	KY	9/3/1963
UNIVERSITY OF KENTUCKY, HOSPITAL & CLINIC BUILDING	LEXINGTON	KY	9/19/1960
LOUISVILLE GAS & ELECTRIC, CANE RUN STATION	LOUISVILLE	KY	1/4/1966
LOUISVILLE INSULATION & SUPPLY	LOUISVILLE	KY	11/11/1963
OWENS-CORNING FIBERGLAS	LOUISVILLE	KY	2/23/1965
THE 800 BUILDING, FOURTH AND YORK ST.	LOUISVILLE	KY	10/12/1962
,			, , ====

TEVAC CAC TRANSPARCION CONTRANY HOME OFFICE	OWENCDODO	107	C /4 O /4 O C4
TEXAS GAS TRANSMISSION COMPANY, HOME OFFICE	OWENSBORO	KY	6/19/1961
EASTERN STATE COLLEGE, ALUMNI COLISEUM	RICHMOND	KY	11/20/1962
EASTERN STATE COLLEGE, LABORATORY SCHOOL	RICHMOND	KY	7/17/1961
BOLTON COMPANY CAMPONITIONAL TRACK	BATON ROUGE	LA	11/1/1967
CAMBON TEAM TRACK	BATON ROUGE	LA	1/1/1967
DEPARTMENT OF EDUCATION BUILDING	BATON ROUGE	LA	8/30/1967
MCCARTY BRANTON, INCORPORATED, SPUR TRACTOR RD.	BATON ROUGE	LA	7/30/1965
STATE OF LOUISIANA EDUCATION BUILDING	BATON ROUGE	LA	9/7/1967
OAKWOOD SHOPPING CENTER	GRETNA	LA	5/31/1966
NORTHSIDE HIGH SCHOOL GYMNASIUM	LAFAYETTE	LA	7/17/1967
NEW ORLEANS AIRPORT, MOISANT INTERNATIONAL AIRPORT	MONNISANT	LA	3/23/1959
ALLEN TOWERS, 225 BARRONE ST.	NEW ORLEANS	LA	5/15/1961
BELOU & COMPANY	NEW ORLEANS	LA	12/13/1965
CAMBON TEAM TRACK, BELOU & COMPANY	NEW ORLEANS	LA	11/1/1967
CIVIC CENTER BUILDING, GRAVIER ST.	NEW ORLEANS	LA	4/20/1962
CRANE CONSTR COMPANY, 225 BARRONE ST.	NEW ORLEANS	LA	4/18/1961
FEDERAL POST OFFICE BUILDING	NEW ORLEANS	LA	6/20/1960
G.F. FAURAT, 731 ST. CHARLES	NEW ORLEANS	LA	4/22/1966
HAASE CONSTRUCTION COMPANY, 2800 LAFITTE ST.	NEW ORLEANS	LA	1/27/1959
HOTEL AMERICA	NEW ORLEANS	LA	6/10/1968
INTERNATIONAL TRADE MART BUILDING	NEW ORLEANS	LA	11/6/1965
KING & COMPANY	NEW ORLEANS	LA	3/6/1959
L. D. REEDER COMPANY	NEW ORLEANS	LA	1/1/1960
LOUISIANA STATE UNIVERSITY LIBRARY BUILDING	NEW ORLEANS	LA	9/25/1963
MCCARTY BRANTON, INCORPORATED	NEW ORLEANS	LA	10/27/1965
ROYAL ORLEANS HOTEL	NEW ORLEANS	LA	5/19/1964
TULANE LIBRARY	NEW ORLEANS	LA	10/20/1967
JOHNSON ACOUSTICS, INCORPORATED	AMHERST	MA	3/8/1966
UNIVERSITY OF MASSACHUSETTS, GRADUATE RESEARCH CENTER	AMHERST	MA	3/26/1969
MANGANARO BROTHERS, INCORPORATED, 221 CONCORD ST.	BELMONT		1/23/1970
BLACKSTONE-MILLVILLE REGIONAL SCHOOL	BLACKSTONE	MA	3/19/1970
BATESON CONSTRUCTION	BOSTON	MA	1/1/1966
BOSTON CITY HALL	BOSTON	MA	9/6/1966
BOSTON UNIVERSITY, NURSING SCHOOL	BOSTON	MA	6/16/1967
CHARLES RIVER PLAZA OFFICE BUILDING	BOSTON	MA	3/6/1967
CHASE BUILDING	BOSTON	MA	6/28/1963
CITY HALL	BOSTON	MA	1/1/1966
COUNTWAY LIBRARY, HARVARD UNIVERSITY MEDICINE	BOSTON	MA	9/15/1964
DEACONESS HOSPITAL	BOSTON	MA	5/20/1969
EDWARD F. BYRNES CORPORATION, 1111 STATLER OFFICE BUILDING	BOSTON	MA	12/14/1970
EDWARD F. BYRNES CORPORATION, 183 ESSEX ST.	BOSTON	MA	2/9/1966
EDWARD F. BYRNES CORPORATION, 250 STUART ST.	BOSTON	MA	2/11/1959
FEDERAL OFFICE BUILDING, GOVERNMENT CENTER	BOSTON	MA	5/17/1965
FILENES DEPARTMENT STORE	BOSTON	MA	11/2/1970
IBM BUILDING	BOSTON	MA	6/3/1958
KEYSTONE BUILDING, 225 CONGRESS ST.	BOSTON	MA	2/27/1970
LIBERTY MUTUAL BUILDING	BOSTON	MA	4/6/1959

LORD & TAYLOR STORE, PRUDENTIAL CENTER	BOSTON	MA 8/2/1967
M.B.T.A., HIGH ST.	BOSTON	MA 11/19/1969
MANGANARO BROTHERS, INC.	BOSTON	MA 1/1/1969
MEDI-CENTER, HARRISON AVE.	BOSTON	MA 11/16/1967
MENTAL HEALTH CENTER, GOVERNMENT CENTER	BOSTON	MA 7/5/1968
MUSEUM OF SCIENCE, SCIENCE PARK	BOSTON	MA 6/24/1960
NEW ENGLAND INSULATION COMPANY	BOSTON	MA 1/2/1963
NEW ENGLAND MUTUAL LIFE INSURANCE BUILDING	BOSTON	MA 8/8/1961
ONE CENTER PLAZA	BOSTON	MA 6/21/1965
		· ·
PRUDENTIAL TOWER, HUNTINGTON AVE.	BOSTON	MA 2/6/1970
STATE OFFICE BUILDING, CAMBRIDGE ST.	BOSTON	MA 7/7/1964
TRAVELERS INSURANCE BUILDING	BOSTON	MA 12/23/1963
TURNER CONSTRUCTION, HUNTINGTON AVE.	BOSTON	MA 5/29/1970
UNEMPLOYMENT SECURITY BUILDING, GOVERNMENT CENTER	BOSTON	MA 4/18/1968
BLUE HILL SPORTS ARENA	BRAINTREE	MA 9/28/1965
BROCKTON HIGH SCHOOL	BROCKTON	MA 9/20/1968
CARDINAL CUSHING HOSPITAL	BROCKTON	MA 6/22/1967
ST. NICHOLAS CHURCH	BROCKTON	MA 10/23/1963
GROSSMAN'S	BROOKLINE	MA 4/25/1969
HEARTHSTONE PLAZA, RT. 9	BROOKLINE	MA 7/24/1970
ACOUSTICAL FIREPROOFING CONTRACTORS	CAMBRIDGE	MA 12/1/1971
BADGER BUILDING, CAMBRIDGE ENTERPRISES	CAMBRIDGE	MA 3/24/1969
CENTER FOR ADVANCE ENGR STUDY	CAMBRIDGE	MA 1/1/1967
		MA 1/9/1959
DILLABY FIREPROOFING COMPANY, 225 BENT ST.	CAMBRIDGE	
EDWARD F. BYRNES CORPORATION, 200 BENT ST.	CAMBRIDGE	MA 3/18/1963
JAMES H. BOYLE & SONS, INCORPORATED	CAMBRIDGE	MA 10/29/1964
MATHER HOUSE, FLAG ST.	CAMBRIDGE	MA 4/15/1969
MIDDLESEX COUNTY COURTHOUSE	CAMBRIDGE	MA 8/6/1969
CHARLES-NEWTON APARTMENTS	CHARLESTOWN	MA 10/15/1970
WARD SIX SCHOOL, HOLLY ST.	CHICOPEE FALLS	MA 11/2/1962
MIDDLESEX INSURANCE COMPANY BUILDING	CONCORD	MA 1/26/1968
1232 DORCHESTER AVE.	DORCHESTER	MA 12/2/1960
EDWARD F. BYRNES CORPORATION, 1177 ADAMS ST.	DORCHESTER	MA 7/18/1958
ELLSWORTH SUPPLY COMPANY	DORCHESTER	MA 11/30/1961
AVCO RESEARCH LAB	EVERETT	MA 3/19/1969
FITCHBURG JUNIOR HIGH SCHOOL	FITCHBURG	MA 3/23/1967
V. PELLETEIRI & SONS, INCORPORATED, 109 RIVER ST.	FITCHBURG	MA 12/19/1966
BOWLING ALLEYS, BUILDING P-654	FORT DEVENS	MA 11/25/1957
NEW ENGLAND TELEPHONE COMPANY, HEADQUARTERS BUILDING	FRAMINGHAM	MA 1/26/1965
ATLANTIC UNION COLLEGE, LIBRARY	LANCASTER	
		MA 3/12/1969
MUNICIPAL PARK GARAGE, COMMON ST.	LAWRENCE	MA 4/23/1970
HENRY J. ROBINSON SCHOOL	LOWELL	MA 10/7/1968
NUCLEAR CENTER, LOWELL TECH	LOWELL	MA 6/13/1969
LOUIE GRANDE & SONS, 100 CHARLES ST.	MALDEN	MA 6/13/1967
MANGANARO BROTHERS, INCORPORATED, 100 CHARLES ST.	MALDEN	MA 1967
MARLBORO COURTHOUSE	MARLBORO	MA 6/16/1969
BONSECORUS HOSPITAL	METHUEN	MA 7/28/1971

NEW TON HIGH SOLION	N.5.4.(TON). ((1.1.5)		10/00/1000
NEWTON HIGH SCHOOL	NEWTONVILLE		12/22/1970
ESSEX TRUST BANK BUILDING	PEABODY		9/30/1971
BERKSHIRE COMMONS	PITTSFIELD		10/31/1969
BERKSHIRE LIFE INSURANCE BUILDING	PITTSFIELD		11/20/1958
PITTSFIELD BOYS CLUB	PITTSFIELD		6/4/1962
PITTSFIELD IRON WORKS	PITTSFIELD		4/2/1963
TEMPLE ANSHE AMONIN	PITTSFIELD		3/10/1964
JORDAN MARSH	QUINCY		11/5/1969
MCGRATH STORAGE & WITHHOLDING	SOMERVILLE		11/25/1957
PLUMBING SALES COMPANY	SOMERVILLE		8/31/1961
SOUTH SENIOR HIGH SCHOOL	SOUTH WEYMOUTH		7/9/1969
SPRINGFIELD HOBBY CLUB	SPRINGFIELD		7/14/1970
ST. PATRICKS CHURCH, ALLEN ST.	SPRINGFIELD		7/2/1963
TWO GUYS, VORNADO INCORPORATED	SPRINGFIELD		6/28/1966
YMCA BUILDING	SPRINGFIELD		10/13/1967
BRANDEIS UNIVERSITY	WALTHAM		5/3/1971
POLAROID BUILDING #4	WALTHAM		6/3/1970
SENIOR HIGH SCHOOL , LEXINGTON ST.	WALTHAM		10/15/1968
WATERTOWN ARSENAL, 463 ARSENAL ST.	WATERTOWN		7/3/1969
WELLESLEY JUNIOR HIGH SCHOOL	WELLESLEY		3/2/1967
LA MONT ASSOCIATES, INCORPORATED	WEST NEWTON		5/31/1968
JOHNSON ACOUSTICS, INCORPORATED, WESTERN AVE.	WEST SPRINGFIELD		3/25/1959
JOHNSON ASBESTOS DISTRIBUTORS	WEST SPRINGFIELD		12/20/1963
WESTERN MASSACHUSETTS ELECTRIC BUILDING	WEST SPRINGFIELD	MA	
ROBERT L. SHALLOW, COMMERCIAL ST.	WEYMOUTH		1/5/1966
TUFFS LIBRARY	WEYMOUTH LANDING	MA	
FREIGHTWAYS MOTOR TRANSPORT	WOBURN		10/16/1963
CIVIC CENTER	WORCESTER	MA	
GUARANTEED BANK, MAIN ST.	WORCESTER		4/3/1970
MECHANICS BANK, WORCESTER CENTER	WORCESTER		4/14/1970
PAUL REVERE INSURANCE BUILDING	WORCESTER		12/22/1967
HOLY NAME CENTRAL HIGH SCHOOL	WORCHESTER		9/20/1966
INDUSTRIAL TECHNICAL INSTITUTE	WORCHESTER		2/9/1965
CARSON CEILINGS & WALLS, INC.	BALTIMORE		1/1/1969
CENTRAL LAB & OFFICE BUILDING	BALTIMORE		10/17/1972
FEDERAL OFFICE BUILDING, LOMBARD AND HANOVER	BALTIMORE		1/15/1965
GAS & ELECTRIC COMPANY, FAYETTE AND LIBERTY	BALTIMORE		12/22/1964
HAMPSHIRE CORPORATION	BALTIMORE		12/9/1957
JOHNS HOPKINS HOSPITAL	BALTIMORE		1/1/1961
JOHNS HOPKINS UNIVERSITY, ADMINISTRATION BUILDING	BALTIMORE		6/1/1970
KORVETTES DEPARTMENT STORE	BALTIMORE		11/6/1965
MCCORMICK ASBESTOS COMPANY	BALTIMORE		6/15/1966
OWENS-CORNING SUPPLY CORPORATION	BALTIMORE		2/23/1965
UNIVERSITY OF MARYLAND, SCHOOL OF LAW	BALTIMORE		11/16/1964
STEBBING COMPANY	BALTIMORE		12/2/1957
UNION TRUST COMPANY	BALTIMORE		6/24/1968
UNIVERSITY OF MARYLAND, SCHOOL OF DENTISTRY	BALTIMORE	MD	10/31/1968

LININ/EDGITY OF MARDYLAND, COLLOCK OF NURSING	DALTIMODE	NAD	4/2/4060
UNIVERSITY OF MARYLAND, SCHOOL OF NURSING WALLACE ASBESTOS, INCORPORATED	BALTIMORE BALTIMORE		4/2/1969
200 ADDISON RD.	BEAVER HEIGHTS		9/17/1963 11/27/1967
	BLADENSBURG		1/6/1959
HAMPSHIRE CORPORATION, 4626 ANNAPOLIS RD. DORCHESTER COUNTY COURTHOUSE	CAMBRIDGE		4/11/1968
STEBBING COMPANY	CHESTERTOWN		4/11/1968
C. & D. TELEPHONE COMPANY DATA PROCESSING CENTER	COCKEYSVILLE COCKEYSVILLE		8/19/1963
NOXZEMA CHEMICAL COMPANY	COCKEYSVILLE		6/7/1963
			10/28/1965
UNIVERSITY OF MARYLAND, STUDENT UNION BUILDING	COLLEGE PARK		6/27/1962
E.I. DUPONT DE NEMOURS & COMPANY	CURTIS BAY		5/24/1965
STEBBING COMPANY, 1631 FILBERT ST.	CURTIS BAY		5/10/1968
WASHINGTON TOWERS	GAITHERSBURG		8/11/1965
HOWARD COUNTY OFFICE BUILDING	HOWARD COUNTY		12/9/1966
LA PLATA HOSPITAL	LA PLATA		2/22/1963
BIRD BANDING RECORD CENTER	LAUREL		2/8/1968
HARRY SCHNEIDER	LUTHERVILLE		5/28/1964
FAIRLAND DATA CENTER	MONTGOMERY COUNT		• •
IBM GENERAL OFFICE BUILDING	MONTGOMERY COUNT		· ·
MARYLAND CUP, REISTERSTOWN RD.	OWINGS MILLS		11/24/1965
CAPITOL CAR DIST, 9200 GEORGE PALMER HWY.	PRINCE GEORGE COUN		• •
PERPETUAL BUILDING, ASSOC. OFFICE BUILDING	PRINCE GEORGE COUN		
6600 KENILWORTH AVE.	RIVERDALE		5/28/1964
YMCA BUILDING	ROCKVILLE		1/14/1970
ZADEK MACGRUDER HIGH SCHOOL	ROCKVILLE		1/1/1970
VICTOR-LYNN TERMINAL	SALISBURY		12/19/1963
BURNETT NOBLE CORPORATION	SILVER SPRINGS		1/28/1966
JEFFERSON BUILDING, CHESAPEAKE AND WASHINGTON	TOWSON		4/11/1961
MISSION HELPERS OF THE SACRED HEART	TOWSON		5/2/1963
ORCHARD ICE RINK	TOWSON		7/22/1965
PRINCE GEORGE COUNTY COURT HOUSE	UPPER MARLBORO		4/4/1958
SEARS ROEBUCK STORE	WHITE OAK		9/10/1965
POST OFFICE & FEDERAL OFFICE BUILDING	AUGUSTA		11/1/1965
BRUNSWICK NAVAL AIR STATION, BUILDING 555	BRUNSWICK		11/23/1964
U.S. NAVAL AIR STATION	BRUNSWICK		8/5/1965
FORT KENT HOSPITAL	FORT KENT		6/5/1969
LENAWEE MEDICAL CARE FACILITY	ADRIAN	MI	5/2/1969
AUTOMOTIVE ASSEMBLY, PILOT PLAN	ALLEN PARK	MI	6/5/1967
GRAND VALLEY STATE COLLEGE, LEARNING CENTER	ALLENDALE	MI	5/11/1964
ALPENA SENIOR HIGH SCHOOL	ALPENA	MI	8/31/1966
OFFICE BUILDING , 1500 WOODWARD AVE.	BLOOMFIELD	MI	2/14/1967
FRANKS MOBILE HOMES	BROWN CITY	MI	8/31/1962
MACOMB COUNTY ADMINISTRATION BUILDING	CLINTON	MI	9/5/1967
9335 ST. STEPHENS ST.	DEARBORN	MI	5/14/1962
FORD ASSEMBLY, ROUGE PLANT	DEARBORN	MI	8/3/1970
FORD MOTOR COMPANY	DEARBORN	MI	4/15/1958
GREENFIELD VILLAGE PLAZA	DEARBORN	MI	5/22/1967

METROPOLITAN HOSPITAL	DEARBORN	МІ	1/30/1961
R.E. LEGGETTE COMPANY	DEARBORN	МІ	5/28/1958
VILLAGE PLAZA, MICHIGAN AVE. & OUTER DR.	DEARBORN	МІ	1/1/1967
12891 ARTESIAN AVE.	DETROIT	МІ	11/2/1963
BLUE CROSS BLUE SHIELD BUILDING	DETROIT	МІ	7/29/1969
BUDD COMPANY, BUILDING A, 12141 CHARLEVOIR	DETROIT	МІ	1/5/1970
BURROUGHS CORPORATION, MILWAUKEE AND 3RD AVE.	DETROIT	МІ	7/7/1958
C.L. HOLMES COMPANY	DETROIT	MI	10/4/1963
DETROIT BANK & TRUST COMPANY	DETROIT	MI	5/22/1970
HANNA, ZABRINSKIE & DARON	DETROIT	MI	10/6/1967
HENRY FORD HIGH SCHOOL	DETROIT	МІ	5/14/1962
JENNINGS HOSPITAL ADDITION	DETROIT	MI	12/8/1959
JOHNS-MANVILLE SALES CORPORATION, 201 SCHWEIZER PL.	DETROIT	MI	10/3/1967
METROPOLITAN HOSPITAL	DETROIT	MI	1/16/1961
OFFICE BUILDING, FORT AND 15TH ST.	DETROIT	MI	9/25/1961
OWENS-CORNING FIBERGLAS	DETROIT	MI	2/16/1965
WAYNE UNIVERSITY, BASIC RESEARCH MEDICAL BUILDING	DETROIT	MI	3/21/1969
MICHIGAN STATE UNIVERSITY	EAST LANSING	MI	2/24/1964
MICHIGAN STATE UNIVERSITY, DORMS 2 & 3	EAST LANSING	MI	8/14/1964
MICHIGAN STATE UNIVERSITY, INTERNATIONAL CENTER	EAST LANSING	MI	2/10/1964
MICHIGAN STATE UNIVERSITY, LIBRARY BUILDING	EAST LANSING	MI	10/31/1966
MEADE PAPERS	ESCANABA	MI	4/10/1968
CHEVEROLET NEW PERSONNEL BUILDING	FLINT	MI	6/14/1965
CITIZENS BANK	FLINT	MI	6/13/1966
FLINT ADMINISTRATION BUILDING	FLINT	MI	5/11/1959
FLINT JUNIOR COLLEGE, SCIENCE & TECHNOLOGY BUILDING	FLINT	MI	4/14/1959
GENERAL MOTORS INSTITUTE, CAMPUS CENTER	FLINT	MI	10/18/1967
GENESEE VALLEY SHOPPING CENTER	FLINT	MI	4/2/1969
OWENS-CORNING FIBERGLAS	FLINT	MI	2/16/1965
R.R. DOCK ENCLOSURE, A.C. SPARK PLUG DIVISION	FLINT	MI	5/2/1960
BOUMA TILE & ACOUSTICS COMPANY	GRAND RAPIDS	MI	3/4/1965
GRAND FLOOR & PLASTERING COMPANY	GRAND RAPIDS	MI	7/8/1970
GROSSE POINT HIGH SCHOOL	GROSSE POINT	MI	1/27/1968
C.M. SMYTHE PLASTERING COMPANY	HASLET	MI	7/14/1965
PITTSBURGH FORGING COMPANY, 919 AMUR ST.	JACKSON	MI	9/8/1971
CONSUMERS POWER COMPANY, CORK ST.	KALAMAZOO	MI	12/1/1964
KALAMAZOO ART CENTER	KALAMAZOO	MI	1/12/1961
UPJOHN COMPANY, PORTAGE AVE. PLANT	KALAMAZOO	MI	11/12/1959
COMMERCIAL INSULATION & ACOUST, 401 W. GREEN LAWN	LANSING	MI	10/9/1970
INGRAHAM HOSPITAL/LANSING HOSPITAL, P.H.I.	LANSING	MI	8/31/1970
OLDSMOBILE ADMINISTRATION BUILDING	LANSING	MI	1/6/1965
WATER & LIGHT BUILDING	LANSING	MI	10/31/1960
SERVICE ARTS PLASTERING, INCORPORATED, FARMINGTON RD. AND CAPITAL AV	ELIVONIA	MI	2/9/1968
DOW CHEMICAL BUILDING	MIDLAND	MI	4/3/1968
MONROE HARBOR TERMINAL	MONROE	MI	7/3/1961
C.L. HOLMES COMPANY, 45240 GRAND RIVER AVE.	NOVI	MI	8/13/1971
PONTIAC MEDICAL BUILDING	PONTIAC	MI	7/16/1962

OAKLAND UNIVERSITY, DORMITORY UNIT	ROCHESTER	MI	4/28/1966
FORT SAGINAW MALL JOB	SAGINAW	MI	5/18/1966
SAGINAW NEWS BUILDING	SAGINAW	MI	6/22/1959
GLEN LEVEY JUNIOR HIGH SCHOOL	SOUTHFIELD	MI	5/10/1963
KOPPY-NEMER OFFICE BUILDING	SOUTHFIELD	MI	10/16/1969
MICHIGAN BELL TELEPHONE COMPANY	SOUTHFIELD	MI	1/26/1961
OFFICE BUILDING, 26711 NORTHWESTERN	SOUTHFIELD	MI	7/18/1970
OFFICE BUILDING, EVERGREEN AND 10 1/2 MILL RD.	SOUTHFIELD	MI	6/12/1971
OFFICE BUILDING, NORTHWESTERN HWY.	SOUTHFIELD	MI	3/29/1968
21501 HARPER AVE.	ST. CLAIR SHORES	MI	10/15/1968
FEDERAL DEPARTMENT STORE	WARREN	МІ	6/20/1967
NATIONAL DISPOSAL SERVICE, FLANDERS AVE.	WARREN	МІ	10/17/1968
OFFICE BUILDING, 30500 VAN DYKE	WARREN	MI	4/24/1969
E. MICHIGAN UNIVERSITY, CLASSROOM BUILDING #1	YPSILANTI	MI	10/2/1967
EAST MICHIGAN COLLEGE, PHYSICAL ED BUILDING	YPSILANTI	MI	11/25/1963
GEIGER & HAMME LABORATORIES	YPSILANTI	MI	10/3/1960
LINCOLN SENIOR HIGH SCHOOL	YPSILANTI	MI	5/2/1960
NORTHERN CITY NATIONAL BANK	DULUTH		5/18/1969
PETERSON & HEDE COMPANY	HOPKINS		4/10/1964
MANKATO CLINIC	MANKATO		12/19/1963
INSULATION SALES COMPANY	MINNEAPOLIS		12/13/1303
NORTHERN STATES POWER COMPANY	MINNEAPOLIS		5/16/1964
RIVER TOWER APARTMENTS	MINNEAPOLIS		12/16/1965
POST OFFICE & CUSTOMS, ELEC. & MISSILE FAC.	ST PAUL		12/10/1962
CONCORDIA SEMINARY LIBRARY	CLAYTON		11/20/1961
RIVER ROADS SHOPPING CENTER	JENNINGS		1/19/1962
HENGES COMPANY, INCORPORATED, 2531 INDIANA AVE.	KANSAS CITY		12/2/1957
HENGES COMPANY, INCORPORATED, 3007 E. 85TH ST.	KANSAS CITY		4/19/1960
KANSAS CITY NATURAL SLATE COMPANY, 3051 SOUTHWEST BLVD.	KANSAS CITY		5/13/1965
KANSAS CITY NATURAL SLATE COMPANY, 5301 EAST 9TH ST.	KANSAS CITY		9/20/1965
KROGER STORE, 87TH AND WARD PKWY.	KANSAS CITY		4/20/1959
OWENS-CORNING SUPPLY CORPORATION	KANSAS CITY		2/23/1965
STANDARD ASBESTOS MANUFACTURING	KANSAS CITY		4/13/1966
WARD PARKWAY SHOPS, UNIT E-8	KANSAS CITY	MO	2/13/1962
GLAZE CONSTRUCTION COMPANY, HOLLISTER, INCORPORATED	KIRKSVILLE	MO	9/14/1964
1100 SOUTH JEFFERSON ST.	ST. LOUIS	MO	10/29/1960
ALCOA CUPPLES PRODUCTS, WORLD TRADE CENTER	ST. LOUIS	MO	7/7/1967
GRANCO STEEL PRODUCTS, 6506 NORTH BROADWAY	ST. LOUIS	MO	9/6/1966
OFFICE BUILDING, 4236 LINDELL BLVD.	ST. LOUIS	MO	3/27/1958
MITCHELL ENGINEERING COMPANY	COLUMBUS	MS	1/15/1959
NORTH BROTHERS, INCORPORATED, 950 S. GALLATIN ST.	JACKSON	MS	3/23/1971
ANACONDA COMPANY	ANACONDA	MT	6/27/1968
ANACONDA FOREST PRODUCTS	BONNER	MT	5/1/1970
ANACONDA COMPANY	BUTTE	MT	10/10/1966
WESTERN RANCH SUPPLY COMPANY, FRONT AND 14TH STS.	HELENA	МТ	6/3/1958
WALDORF-HOERNER, PAPER PRODUCTS DIVISION	MISSOULA	МТ	8/31/1966
ACOUSTICS, INCORPORATED, 2501 SOUTH BLVD.	CHARLOTTE	NC	9/26/1961
,		-	, ,

ACOUSTI-ENGINEERING COMPANY	CHARLOTTE		1/27/1958
AMERICAN COMMERCIAL BANK BUILDING	CHARLOTTE	NC	9/15/1960
COULWOOD JUNIOR HIGH SCHOOL	CHARLOTTE	NC	7/3/1962
METHODIST HOME, 3420 SHAMROCK DRIVE	CHARLOTTE	NC	1/14/1963
HOME SECURITY LIFE INSURANCE COMPANY	DURHAM	NC	11/14/1957
NORTHERN HIGH SCHOOL	DURHAM	NC	3/27/1958
ELKINS PUBLIC LIBRARY	ELKINS	NC	7/23/1968
CUMMINGS SENIOR HIGH SCHOOL	GRAHAM	NC	11/18/1969
FITE & WARMATH, P. LORILLARD CORPORATION	GREENSBORO	NC	5/26/1969
FOUR SEASONS OFFICE BUILDING, PINE CROFT RD.	GREENSBORO	NC	9/3/1968
GREENSBORO COLLEGE, PHYSICAL EDUCATION BUILDING	GREENSBORO	NC	7/21/1960
NATIONAL GUARD ARMORY	GREENSBORO	NC	8/16/1961
U.N.C. NURSES EDUCATION BUILDING, MARTINSVILLE HOSPITAL	GREENSBORO	NC	1/27/1969
U.S. POST OFFICE BUILDING, W. MARKET ST.	GREENSBORO		8/25/1965
WESTERN ELECTRIC COMPANY, GUILFOFD CENTER	GREENSBORO	NC	11/20/1969
NORTHWESTERN BANK BUILDING	HICKORY	NC	1/25/1967
HEADQUARTERS FIRE STATION	HIGH POINT	NC	1/24/1967
NATIONAL FURNITURE MART, S. MAIN ST.	HIGH POINT	NC	6/19/1964
SOUTHERN FURNITURE EXPOSITION BUILDING	HIGH POINT	NC	1/24/1967
FIRST BAPTIST CHURCH	MORGANTON	NC	1/1/1966
VETERANS HOSPITAL	OTEEN	NC	1/1/1966
WEYERHAUSER COMPANY	PLYMOUTH	NC	4/26/1966
K. & W. CAFETERIA, 512 DANIELS ST.	RALEIGH	NC	7/8/1968
W.H. WERNER CONSTRUCTION COMPANY	RALEIGH	NC	8/4/1966
NORTHWESTERN DISTRICT HOSPITAL	ROANOKE RAPIDS	NC	7/19/1971
VANCE MOTOR HOTEL	STATESVILLE	NC	5/10/1962
F.L. SHOWALTER, INCORPORATED	SWEPSONVILLE	NC	1/26/1970
TARBORO SCHOOL	TARBORO	NC	7/21/1960
U.S. POST OFFICE BUILDING	WILKESBORO	NC	10/18/1968
GLAZIER CONSTRUCTION COMPANY	WILMINGTON	NC	8/7/1959
BOWMAN GRAY SCHOOL OF MEDICINE	WINSTON-SALEM	NC	5/17/1968
GOODWILL REHAB CENTER	WINSTON-SALEM	NC	3/31/1964
MEDICAL PARK HOSPITAL	WINSTON-SALEM		4/6/1970
NATIONAL FURNITURE MART, S. MAIN ST.	WINSTON-SALEM	NC	1/1/1964
PIEDMONT AIRLINES, N. LIBERTY ST.	WINSTON-SALEM	NC	9/11/1967
R.J. REYNOLDS TOBACCO, BUILDING #24	WINSTON-SALEM	NC	10/7/1963
R.J. REYNOLDS TOBACCO, COMPRESSOR ROOM	WINSTON-SALEM	NC	3/22/1962
R.J. REYNOLDS TOBACCO, VINE ST.	WINSTON-SALEM	NC	6/6/1960
REYNOLDS BUILDING #91, VINE ST.	WINSTON-SALEM	NC	1/1/1960
·			• •
REYNOLDS BUSINESS CENTER	WINSTON-SALEM	NC	10/3/1960
REYNOLDS MEMORIAL HOSPITAL, 1101 E. 7TH ST.	WINSTON-SALEM	NC	3/1/1968
SALEM COLLEGE FINE ARTS BUILDING	WINSTON-SALEM	NC	9/15/1964
SCHLITZ BREWING COMPANY	WINSTON-SALEM	NC	4/21/1969
SECURITY LIFE & TRUST COMPANY	WINSTON-SALEM	NC	3/13/1962
SHIELDS, INC., U.S. POST OFFICE	WINSTON-SALEM	NC	1/1/1966
SHIELDS, INCORPORATED	WINSTON-SALEM	NC	3/18/1960
WACHOVIA BANK BUILDING, N. MAIN AND 3RD STS.	WINSTON-SALEM	NC	1/25/1965

WAYE FOREST COULEDS CHARE!	NAME TO N. CALEDA		= /2.4 / 4.0.5.4
WAKE FOREST COLLEGE CHAPEL	WINSTON-SALEM	NC	7/31/1964
WAKE FOREST COLLEGE, RIFLE RANGE	WINSTON-SALEM	NC	•
WINTERVILLE SCHOOL	WINTERVILLE	NC	•
CHRIS'S HATCHERY	BEATRICE	NE	10/24/1958
BEHLEN MANUFACTURING COMPANY	COLUMBUS	NE	12/19/1957
NEW MUNICIPAL AIRPORT	LINCOLN	NE	11/1/1963
FORMAN BROTHERS	OMAHA	NE	11/5/1970
UNITED BENEFIT LIFE INSURANCE COMPANY	OMAHA	NE	11/11/1968
REGIONAL CENTER	DURHAM	NH	11/20/1968
EXETER ACADEMY, ASSEMBLY HALL	EXETER	NH	5/26/1969
MITTERSILL INN	FRANCONIA	NH	2/19/1968
CENTRAL FACILITIES	HAMPDEN BEACH	NH	5/20/1963
NATIONAL GRANGE INSURANCE	KEENE	NH	4/27/1966
ELEMENTARY SCHOOL, JEWETT ST.	MANCHESTER	NH	2/14/1967
ELEMENTARY SCHOOL, RESERVOIR AVE.	MANCHESTER	NH	3/14/1967
HILLSBORO COURT HOUSE	MANCHESTER	NH	
TELEPHONE BUILDING, ELM ST.	MANCHESTER	NH	
18 HARBOR AVE.	NASHUA	NH	
FRANCIS P CONNOR INC, 18 HARBOR AVE.	NASHUA	NH	
FRANCIS P. CONNOR, INCORPORATED	NASHUA	NH	1/12/1966
ELEMENTARY SCHOOL	PLYMOUTH	NH	
FEDERAL POST OFFICE & OFFICE BUILDING	PORTSMOUTH	NH	
FRANKLIN PIERCE COLLEGE	RINDGE	NH	3/18/1968
L.M. PIKE & SONS	TILTON	NH	7/17/1964
CARLTON HOUSE, 510 DEAL LAKE RD.	ASBURY PARK	NJ	9/24/1963
MILLER MASONS	ATLANTIC CITY	NJ	4/4/1962
NEW CITY HALL BUILDING	ATLANTIC CITY	NJ	8/21/1967
SUSSEX BUILDING SUPPLY COMPANY	AUGUSTA	NJ	11/5/1965
BAYONNE PLUMBING SUPPLY	BAYONNE	NJ	9/10/1963
MILITARY OCEAN TERMNAL, NAVAL SUPPLY CENTER	BAYONNE	NJ	8/14/1967
A. WILSON & COMPANY	BELLEVILLE	NJ	
OCF BERLIN KAYLO PLANT	BERLIN	NJ	1/19/1967
SALVATORIAN FATHERS SCHOOL/FACULTY BUILDING	BLACKWOOD	NJ	6/15/1959
CEDAR KNOLLS ACOUSTICAL LAB	CEDAR KNOLLS	NJ	4/7/1967
AUTOMATIC-DATA PROCESSING, RT. 3	CLIFTON	NJ	12/18/1967
COUNTRY CLUB TOWERS	CLIFTON	NJ	10/16/1963
HOFFMAN LA ROCHE, RT. 3	CLIFTON	NJ	5/4/1964
MILES CHEMICAL, PAKAMAINE PLANT	CLIFTON	NJ	3/3/1965
WOOLSULATE CORPORATION	CLIFTON	NJ	10/14/1963
HOWMET CORPORATION	DOVER	NJ	1/1/1968
JERSEY ACOUSTICS & FIREPROOF	DOVER	NJ	1/1/1959
MICROCAST RECEIVING, ROY ST.	DOVER	NJ	1/1/1969
NATIONAL METAL HOSE & TUBING COMPANY	DOVER	NJ	8/3/1965
CONTINENTAL BAKING COMPANY	EAST BRUNSWICK	NJ	9/16/1970
AMERICAN SPECIALTY PRODUCTS	EAST ORANGE	NJ	3/27/1961
L.D. REEDER COMPANY, EASTERN DIVISION	EAST ORANGE	NJ	9/21/1962
OFFICE BUILDING, 100 HALSTEAD ST.	EAST ORANGE	NJ	9/16/1964

COLLADEL & DEVED INCORDODATED CO MESCOTT ST	FACT ODANICE	NII	1/16/1067
SCHADEL & BEYER INCORPORATED, 60 WESCOTT ST.	EAST ORANGE EAST ORANGE	NJ	1/16/1967
SCHADEL & BEYER, INCORPORATED, 555 PROSPECT ST. EDISON WATER DEPARTMENT	EDISON	NJ	3/24/1965
	EDISON	NJ	5/25/1965 11/26/1965
MUSCARELLE, MEAD CONTAINER PLANT REVLON, INCORPORATED	EDISON	NJ	8/22/1967
	ELIZABETH	NJ	6/16/1965
ELIZABETHTOWN GAS, WEST JERSEY ST.			
LIPTON BUILDING, BOILER ROOM	ENGLEWOOD CLIFFS FAIRVIEW	NJ	5/6/1963 1/1/1969
FIRECOTE COMPANY, INC.	FAIRVIEW	NJ	1/1/1969
FIRECOTE COMPANY, INC., ALLAN MCKAY KANE ACOUSTICAL COMPANY	FAIRVIEW	NJ	5/19/1961
FINDERNE, LAULER	FINDERNE	NJ	1/1/1964
JOHNS-MANVILLE SALES CORPORATION, RESEARCH LAB	FINDERNE	NJ	1/1/1964
R.D. WOOD COMPANY	FLORENCE	NJ	6/30/1964
OYSTER CREEK NUCLEAR POWER PLANT	FORKED RIVER	NJ	1/31/1968
E.I. DUPONT DE NEMOURS & COMPANY, REPAUNTO WORKS	GIBBSTOWN	NJ	5/21/1964
THE RECORD, RIVER ST.	HACKENSACK	NJ	10/21/1964
GENNER BROS OF NEW JERSEY	HACKETTSTOWN	NJ	8/23/1962
HACKETTSTOWN ELEMENTARY SCHOOL	HACKETTSTOWN	NJ	9/2/1965
OWENS-CORNING SUPPLY CORPORATION	HARRISON	NJ	2/7/1968
MCGRAW-HILL BUILDING, RT. 571	HIGHTSTOWN	NJ	1/1/1962
MCGRAW-HILL PUBLISHING COMPANY BUILDING, RT. 571	HIGHTSTOWN	NJ	5/18/1962
SHOPPING CENTER, HOMER LEE AND HOPATCONG RDS.	HOPATCONG	NJ	2/15/1965
METROPOLITAN REFRACTORIES CORPORATION	KEARNY	NJ	7/24/1961
TOMPKINS TIDEWATER TERMINAL	KEARNY	NJ	2/27/1961
NAVAL AIR STATION, COMMISSARY BUILDING	LAKEHURST	NJ	2/12/1960
PRINCETON PRESS, LAWRENCEVILLE RD.	LAWRENCE TWP	NJ	8/14/1965
RIDER COLLEGE CAMPUS	LAWRENCEVILLE	NJ	5/5/1966
ALLEGHANY BUILDING UNITS	LONG BRANCH	NJ	5/26/1960
ELECTRONICS ASSOCIATION, INCORPORATED	LONG BRANCH	NJ	3/4/1964
SAYREWOOD INSULATION COMPANY, COTTRELLS RD.	MADISON TOWNSHIP	NJ	1/1/1966
JOHNS-MANVILLE PRODUCTS CORPORATION, RESEARCH DEPARTMENT	MANVILLE	NJ	11/9/1964
PRUDENTIAL INSURANCE COMPANY	MILLVILLE	NJ	12/16/1957
JOHN WANNAMAKER STORE	MOORESTOWN	NJ	1/18/1963
UNION CARBIDE PLASTIC CORPORATION	MORRIS TOWNSHIP	NJ	1/30/1967
MORRIS COUNTY COURTHOUSE	MORRISTOWN	NJ	1/1/1970
DIXON BROTHERS, POCONO RD.	MT. LAKES	NJ	6/4/1958
CITIZENS NATIONAL BANK	NETCONG	NJ	2/24/1964
ST. MICHAELS PAROCHIAL SCHOOL	NETCONG	NJ	8/24/1962
MIDDLESEX COUNTY COURTHOUSE	NEW BRUNSWICK	NJ	9/28/1959
MONMOUTH COUNTY COURTHOUSE PROJECT	NEW BRUNSWICK	NJ	1/1/1960
ASBESTOSPRAY CORPORATION, 300 THOMAS ST.	NEWARK	NJ	1/5/1967
BONDED INSULATION & ALUMINUM	NEWARK	NJ	5/5/1959
ESSEX COUNTY COURTHOUSE	NEWARK	NJ	9/17/1968
FEDERAL OFFICE BUILDING, BROAD AND WALNUT STS.	NEWARK	NJ	8/1/1966
KANE ACOUSTICAL COMPANY	NEWARK	NJ	1/1/1966
MANHATTAN WELDING, INCORPORATED	NEWARK	NJ	9/27/1963
MEDICAL CENTER, 131 CLIFFORD ST.	NEWARK	NJ	1/16/1967

OWENS CORNING SUPPLY CORPORATION	NEWARK		10/00/1000
OWENS-CORNING SUPPLY CORPORATION	NEWARK	NJ	10/28/1963
PRUDENTIAL BUILDING, ACADEMY AND HALSEY STS.	NEWARK	NJ	5/14/1964
RIGHT FINISHING COMPANY	NEWARK	NJ	1/1/1958
SPRAYED FIREPROOFING CONTRACTING	NEWARK	NJ	1/1/1964
SPRAYON INSULATION & ACOUSTIC, INCORPORATED	NEWARK	NJ	12/4/1957
STATE OFFICE BUILDING	NEWARK	NJ	11/2/1966
ARMCO BUILDING, DELL AVE.	NORTH BERGEN	NJ	10/4/1963
DIANA STORES	NORTH BERGEN	NJ	9/22/1964
HOFFMAN LA ROCHE, BUILDING #59	NUTLEY	NJ	10/19/1962
HOFFMAN LA ROCHE, KINGSLAND AVE.	NUTLEY	NJ	5/10/1966
SAYREWOOD INSULATION COMPANY	OLD BRIDGE	NJ	9/28/1965
MEDEC REALTY CORPORATION	ORADELL	NJ	9/21/1962
BOWLING ALLEY, RECREATION CENTER	PARAMUS	NJ	7/12/1962
MUNICIPAL BUILDING, RT. 17	PARAMUS	NJ	5/10/1962
PASSAIC COUNTY COURTHOUSE	PATERSON	NJ	5/13/1968
INDUSTRIAL INSULATORS CORPORATION	PERTH AMBOY	NJ	12/28/1960
PHILLIP CAREY MANUFACTURING COMPANY	PERTH AMBOY	NJ	11/7/1966
PRINCETON UNIVERSITY	PRINCETON	NJ	1/1/1963
PRINCETON UNIVERSITY, MUSIC PRACTICE BUILDING	PRINCETON	NJ	4/4/1963
COUNTY COLLEGE OF MORRIS	RANDOLPH	NJ	4/28/1969
JAM INDUSTRIES, INCORPORATED	REIGELSVILLE	NJ	7/16/1964
MORRIS HILLS REGIONAL HIGH SCHOOL	ROCKAWAY	NJ	12/4/1957
FRACO MAINTENANCE CONTRACTING, INCORPORATED	RUTHERFORD	NJ	12/14/1961
KENNEDY COATING COMPANY	RUTHERFORD	NJ	3/2/1964
INSULATION SALES COMPANY	SADDLEBROOK	NJ	1/1/1962
CAFCO MACHINE SHOP	STANHOPE	NJ	1/1/1959
CELLOFOAM BUILDING, USM	STANHOPE	NJ	4/9/1962
CHANGE ROOM & WAREHOUSE, USM	STANHOPE	NJ	4/23/1962
CON EDISON DEMONSTRATION	STANHOPE	NJ	5/5/1967
NEW OFFICE BUILDING, USM	STANHOPE	NJ	7/9/1962
HOLLAND INDUSTRIES	SUCCASUNNA	NJ	1/1/1959
MILLBURN SHOPPING CENTER, BUILDING B	SUMMIT	NJ	4/12/1961
BELL TELEPHONE PROJECT, TEANECK RD.	TEANECK	NJ	1/27/1966
MAHONEY TROAST CONSTRUCTION COMPANY, TEANECK RD.	TEANECK	NJ	1/17/1966
TOMS RIVER CHEMICAL COMPANY	TOMS RIVER	NJ	11/5/1965
BALDWIN-EHRET-HILL, INCORPORATED	TRENTON	NJ	6/7/1967
GEORGE L. MOFFETT COMPANY	TRENTON	NJ	10/3/1963
JAM INDUSTRIES, INCORPORATED, 413 STOKES AVE.	TRENTON	NJ	10/3/1963
LAFAYETTE ARMS APPARMENTS	TRENTON	NJ	9/25/1964
MERCER HOSPITAL	TRENTON	NJ	10/27/1969
NEW JERSEY MANUFACTURERS CASUALTY INSURANCE, OFFICE BUILDING	TRENTON	NJ	3/31/1965
US STEEL PLANT, AMERICAN BRIDGE DIVISION	TRENTON	NJ	10/3/1968
BORDEN METAL PRODUCTS	UNION	NJ	6/15/1959
BYRNE & WEST, INCORPORATED	UNION	NJ	6/20/1963
PLAYBOY CLUB, GREAT GORGE	VERNON	NJ	9/3/1970
AMERICAN CYANAMID BUILDING, KANE ACOUSTIC, BERMAN RD.	WAYNE	NJ	5/18/1961
C.S. WOOD & COMPANY	WEST ORANGE	NJ	9/7/1960
5.5. 1.500 a com / m	WEST OWNINGE	1 113	5, , , 1500

BELL LABS, WESTERN ELECTRIC COMPANY	WHIPPANY	NJ	3/20/1968
ROWE MANUFACTURING COMPANY, TROY HILL RD.	WHIPPANY	NJ	4/4/1962
FLORIDA INSULATION & FIREPROOF	ALBUQUERQUE	NM	1/1/1961
ACE BUILDING & SUPPLY COMPANY	HOBBS	NM	8/7/1964
ACE WELDING SERVICE, 905 NORTH GRIME ST.	HOBBS	NM	2/1/1958
WONDER BUILDING DISPLAY/WYLIES BOWLING ALLEY	HOBBS	NM	11/28/1958
COUNTY FAIR GROUNDS	PORTALES	NM	8/16/1963
APEX PLASTERING COMPANY, VIA U.P. TEAM TRACK	LAS VEGAS	NV	1/23/1968
U.P. TEAM TRACK	LAS VEGAS	NV	9/22/1964
ALBANY MEDICAL CENTER	ALBANY	NY	8/7/1962
ALBANY MEDICAL COLLEGE	ALBANY	NY	6/4/1962
COLLUM ACOUSTICAL CORPORATION, 616 DELAWARE AVE.	ALBANY	NY	11/21/1963
SENATE CHAMBERS BUILDING, STATE CAPITAL	ALBANY	NY	7/12/1963
STATE TEACHERS COLLEGE, NEW DORMITORY	ALBANY	NY	7/17/1959
TRI-CITY INSULATION COMPANY	ALBANY	NY	3/11/1964
GIEGY CHEMICAL CORPORATION, SAW MILL RIVER RD.	ARDSLEY	NY	10/2/1963
ASTORIA STATION, 20th AVE. & 21st ST.	ASTORIA	NY	1/1/1968
			3/18/1964
ASTORIA STATION, CON EDISON COMPANY, BUILDING #136, 20TH AVE. AND 21ST		NY	
HENDERSON-JOHNSON COMPANY	AUBURNE	NY	6/1/1971
MERCY HOSPITAL, HEALTH & REHAB CENTER	AUBURNE	NY	5/7/1971
TREADWAY INN, OAK ST.	BATAVIA	NY	10/26/1960
MACY'S DEPARTMENT STORE	BAY SHORE	NY	10/29/1962
G. & M. HOME INSULATION COMPANY, 2470 N. JERUSALEM RD.	BELLMORE / LONG ISLAN		3/31/1960
BINGHAMTON BOYS CLUB, WASHINGTON ST.	BINGHAMTON	NY	3/25/1959
FIRST CITY NATIONAL BANK, COURT ST.	BINGHAMTON	NY	6/18/1971
HARPER COLLEGE, PHYSICAL SCIENCE BUILDING	BINGHAMTON	NY	12/18/1967
THEODORE ROOSEVELT, BEVIR ST.	BINGHAMTON	NY	6/15/1970
ALLIED ACOUSTICAL COMPANY, BETWEEN 181-182 STS.	BRONX	NY	9/29/1960
COLLETTI & LAGUIDARA, INCORPORATED, 4231 VIREO AVE.	BRONX	NY	10/16/1968
CON EDISON, 134TH ST. AND LOCUST AVE.	BRONX	NY	12/31/1969
CO-OP CITY, CUSTOMERS TRAILER	BRONX	NY	4/16/1970
CO-OP CITY, NEW ENGLAND THRUWAY	BRONX	NY	10/17/1970
CO-OP CITY, SCHOOL	BRONX	NY	8/7/1970
HELL GATE GENERATOR STATION	BRONX	NY	1/31/1969
P.S. 162, 149TH ST. AND ST. ANN'S AVE.	BRONX	NY	3/23/1970
RECEIVEWAY RESTAURANT, BOSTON POST RD.	BRONX	NY	8/7/1959
RICHLEE ACOUSTICS & FIREPROOF, BOSTON POST RD. AND E. CHESTER RD.	BRONX	NY	10/17/1958
RICHLEE ACOUSTICS & FIREPROOF, PELHAM PKWY. AND STILLWELL AVE.	BRONX	NY	7/10/1959
RICHLEE ACOUSTICS, 3181 WESTCHESTER AVE.	BRONX	NY	9/23/1958
SECURITY MUTUAL INSURANCE BUILDING	BRONX	NY	6/3/1965
SOCIAL SECURITY OFFICE BUILDING	BRONX	NY	11/20/1959
GRAMATAN NATIONAL REALTY COMPANY	BRONXVILLE	NY	1/29/1960
111 LIVINGSTON ST.	BROOKLYN	NY	5/29/1970
111 LIVINGSTON ST. 134 MARSHALL ST.	BROOKLYN	NY	2/13/1968
3 HOLMES LN.			
	BROOKLYN	NY	8/6/1964
4801 1ST AVE.	BROOKLYN	NY	2/27/1961
ARISTOCRAFT FIBER SPRAY COMPANY	BROOKLYN	NY	1/17/1966

CHEVROLET	BROOKLYN	NY	8/3/1965
CON EDISON, 1 HUDSON AVE.	BROOKLYN	NY	10/27/1967
CON EDISON, 107 LAWRENCE ST.	BROOKLYN	NY	5/22/1964
CON EDISON, 238 SCHERMERHORN ST.	BROOKLYN	NY	6/19/1964
EMPIRE CHEVROLET AGENCY	BROOKLYN	NY	9/17/1963
EMPIRE-ACE INSULATION MANUFACTURING COMPANY, 1 COZINE AVE.	BROOKLYN	NY	9/28/1964
HAMILTON FEDERAL SAVINGS BANK	BROOKLYN	NY	3/9/1964
LONG ISLAND UNIVERSITY, WILLOUGHBY ST.	BROOKLYN	NY	5/1/1967
MEMORIAL HOME HOSPITAL, 15 ARMON ST.	BROOKLYN	NY	12/2/1965
METROPOLITAN SPRAY, INCORPORATED, GIAMBOI BROTHERS	BROOKLYN	NY	2/12/1959
RELIABLE SAMPLE COMPANY, 30 SANFORD ST.	BROOKLYN	NY	3/10/1964
S.S. MARIA COSTA, PIER 5	BROOKLYN	NY	10/26/1966
SPRAYED FIREPROOFING CONTRACTOR, 2550 ATLANTIC AVE.	BROOKLYN	NY	1968
ST. ALOYSIUS CHURCH	BROOKLYN	NY	9/29/1967
BUFFALO ACOUSTICS, 605 INDIAN CHURCH RD.	BUFFALO	NY	1/22/1960
DAVIS-FETCH & COMPANY, INCORPORATED	BUFFALO	NY	7/7/1958
HUDSON PLASTERING COMPANY	BUFFALO	NY	12/29/1965
KOBACKERS DEPARTMENT STORE	BUFFALO		
		NY	3/14/1966
MAIN PLACE, PEARL AND COURT STS.	BUFFALO	NY	9/22/1967
NEW YORK STATE UNIVERSITY, STUDENT UNION BUILDING	BUFFALO	NY	1/5/1967
POLICE HEADQUARTERS GARAGE	BUFFALO	NY	11/9/1966
STATE UNIVERSITY COLLEGE, LECTURE HALL CENTER	BUFFALO	NY	4/18/1966
TEMPLE BETH ZION, DELAWARE AND BARKER AVE.	BUFFALO	NY	7/21/1965
ROBERT POTTER, RT. 11	CICERO	NY	10/20/1967
COMFORTI & EISELE COMPANY, COLONIE CENTER	COLONIE	NY	7/1/1966
JOHN EISELE COMPANY	COLONIE	NY	5/31/1966
TENANTS BUILDING, COLONIE CENTER	COLONIE	NY	6/9/1966
STATE TEACHERS COLLEGE, LECTURE HALL	CORTLAND	NY	1/11/1966
FOSTER WHEELER CORPORATION	DANSVILLE	NY	12/23/1964
DEWITT SHOPPING TOWN	DEWITT	NY	5/11/1966
CHEMING CANAL BANK, WATER AND BALDWIN STS.	ELMIRA	NY	11/10/1970
CHEMING CANAL TRUST COMPANY, MAIN ST.	ELMIRA	NY	9/21/1970
NEW YORK TELEPHONE BUILDING, CHURCH ST.	ELMIRA	NY	5/26/1971
PARLEY COBURN SCHOOL	ELMIRA	NY	7/28/1960
ST. BONIFICE CHURCH	ELMONT	NY	6/5/1967
BILL GARGANO, INCORPORATED	ENDICOTT	NY	11/5/1964
IBM, BUILDING #14, NORTH ST.	ENDICOTT	NY	11/10/1970
P.S. 219, MAIN ST.	FLUSHING	NY	7/7/1965
EMPIRE-ACE INSULATION MANUFACTURING COMPANY, 8000 COOPER AVE.	GLENDALE	NY	3/23/1961
KING AIRFLOW, INCORPORATED, 88-50 75TH AVE.	GLENDALE	NY	8/17/1964
H.C. BURNER, 265 GREAT NECK RD.	GREAT NECK	NY	1/27/1964
MARIO & DIBONO PLASTERING	GREAT NECK	NY	2/21/1969
S. & N. CATERERS	GREAT NECK	NY	5/3/1962
NEW YORK TELEPHONE BUILDING	GREENBERG	NY	5/26/1960
NEW YORK POWER POOL POWER CONTROL CENTER	GUILDERLAND	NY	4/15/1969
COLLETTI & LAGUIDARA INCORPORATED, ANDERSON HILL RD.	HARRISON	NY	7/14/1969
IBM BUILDING, 1000 WESTCHESTER AVE.	HARRISON	NY	10/2/1967
isiti solesitto, 1000 Westerlettive.		141	10,2,100,

DEDCI COLA DUIL DINC. ANDEDCON LILL DD	HARRICON	N 137	40/20/4060
PEPSI-COLA BUILDING, ANDERSON HILL RD.	HARRISON	NY	10/28/1968
RICHLEE ACOUSTICS & FIREPROOF, CENTRAL AND HARTSDALE AVE.	HARTSDALE	NY	3/28/1958
STATE OFFICE BUILDING, VETERANS MEMORIAL HWY.	HAUPPAUGE	NY	4/10/1972
NATIONAL ASBESTOS COMPANY, 6 LIBERTY ST.	HAWTHORNE	NY	6/5/1965
4 BURNS ST.	HICKSVILLE	NY	8/9/1965
BRUDERLE & LAMPARTER	HICKSVILLE	NY	11/15/1962
A. & P. JOB, RUST ENGINEERING	HORSEHEADS	NY	2/10/1965
CHERRY RIDGE FARMS, RT. 23	HUDSON	NY	10/12/1964
COLUMBUS MEMORIAL HOSPITAL	HUDSON	NY	5/20/1964
MCARTHUR AIRPORT, FEDERAL AVIATION ADMINISTRATION CONTROL CENTER	ISLIP	NY	 7/5/1962
HILLCREST JEWISH CENTER	JAMAICA	NY	6/17/1959
WILSON MEMORIAL HOSPITAL	JOHNSON CITY	NY	12/4/1968
SYRACUSE INSULATION	LIVERPOOL	NY	10/1/1963
JOHN F. KENNEDY AIRPORT, NATIONAL AIRLINES TERMINAL	LONG ISLAND	NY	10/22/1968
·	LONG ISLAND	NY	
KENNEDY AIRPORT, AMERICAN AIRLINES			10/28/1969
51-09 2ND ST.	LONG ISLAND CITY	NY	12/2/1960
E.J. BRANDT, 50-20 25TH ST.	LONG ISLAND CITY	NY	9/23/1966
NEW YORK ARTCRAFTS, INCORPORATED, 51-09 2ND ST.	LONG ISLAND CITY	NY	4/30/1964
CHATEAU TOWERS RESTAURANT	MAMARONECK	NY	5/7/1958
RICHLEE ACOUSTICS & FIREPROOFING, 641 CENTER AVE.	MAMARONECK	NY	11/20/1957
MEADOWBROOK HOSPITAL	MEADOWBROOK	NY	2/21/1969
THE KRAUSS PERIODICALS, RT. 100	MILLWOOD	NY	11/27/1964
RIGHT FINISHING COMPANY, 14 ST. JOHN ST.	MONTICELLO	NY	11/22/1957
ST. JOHNS ATONEMENT SEMINAR	MONTOUR FALLS	NY	11/8/1965
CALDOR DEPARTMENT STORE	MOUNT VERNON	NY	8/11/1965
COLLETTI & LAGUIDARA, INCOPORATED	MOUNT VERNON	NY	12/6/1962
JOHNS-MANVILLE SALES CORPORATION	MOUNT VERNON	NY	4/27/1967
SINAI TEMPLE, 132 CRARY AVE.	MOUNT VERNON	NY	6/25/1959
C.S. BRAUN, MOUNTAIN AVE. AND DIAMOND HILL	MURRAY HILL	NY	2/26/1965
UTICA MUTUAL LIFE INSURANCE COMPANY, RT. 5	NEW HARTFORD	NY	2/27/1970
CARNEY & DEL GUIDICE, INCORPORATED	NEW HYDE PARK		11/11/1963
STATE COLLEGE	NEW PALTZ	NY	6/23/1970
ELDERADO SHORE & YACHT CLUB	NEW ROCHELLE	NY	11/4/1969
INDOOR TENNIS COURT, CLIFF ST.	NEW ROCHELLE	NY	11/5/1965
,			
NATIONAL BANK OF WESTCHESTER, NORTH AVE.	NEW ROCHELLE	NY	6/2/1969
1133 SIXTH AVE.	NEW YORK	NY	1/1/1968
114 BROADWAY	NEW YORK	NY	2/18/1966
120 CHURCH ST.	NEW YORK	NY	3/28/1963
130 JOHN ST.	NEW YORK	NY	4/23/1969
140 BROADWAY AVE.	NEW YORK	NY	12/17/1965
14TH ST. AND EAST RIVER DR.	NEW YORK	NY	11/15/1968
15 BROAD ST.	NEW YORK	NY	1/17/1963
277 PARK AVE.	NEW YORK	NY	3/25/1964
330 MADISON AVE.	NEW YORK	NY	9/12/1963
370 WEST ST.	NEW YORK	NY	9/21/1960
39TH ST. AND 3RD AVE.	NEW YORK	NY	7/8/1970
416 EAST 36TH ST.	NEW YORK	NY	8/22/1958

42ND ST.	NEW YORK	NY	10/24/1962
437 MADISON AVE.	NEW YORK	NY	1/16/1967
50TH ST. AND 3RD AVE.	NEW YORK	NY	9/23/1968
59TH STREET STATION, 850 - 12TH AVE.	NEW YORK	NY	 4/21/1969
600 3RD AVE.	NEW YORK	NY	6/19/1970
600 THIRD AVE. & 39th STREET	NEW YORK	NY	1/1/1970
65TH & CENTRAL PARK WEST	NEW YORK	NY	10/31/1969
708 1st AVE.	NEW YORK	NY	1/1/1958
77 WATER ST.	NEW YORK	NY	4/29/1969
81ST ST. & YORK AVE.	NEW YORK	NY	5/13/1966
850 - 12TH AVE.	NEW YORK	NY	6/5/1968
95 WALL ST.	NEW YORK	NY	5/9/1969
ALLIED PLASTERING COMPANY, LAFAYETTE ST.	NEW YORK	NY	6/9/1966
BANKERS TRUST COMPANY BUILDING, 280 PARK AVE.	NEW YORK	NY	10/11/1961
BANKERS TRUST COMPANY, PARK AVE. & 49th ST.	NEW YORK	NY	1/1/1961
BANKERS TRUST, 43RD AND 6TH AVE.	NEW YORK	NY	2/9/1968
CBS-5, 530 W. 57TH ST.	NEW YORK	NY	6/13/1963
·			
CELANESE BUILDING, 1201 6TH AVE.	NEW YORK	NY	5/5/1972
CHANIN BUILDING, 39TH ST. AND BROADWAY	NEW YORK	NY	8/9/1968
COLLETTI & LAGUIDARA, INC., 1120 AVE. OF THE AMERICAS	NEW YORK	NY	1/1/1961
COLUMBIA PICTURES, 711 5TH AVE.	NEW YORK	NY	8/19/1958
COLUMBIA UNIVERSITY, 120TH ST.	NEW YORK	NY	5/23/1963
COLUMBIA UNIVERSITY, URIS HALL	NEW YORK	NY	6/5/1963
CON EDISON COMPANY	NEW YORK	NY	6/22/1961
CON EDISON COMPANY, 708 - 1st AVE.	NEW YORK	NY	1/1/1958
CON EDISON, 201ST ST. AND 9TH AVE.	NEW YORK	NY	5/24/1967
CON EDISON, 66TH ST. AND WEST END AVE.	NEW YORK	NY	5/21/1964
CON EDISON, 820 EAST 14TH ST.	NEW YORK	NY	8/13/1969
CON EDISON, EXCAVATION WORK, 66TH AND 3RD AVE.	NEW YORK	NY	8/12/1960
CO-OP CITY, BUILDING #6	NEW YORK	NY	6/10/1969
DIESEL, 75 WATER ST.	NEW YORK		4/23/1969
DRYDEN HOTEL, 150 E. 39TH ST.	NEW YORK	NY	8/12/1960
E.H. DAHLY	NEW YORK	NY	9/11/1958
EAST RIVER STATION	NEW YORK	NY	5/10/1968
EMPIRE MUTUAL INSURANCE BUILDING	NEW YORK	NY	5/4/1965
EUCLID CONTRACTION COMPANY	NEW YORK	NY	3/25/1961
FEDERAL OFFICE BUILDING, BROADWAY AVE.	NEW YORK	NY	3/30/1966
FEDERAL OFFICE BUILDING, DAUNE AND WORTH STS.	NEW YORK	NY	4/14/1966
FEDERAL OFFICE BUILDING, LAFAYETTE ST.	NEW YORK	NY	6/10/1966
FORDHAM UNIVERSITY	NEW YORK	NY	5/3/1967
GEORGE A. FULLER COMPANY, 300 MADISON AVE.	NEW YORK	NY	10/26/1962
GIRL SCOUT COUNCIL BUILDING, 335 E. 46TH ST.	NEW YORK	NY	1/21/1966
GREYHOUND TERMINAL, PORT AUTHORITY	NEW YORK	NY	8/5/1963
HARKNESS MEDICAL BUILDING			
	NEW YORK	NY	5/22/1967
HIPPODROME, 1120 AVE. OF AMERICAS	NEW YORK	NY	10/26/1961
HIPPRODROME, 6th AVE. & 43rd ST.	NEW YORK	NY	1/1/1961
IBM PAVILLION, NEW YORK WORLD'S FAIR	NEW YORK	NY	8/13/1963

IDLEWILD AIRPORT, AMERICAN AIRLINES	NEW YORK	NY	5/4/1959
JEWISH BLIND INSTITUTE	NEW YORK	NY	10/17/1969
JOHNS AND WATER STS.	NEW YORK	NY	4/16/1969
JOHNS-MANVILLE INTERNATIONAL	NEW YORK	NY	7/13/1964
KENNEDY AIRPORT, PAN-AM CARGO BUILDING	NEW YORK	NY	1/6/1966
KING TYPOGRAPHIC SERVICE, 544 W. 43RD ST.	NEW YORK	NY	7/12/1963
L. D. REEDER COMPANY, 280 PARK AVE.	NEW YORK	NY	1/1/1962
MARIO & DIBONO PLASTERING	NEW YORK	NY	7/7/1966
MARIO & DIBONO PLASTERING, 1133 SIXTH AVE.	NEW YORK	NY	1/1/1968
MARIO & DIBONO PLASTERING, 600 THIRD AVE. & 39th ST.	NEW YORK	NY	1/1/1970
MARIO & DIBONO PLASTERING, WALL ST.	NEW YORK	NY	1/1/1969
·	NEW YORK		
MCGRAW-HILL JOB, 48TH AVE.		NY	8/13/1970
MCGRAW-HILL JOB, 7TH AVE.	NEW YORK	NY	8/4/1970
NATIONAL MARITIME UNION BUILDING	NEW YORK	NY	10/15/1962
NEW BANK BUILDING, 176 BROADWAY AVE.	NEW YORK	NY	9/19/1967
NEW OFFICE BUILDING, 39th ST. & BROADWAY AVE.	NEW YORK	NY	1/1/1968
NEW YORK ARTCRAFTS, INCORPORATED	NEW YORK	NY	5/1/1963
NEW YORK COLISEUM, 58TH ST. AND COLUMBUS	NEW YORK	NY	7/28/1966
NEW YORK DOCK	NEW YORK	NY	5/25/1966
NEW YORK LIFE INSURANCE COMPANY	NEW YORK	NY	5/19/1959
OFFICE BUILDING, 1133 6TH AVE.	NEW YORK	NY	2/15/1968
OFFICE BUILDING, 43RD AND 6TH AVE.	NEW YORK	NY	11/1/1968
OFFICE BUILDING, 825 3RD AVE.	NEW YORK	NY	5/27/1968
P.S. 185, 111TH ST.	NEW YORK	NY	7/15/1966
PORT AUTHORITY HELIPORT, NEW YORK WORLD'S FAIR	NEW YORK	NY	5/6/1963
PRESBYTERIAN HOSPITAL, NEUROLOGICAL INSTITUTE	NEW YORK	NY	9/15/1959
RICHLEE ACOUSTICS & FIREPROOF, 207TH ST.	NEW YORK	NY	4/6/1960
ROBERT A. KEASBEY COMPANY	NEW YORK	NY	6/29/1964
ROBERT LEVINE	NEW YORK	NY	5/31/1960
ROSE ASSOCIATES, 280 PARK AVE.	NEW YORK	NY	6/21/1963
S.S. AMERICAN CHALLENGER, NORTH RIVER, PIER 54	NEW YORK		11/29/1966
S.S. AMERICAN CHALLENGER, NORTH RIVER, PIER 54	NEW YORK	NY	1/1/1966
, ,	NEW YORK	NY	3/17/1970
SIGMUND SOMMERS CONSTRUCTION, 3RD AVE.			• •
SPRAYED FIREPROOFING CONTRACTOR 240 F 24TH ST	NEW YORK	NY	5/24/1963
SPRAYED FIREPROOFING CONTRACTOR, 249 E. 34TH ST.	NEW YORK	NY	5/10/1963
SPRAYED FIREPROOFING CONTRACTORS, 39TH ST. AND BROADWAY AVE.	NEW YORK	NY	7/22/1968
SPRAYED FIREPROOFING CONTRACTORS, 93-99 LAFAYETTE ST.	NEW YORK	NY	3/7/1969
SPRAYON INSULATION & ACOUSTICS INC., DU BOISE	NEW YORK	NY	1/1/1958
SPRAYON INSULATION & ACOUSTICS, INC., DU BOISE	NEW YORK	NY	1/1/1958
ST. AUGUSTINE CHAPEL	NEW YORK	NY	3/13/1963
THREE-EIGHTY GARAGE, 19 E. 46TH ST.	NEW YORK	NY	8/9/1958
TRAVELERS INSURANCE, NEW YORK WORLD'S FAIR	NEW YORK	NY	2/20/1964
UNION CARBIDE BUILDING, 270 PARK AVE.	NEW YORK	NY	8/15/1958
URIS BROTHERS, 350 PARK AVE.	NEW YORK	NY	1/28/1960
WALL ST.	NEW YORK	NY	1/1/1969
WATER ST., JOHN AND PEARL STS.	NEW YORK	NY	4/3/1969
WATERSIDE PLANT, UNIT #13, 708 1ST AVE.	NEW YORK	NY	7/30/1958
			-

WATERSIDE STATION, 674 1ST AVE.	NEW YORK	NY	5/10/1968
WILLIAM ISHERWOOD, INCORPORATED	NEW YORK	NY	7/12/1966
WORLD TRADE CENTER	NEW YORK	NY	3/29/1969
WORLD TRADE CENTER, CUSTOM HOUSE	NEW YORK	NY	6/8/1971
WORLD TRADE CENTER, TOWER A	NEW YORK	NY	5/21/1971
WORLD TRADE CENTER, TOWER B	NEW YORK	NY	10/13/1970
NIAGRA FALLS HIGH SCHOOL	NIAGRA FALLS	NY	2/14/1962
PUBLIC SAFETY BUILDING, HYDE PARK BLVD.	NIAGRA FALLS	NY	10/31/1961
STATE UNIVERSITY CAMPUS, DINING HALL	ONEONTA	NY	8/16/1967
YMCA BUILDING	ONEONTA	NY	2/15/1965
STATE TEACHERS COLLEGE, INDUSTRIAL ARTS BUILDING	OSWEGO	NY	3/23/1964
PHOENIX JUNIOR HIGH SCHOOL	PHOENIX	NY	9/23/1964
ELECTRIC POWER STATION	PORT JEFFERSON	NY	5/13/1968
YMCA BUILDING	POUGHKEEPSIE	NY	5/15/1968
134TH ST. & UNION TPKE.	QUEENS	NY	8/29/1963
67TH AVE. AND 102ND ST.	QUEENS	NY	9/5/1962
AMERICAN AIRLINE PASSENGER TERMINAL BUILDING	QUEENS	NY	5/15/1959
MALMED REALTY CORPORATION, 48-29 31st PL.	QUEENS	NY	1/1/1963
QUEENS FAMILY COURT	QUEENS	NY	1/12/1970
ST. JOHNS UNIVERSITY	QUEENS	NY	9/22/1960
R.H. MACY, QUEENS BLVD.	REGO PARK, QUEENS	NY	1/28/1965
3-M BUILDING, 1900 MOUNT REED BLVD.	ROCHESTER	NY	11/13/1967
ELMER W. DAVIS, INCOPORATED	ROCHESTER	NY	12/31/1964
EVERGARD COATINGS, INC.	ROCHESTER	NY	1/1/1968
MCCURDYS DEPARTMENT STORE ADDITION	ROCHESTER	NY	4/20/1961
ROBERT F. HYLAND & SONS	ROCHESTER	NY	7/27/1966
ROCHESTER DAVIS-FETCH CORPORATION, 22 WRIGHT ST.	ROCHESTER	NY	6/26/1958
ROCHESTER INDUSTRIES INSULATION	ROCHESTER	NY	11/10/1964
SECURITY TRUST BUILDING, MAIN AND ELM STS.	ROCHESTER	NY	3/23/1964
ST. MARYS HOSPITAL	ROCHESTER	NY	• •
TURNER CONSTRUCTION COMPANY	ROCHESTER		9/19/1968
UNIVERSITY OF ROCHESTER, DINING HALL	ROCHESTER	NY	3/25/1963
UNIVERSITY OF ROCHESTER, RUSH RHEES LIBRARY	ROCHESTER	NY	3/9/1964
WILLIAM SUMMERHAYS SONS CORPORATION	ROCHESTER	NY	· . · .
CONTINENTAL BAKING COMPANY	RYE	NY	3/8/1968
MILTON HARBOR HOUSES	RYE	NY	4/19/1967
SCHENECTADY SAVINGS & LOAN ASSOCIATION	SCHENECTADY	NY	6/22/1959
SUFFOLK COMMUNITY COLLEGE, LIBRARY BUILDING	SELDEN	NY	2/26/1968
SPENCER VAN ELTON SCHOOL	SPENCER	NY	9/21/1970
RICHLEE ACOUSTICS & FIREPROOF, RT. 59 AND N.Y. THRUWAY	SPRING VALLEY	NY	6/27/1962
SOUTH EAST QUEENS HIGH SCHOOL	SPRINGFIELD GARDENS	NY	5/6/1965
NEW YORK STATE ANIMAL HOSPITAL	SPRINGVILLE	NY	11/12/1958
STAMFORD HOSPITAL	STAMFORD	NY	12/1/1969
RICHMOND COLLEGE	STATEN ISLAND	NY	10/11/1968
ST. GEORGE PLAZA	STATEN ISLAND	NY	5/6/1966
SUSAN WAGNER HIGH SCHOOL	STATEN ISLAND	NY	12/6/1966
HENDERSON-JOHNSON COMPANY, 918 CANAL ST.	SYRACUSE	NY	7/28/1960
	J		., 20, 1500

MADINE MIDI AND DANK	CVDACUCE	NIV	4/12/1002
MARINE MIDLAND BANK	SYRACUSE	NY	4/12/1963
MONY INSURANCE BUILDING	SYRACUSE	NY	12/10/1965
MONY PLAZA	SYRACUSE	NY	7/22/1968
ONONDAGA COMPANY SAVINGS BANK, OFFICE BUILDING	SYRACUSE	NY	8/21/1963
PUBLIC SAFETY BUILDING, TOWNSEND ST.	SYRACUSE	NY	5/10/1963
ROBERT A. KEASBEY COMPANY	SYRACUSE	NY	10/1/1963
SIBLEY PROJECT, PHASE B, SOUTH SALINA ST.	SYRACUSE	NY	1968
TOWER APARTMENTS, 770 JAMES ST.	SYRACUSE	NY	9/19/1962
VETERANS ADMINISTRATION HOSPITAL	SYRACUSE	NY	2/21/1966
VINCENT J. SMITH COMMUNITY HOSPITAL	SYRACUSE	NY	6/21/1962
WARREN STREET PARKING GARAGE	SYRACUSE	NY	5/23/1962
WATER FILTRATION PLANT, MAINTENANCE BUILDING	TROY	NY	9/24/1964
SAVINGS BANK OF UTICA	UTICA	NY	7/18/1963
STATE OFFICE BUILDING, GENESSEE ST.	UTICA	NY	3/16/1970
UTICA MUTUAL LIFE INSURANCE, RT. #5	UTICA	NY	1/1/1970
HARPER COLLEGE, PHYSICAL EDUCATION BUILDING	VESTAL	NY	9/23/1968
TEMPLE ISRAEL BUILDING	VESTAL	NY	4/19/1968
ST. FRANCIS RECTORY, WAMTAUGH AVE.	WAMTAUGH	NY	5/23/1967
YMCA - YWCA POOL BUILDING	WATERTOWN	NY	10/4/1968
KAY FRIES CHEMICAL	WEST HAVERSTRAW	NY	7/18/1962
FORTUNOFF'S DEPARTMENT STORE	WESTBURY	NY	11/25/1963
CROSS WESTCHESTER TOWERS, NORTH ELMSFORT	WESTCHESTER COUNTY	NY	11/1/1969
KORVETTES DEPARTMENT STORE	WESTCHESTER COUNTY	NY	11/12/1965
COLLETTI & LAGUIDARA, INCORPORATED, BLOOMINGDALE RD.	WHITE PLAINS	NY	3/29/1967
IBM BUILDING, FACULTY COMPLEX BUILDING	WHITE PLAINS	NY	1/1/1968
IBM HEADQUARTERS, FACULTY COMPLEX BUILDING	WHITE PLAINS	NY	5/27/1968
MACYS DEPARTMENT STORE	WHITE PLAINS	NY	1/1/1961
OFFICE BUILDING, 20 GLEN ST.	WHITE PLAINS	NY	10/15/1969
STEWART MULLER, 222 WESTCHESTER AVE.	WHITE PLAINS WHITE PLAINS	NY	3/14/1968
WHITE PLAINS PLAZA	WHITE PLAINS WHITE PLAINS	NY	
	_		10/16/1967
CUDDIHY, HUEBNER & LAMPARTER	WYANDANCH		3/18/1968
NEW YONKERS PUBLIC LIBRARY	YONKERS	NY	8/3/1961
ADAMSON UNITED COMPANY, 730 CARROLL ST.	AKRON	OH	9/11/1964
AKRON CITY HOSPITAL	AKRON	ОН	3/6/1964
AKRON SAFETY BUILDING, BOWERY ST.	AKRON	ОН	7/21/1965
AKRON UNIVERSITY, SOMMER ST.	AKRON	ОН	8/10/1966
AKRON UNIVERSITY, STUDENT CENTER	AKRON	ОН	10/3/1966
NATIONAL DRYWALL, INCORPORATED, 145 OLIVER ST.	AKRON	ОН	•
SEARS, CHAPEL HILL SHOPPING CENTER	AKRON	ОН	3/23/1966
WILLIAM LAY COMPANY, 2501 LAY DRIVE	AKRON	ОН	10/28/1960
ASHTUBULA METALS COMPANY, MIDDLE RD.	ASHTUBULA	ОН	3/28/1963
PEARL OIL COMPANY	BOWLING GREEN	ОН	3/8/1965
REPUBLIC STEEL CORPORATION	CANTON	ОН	11/6/1963
B. & O., 2ND AND SMITH STS.	CINCINNATI	ОН	4/11/1962
BROWN PLASTERING COMPANY	CINCINNATI	ОН	4/29/1959
EMERY PARKING DECK, 16TH AND RACE ST.	CINCINNATI	ОН	6/29/1959
ESSEX HOUSE, 7610 READING RD.	CINCINNATI	ОН	9/27/1967

H. & S. POGUE DEPTARTMENT STORE, KENWOOD PLAZA	CINCINNATI	ОН	5/12/1966
MARJORIE P. LEE JOB, 3550 SHAW AVE.	CINCINNATI	ОН	7/1/1963
NATIONAL LIFE INSURANCE COMPANY, 180 E. MCMILLAN ST.	CINCINNATI	ОН	4/1/1960
OWENS-CORNING FIBERGLAS	CINCINNATI	ОН	2/16/1965
R.B. BRUNEMANN & SONS, INCORPORATED	CINCINNATI	ОН	2/17/1966
UNIVERSITY OF CINCINNATI, SCIENCE COMPLEX	CINCINNATI	ОН	10/2/1967
UNIVERSITY OF CINCINNATI, STUDENT UNION BUILDING	CINCINNATI	ОН	1/25/1965
ACOUSTICAL DEPARTMENT, 10223 MILES AVE.	CLEVELAND	ОН	12/21/1964
ALTON COMPANY	CLEVELAND	ОН	2/18/1960
AUTOMOBILE CLUB, 6000 S. MARGINAL DR.	CLEVELAND	ОН	4/29/1968
CLEVELAND TRUST COMPANY, COMPUTER CENTER	CLEVELAND	ОН	6/25/1963
E.F. DONLEYS SONS, INCORPORATED, 10203 MILES AVE.	CLEVELAND	ОН	4/3/1962
FEDERAL OFFICE BUILDING, E. T. SAMPSON, LAKESIDE AND E. 9TH ST.	CLEVELAND	ОН	
FEDERAL OFFICE BUILDING, LAKESIDE AVE. & E. 9th ST., FRANK BRISCOE COMPAN		ОН	
FEDERAL OFFICE BUILDING, N.E. CORRIDOR	CLEVELAND	ОН	6/17/1965
	CLEVELAND	OH	1/30/1968
LIMBACH COMPANY, 1271 E. 6TH ST.			
REPUBLIC STEEL, BABCOCK & WILCOX, STANDARD	CLEVELAND	OH	1/1/1965
TERMINAL TOWER, PROSPECT AVE.	CLEVELAND	OH	5/13/1965
UNIVERSITY HOSPITAL, GENERAL SERVICE BUILDING	CLEVELAND	ОН	6/23/1965
WESTERN RESERVE UNIVERSITY RESEARCH BUILDING	CLEVELAND	ОН	10/9/1961
WESTGATE SHOPPING CENTER	CLEVELAND	ОН	4/25/1961
STEEL BUILDINGS COMPANY	CLYDE	ОН	6/21/1960
BUEL GATTERDAM COMPANY, INCORPORATED	COLUMBUS	ОН	6/24/1959
CHEMICAL ABSTRACT BUILDING	COLUMBUS	ОН	10/31/1963
COLUMBUS & SOUTHERN OHIO ELECTRIC COMPANY	COLUMBUS	ОН	11/6/1957
COLUMBUS DISPATCH BUILDING, 34 S. 3RD ST.	COLUMBUS	ОН	9/16/1963
OHIO EDUCATION ASSOCIATION	COLUMBUS	ОН	2/26/1965
OHIO STATE UNIVERSITY, MEDICAL BASIC SCIENCE BUILDING	COLUMBUS	ОН	11/26/1968
OHIO STATE UNIVERSITY, RESEARCH SPACE	COLUMBUS	ОН	1963
OHIO STATE UNIVERSITY, SCHOOL OF NURSING	COLUMBUS	ОН	7/17/1967
LAWSON DAIRY COMPANY, MAIN ST.	CUYAHOGA FALLS	ОН	8/25/1965
7581 BRANDT PK.	DAYTON	ОН	7/19/1967
DAYTON POWER & LIGHT	DAYTON	ОН	9/4/1964
JAMES C. OREN	DAYTON	ОН	10/13/1965
MEADOWDALE SCHOOL	DAYTON	ОН	12/2/1966
UNIVERSITY OF DAYTON, SCHOOL OF BUSINESS	DAYTON	ОН	6/11/1965
WRIGHT STATE CAMPUS, SCIENCE BUILDING	DAYTON	ОН	5/2/1967
WRIGHT-PATTERSON AIRFORCE BASE	DAYTON	ОН	
T.R.W., BUILDING 1, 23555 EUCLID AVE.	EUCLID	ОН	11/22/1965
T.R.W., TAPCO DIVISION BUILDING #19 & #24	EUCLID	ОН	
COMPOSITE MEDI-FACILITY, AIRFORCE BASE	FAIRBORN	ОН	7/19/1968
WRIGHT-PATTERSON AIRFORCE BASE, AREA B	FAIRBORN	ОН	5/11/1960
·			
FAIRVIEW PARK HOSPITAL	FAIRVIEW PARK	OH	1/4/1967
FREMONT PROGRESS AIRPORT	FREMONT	OH	10/24/1963
DENNISON UNIVERSITY	GRANVILLE	OH	• •
PLYMER PLANT, GLIDDEN COMPANY	HURON	OH	1/11/1966
KENT UNIVERSITY, APPLIED SCIENCE & FINE ARTS	KENT	OH	6/27/1966

KENT UNIVERSITY, STUDENT BUILDING	KENT	ОН	3/15/1971
ST. JOHN'S HOSPITAL	LAKEWOOD	ОН	9/25/1964
J.C. PENNEY BUILDING	LIMA	ОН	8/10/1965
LIMA MEMORIAL HOSPITAL	LIMA	ОН	3/23/1968
MAY COMPANY JOB, LAKE ERIE HALL	MENTOR	ОН	12/2/1963
T.R.W., METALS DIVISION	MINERVA	ОН	9/16/1966
MUSKIGEEN COLLEGE, NATIONAL SCIENCE BUILDING	NEW CONCORDE	ОН	8/19/1970
OHIO EDISON, WESTINGHOUSE TURBINE	STRATTON	ОН	9/29/1967
ST. PAUL'S METHODIST CHURCH	TIFFIN	ОН	• •
LIBBEY-OWENS FORD BUILDING	TOLEDO	ОН	
TOLEDO STEEL TUBE, 8201 W. CENTRAL AVE.	TOLEDO	ОН	10/5/1964
JAMES CAMPBELL SMITH, INCORPORATED	WILLOUGHBY	ОН	7/7/1965
HOME INSULATION COMPANY, 999 TRUMBULL AVE.	YOUNGSTOWN		1957
HOME INSULATION, 2136 W. FEDERAL ST.	YOUNGSTOWN		8/12/1959
SHEET & TUBE, OFFICE BUILDING	YOUNGSTOWN	ОН	11/8/1957
TODD INSULATION & REMODELING	YOUNGSTOWN	ОН	1/1/1967
MUSKUGEE HIGH SCHOOL	MUSKUGEE	ОК	3/7/1966
AIRPORT / LILLARD PARK	OKLAHOMA CITY	OK	
·			12/26/1957
CEILING SYSTEMS, INCORPORATION	OKLAHOMA CITY	OK	, ,
OWENS-CORNING SUPPLY CORPORATION	OKLAHOMA CITY	OK	2/25/1965
SANTA FE DELIVERY, 28TH ST. TEAM TRACK	OKLAHOMA CITY	OK	5/12/1958
SCOVIL & SUBLETT, 100 N.E. 26TH ST.	OKLAHOMA CITY	OK	12/8/1961
SCOVIL & SUBLETT, 20 N.E. 27TH ST.	OKLAHOMA CITY	OK	11/18/1957
SUBLETT & ASSOCIATES, 100 N.E. 26TH ST.	OKLAHOMA CITY	OK	5/14/1964
CINERAMA THEATRE, 31ST AND MEMORIAL	TULSA	OK	1/12/1966
JOHNS-MANVILLE CORPORATION, 1701 E. 7TH ST.	TULSA	ОК	6/11/1964
E.J. BARTELLS COMPANY	PORTLAND	OR	8/28/1964
MASONS SUPPLY COMPANY	PORTLAND	OR	11/8/1961
ALIQUIPPA HIGH SCHOOL	ALIQUIPPA	PA	1/8/1960
ALIQUIPPA JUNIOR HIGH SCHOOL	ALIQUIPPA	PA	11/12/1959
A.B.E. AIRPORT	ALLENTOWN	PA	12/1/1969
DUGGAN & MARCON, INCORPORATED	ALLENTOWN	PA	4/17/1964
EASTERN INDUSTRIAL INSULATION	ALLENTOWN	PA	6/20/1963
THERMAL ENGINEERING, INCORPORATED	ALLENTOWN	PA	10/24/1966
AMERICAN INDUSTRIAL CONTRACTOR	ALLISON PARK	PA	8/4/1966
ALTOONA VOCATIONAL TECHNOLOGY HIGH SCHOOL	ALTOONA	PA	10/31/1969
WRIGHT ELEMENTARY SCHOOL	ALTOONA	PA	9/4/1970
FEDERAL SAVINGS BANK	ARDMORE	PA	8/19/1959
WESTERN ELECTRIC CORPORATION	BEAVER	PA	9/13/1963
WESTINGHOUSE ELECTRIC CORPORATION	BEAVER	PA	5/18/1966
OWENS-CORNING FIBERGLAS	BERWYN	PA	3/13/1967
BETHLEHEM STEEL COMPANY	BETHLEHEM	PA	11/10/1964
LEHIGH PIPE SHOP	BETHLEHEM	PA	9/14/1964
MORAVIAN COLLEGE, SCIENCE BUILDING	BETHLEHEM	PA	3/19/1970
MIDDLETOWN SCHOOL	BORNSTEIN	PA	2/17/1959
EDGAR THOMPSON WORKS	BRADDOCK	PA	10/3/1967
BRYN MAWR HOSPITAL	BRYN MAWR	PA	10/3/1967
DIVIN INICAMIVITOSETTAL	UNAWINI NI VIO	гA	11/3/1303

JOHN F. CASEY, INCORPORATED, PULLMAN STANDARD	BUTLER	PA	
NEW STEEL BUILDING, SPANG STEEL COMPANY	BUTLER	PA	
DICKINSON SCHOOL OF LAW	CARLYLE	PA	
CEDERBROOK HILLS APARTMENTS	CEDERBROOK HILLS	PA	
CHESTNUT HILL ACADEMY	CHESTNUT HILL	PA	
KEYSTONE PROJECT, CLAIRTON WORKS	CLAIRTON	PA	
ACOUSTICRAFT, INCORPORATED, ELM AND POPLAR STS.	CONSHOHOCKEN	PA	
WILLIAM ARMSTRONG & SONS	CONSHOHOCKEN	PA	
AMP, INCORPORATED, MAIN OFFICE BUILDING, RT. 283	DAUPHIN COUNTY	PA	
CHEW-BITTEL COMPANY	DOYLESTOWN	PA	
EASTON HIGH SCHOOL	EASTON	PA	
KEYSTONE PROJECT, WESTINGHOUSE ELECTRIC	ELDERTON	PA	
WILLIAM ARMSTRONG & SONS, PHIL ELLENA AND CHEW STS.	GERMANTOWN		4/4/1962
STRAWBRIDGE & CLOTHER DEPARTMENT STORE	GERMANTOWN PIKE	PA	, ,
AMERICAN INDUSTRIAL, EAST BEAVER ST.	GLENFIELD	PA	
6 NORTH PROGRESS AVE.	HARRISBURG	PA	
600 NORTH PROGRESS ST.	HARRISBURG	PA	
AMP, INCORPORATED, INTERSTATE 83	HARRISBURG	PA	
NOVINGERS, INCORPORATED	HARRISBURG	PA	-, -,
PENN STATE EDUCATIONAL ASSOCIATION BUILDING	HARRISBURG	PA	
STATE OFFICE BUILDING, COMMONWEALTH AVE. AND FOSTER ST.	HARRISBURG	PA	
FOUNDERS HALL, RT. 322	HERSHEY	PA	
WAYNE COUNTY MEMORIAL HOSPITAL	HONESDALE	PA	, ,
ELLIOT STEEL COMPANY	JEANETTE	PA	
BEAVER HILL APARTMENTS CAMPBIA LEASING CORPORATION, 73 LUCKORY ST	JENKINTOWN		8/7/1967
CAMBRIA LEASING CORPORATION, 73 HICKORY ST.	JOHNSTOWN	PA	
MEADOWVALE SCHOOL, MESSENGER ST. KING OF PRUSSIA SHOPPING CENTER	JOHNSTOWN KING OF PRUSSIA	PA	
		PA	
ARMSTRONG CONTRACTORS & SUPPLY	LANCASTER LANCASTER	PA PA	, ,
HESS DEPARTMENT STORE, HARRISBURG PENN DAIRIES, GREENFIELD RD.	LANCASTER		• •
WALL & SHAND, DUKE AND VINE STS.	LANCASTER	PA PA	6/24/1964 5/29/1963
LATROBE STEEL COMPANY	LATROBE	PA	· · · · · · · · · · · · · · · · · · ·
WILLIAM A. SHOEMAKER SCHOOL	MACUNGIE	PA	
KARL R.H. KNAPP, INCORPORATED, THOMAS ST. AND MCKEES AVE.	MCKEES ROCK	PA	
THOMAS ST. AND MCKEES AVE.	MCKEES ROCK	PA	
ALCOA RESEARCH LABS	NEW KENSINGTON	PA	· · ·
A.C. & S., INCORPORATED	PHILADELPHIA	PA	
BASIC CONSTRUCTION COMPANY, 17TH AND MARKET STS.	PHILADELPHIA	PA	
BOTFIELD REFRACTORIES COMPANY, 776 S. SWANSON ST.	PHILADELPHIA	PA	
FELS NAPTHA SOAP COMPANY, 73RD AND WOODLAND	PHILADELPHIA	PA	
FRANKLIN MOTEL, 21ST AND SPRING GARDEN STS.	PHILADELPHIA	PA	
GERARD TRUST BUILDING, 15TH AND MARKET STS.	PHILADELPHIA	PA	
IBM BUILDING, 17TH AND MARKET STS.	PHILADELPHIA	PA	
IMMACULATE CONCEPTION CATHEDRAL	PHILADELPHIA	PA	
OWENS-CORNING FIBERGLAS	PHILADELPHIA	PA	
PHIL ELLENA AND CHEW STS.	PHILADELPHIA	PA	
THE ELLIVE AIND CHEVY 313.	THEADELINA	ГA	7/ 11/ 1303

PHILADELPHIA ELECTRIC COMPANY	PHILADELPHIA	PA	5/19/1969
PHILADELPHIA GREYHOUND TERMINAL	PHILADELPHIA	PA	3/23/1964
PHILLIP CAREY MANUFACTURING COMPANY	PHILADELPHIA	PA	1/1/1967
TEMPLE UNIVERSITY	PHILADELPHIA	PA	5/8/1968
TRADE & CONVENTION CENTER, 34TH AND CURIE	PHILADELPHIA	PA	7/14/1966
WEATHERSPOONG BUILDING, JUNIPER AND WALNUT STS.	PHILADELPHIA	PA	6/7/1963
WILLIAM ARMSTRONG & SONS	PHILADELPHIA	PA	4/2/1964
ALCOA RESEARCH LABS	PITTSBURGH	PA	1/1/1968
ALLEGHANY AIRPORT, PITTSBURGH AIRPORT	PITTSBURGH	PA	3/11/1958
ANSELM SCHOOL, 7446 MCCLURE AVE.	PITTSBURGH	PA	11/23/1960
BETHEL SENIOR HIGH SCHOOL	PITTSBURGH	PA	10/27/1959
DIETZ TRUCKING COMPANY, 166 39TH ST.	PITTSBURGH	PA	8/10/1959
GATEWAY CENTER BUILDING #4	PITTSBURGH	PA	5/12/1959
H.K. PORTER, GRANT AND 6TH STS.	PITTSBURGH	PA	1/16/1958
IBM BUILDING, BLVD. OF THE ALLIES	PITTSBURGH	PA	1/1/1962
IBM BUILDING, STANWIX ST.	PITTSBURGH	PA	9/12/1962
KARL R.H. KNAPP, INCORPORATED, 1015 PERALTA ST.	PITTSBURGH	PA	
KAUFMANS DEPARTMENT STORE	PITTSBURGH	PA	3/23/1965
NORTHVIEW HEIGHTS ELEMENTARY SCHOOL	PITTSBURGH	PA	5/10/1962
	PITTSBURGH		
NORTHVIEW HOUSING, HAZLET AND SPRING HILL		PA	3/13/1962
PITTSBURGH AIRPORT	PITTSBURGH	PA	1/1/1958
ST. JAMES ACTIVITY BUILDING	PITTSBURGH	PA	10/27/1959
STANDARD FLOOR COMPANY	PITTSBURGH	PA	11/12/1958
WESTINGHOUSE ELECTRIC, RESEARCH LAB	PITTSBURGH	PA	7/7/1961
POTTSVILLE AREA MIDDLE SCHOOL	POTTSVILLE	PA	11/14/1968
POTTSVILLE HOSPITAL	POTTSVILLE	PA	1/4/1965
ALTMAN STORE, LANCASTER PK. AND IVEN RD.	RADNOR	PA	4/19/1965
EASTERN INDUSTRIAL INSULATION	READING	PA	4/28/1964
SCHMIDT LUMBER MILLWORK	SCRANTON	PA	9/27/1963
FIRST FEDERAL SAVINGS & LOAN	SHARON		4/27/1967
P.H. GLATFELDER COMPANY	SPRING GROVE	PA	3/30/1967
WESSEX HOUSE APARTMENTS	ST. DAVIDS	PA	4/27/1966
TOBYHANNA SIGNAL DEPOT, HANGAR BUILDING	TOBYHANNA	PA	7/12/1960
PRECISION PRODUCTS COMPANY, BERRY RD.	WASHINGTON	PA	7/28/1967
PENN-DIXIE CEMENT COMPANY	WEST WINFIELD	PA	4/27/1967
CHEW-BITTEL COMPANY	WILLOW GROVE	PA	11/5/1957
VALLEY FORGE	WILLOW GROVE	PA	1/28/1958
VALLEY FORGE HOSPITAL	WILLOW GROVE	PA	3/3/1958
1709 GRAHAM AVE.	WINDBAR	PA	3/24/1970
CUSTOM INSULATION COMPANY	YORK	PA	4/22/1960
WESTSIDE OSTEOPATHIC HOSPITAL	YORK	PA	10/20/1961
ALLOY PROPELLER COMPANY, RT. 19	ZELIENPOLE	PA	9/30/1964
MANGANARO BROTHERS, INC.	NEWPORT	RI	1/1/1970
OWENS-CORNING SUPPLY CORPORATION	PAWTUCKET	RI	1/10/1964
HOLIDAY INN	PROVIDENCE	RI	8/7/1968
INTELEX SYSTEMS, GILBANE GENERAL CONTRACTOR	PROVIDENCE	RI	7/11/1960
MANGANARO BROTHERS, INC.	PROVIDENCE	RI	1/1/1968
- /········, ··· - ·			_, _, _000

MUNICIPAL WHARF	PROVIDENCE	RI	12/2/1960
ST. JOSEPH'S HOSPITAL	PROVIDENCE	RI	2/10/1964
U.S. NAVAL AIR STATION, TEST FACILITY	QUONSET POINT	RI	8/20/1959
SEARS ROEBUCK, DIMEO CONSTRUCTION	WARWICK	RI	5/25/1966
ST. JOHN THE BAPTIST SCHOOL	WEST WARWICK	RI	7/9/1958
FOSTER WHEELER / CELANESE FIBERS COMPANY	CELRIVER	SC	2/25/1966
JOHNS-MANVILLE PRODUCTS CORPORATION	CELRIVER	SC	1/1/1966
REID HAYDEN, INCORPORATED	CELRIVER	SC	8/1/1966
NORTH BROTHERS, INCORPORATED, 79 REID ST.	CHARLESTON	SC	1/7/1970
NORTH BROTHERS, INCORPORATED, 1627 COLUMBIA COLLEGE DR.	COLUMBIA	SC	6/12/1970
CLINIC FOR DOCTORS PRICE & HART	FLORENCE	SC	9/2/1958
CONGREGATION BETH ISRAEL, COMMUNITY CENTER	GREENVILLE	SC	5/28/1958
HARPER BUILDING	GREENVILLE	SC	12/23/1957
LINEBERGER & PAUL GENERAL CONTRACTORS	GREENVILLE	SC	7/14/1959
UNITED OFFICE INCORPORATED, LINEBERGER AND PAUL	GREENVILLE	SC	7/29/1959
WEST VIRGINIA PULP & PAPER COMPANY	NORTH CHARLESTON	SC	3/7/1963
ALDERSGATE METHODIST CHURCH	SUMTER	SC	2/25/1958
S.C. ELECTRIC & GAS, WATEREE STEAM PLANT	WATEREE	SC	2/5/1970
VETERANS ADMINISTRATION HOSPITAL	FORT MEADE	SD	1/12/1965
V.C. PENGRA CONSTRUCTION COMPANY	RAPID CITY	SD	1/13/1960
PROVIDENCE LIFE & ACCIDENT COMPANY BUILDING	CHATTANOOGA	TN	1/25/1960
APPLIED FIBERS COMPANY	CLARKSVILLE	TN	5/13/1960
COLUMBIA CENTRAL HIGH SCHOOL	COLUMBIA	TN	5/10/1961
NORTH BROTHERS, INCORPORATED, 560 INDIAN ST.	KNOXVILLE	TN	4/2/1969
NORTH BROTHERS, INCORPORATED, 5800 MIDDLEBROOK PK.	KNOXVILLE	TN	5/25/1970
AMERICAN ENKA, FILAMENT CHEMICAL PAPER ROOM	LOWLAND	TN	10/11/1963
1212 SOUTH BELLEVUE	MEMPHIS	TN	5/25/1970
AIR ROUTE AIRPORT	MEMPHIS	TN	6/13/1961
COURT HOUSE & FEDERAL OFFICE BUILDING	MEMPHIS	TN	9/24/1962
MUNICIPAL AIRPORT, AIR TRAFFIC CONTROL	MEMPHIS	TN	8/7/1961
NORTH BROTHERS, INCORPORATED, 995 N. HOLLYWOOD ST.	MEMPHIS	TN	
RALEIGH SPRING MALL, J.C. PENNEY	MEMPHIS	TN	4/15/1971
WOOLSULATE CORPORATION	MEMPHIS	TN	1/13/1970
YOUNG INSULATION COMPANY	MEMPHIS	TN	1/1/1970
TRI-CITY INSULATION COMPANY	NASHVILLE	TN	5/22/1959
4714 BUFFALO TRAIL	AMARILLO	TX	6/18/1963
HALLMARK BUILDERS, INCORPORATED	AMARILLO	TX	6/3/1968
HALLMARK SERVICE & SUPPLY, CANYON DR.	AMARILLO	TX	1/1/1971
ROBINSON INSULATION COMPANY	AMARILLO	TX	11/28/1962
THRIF-T-BUILT, INCORPORATED	AMARILLO	TX	9/24/1965
WONDER BUILDING COMPANY, 16TH AND GRAND	AMARILLO	TX	2/18/1959
TEXAS HIGHWAY DEPARTMENT	CORPUS CHRISTI	TX	12/28/1961
AMERICAN AIRLINES TERMINAL, LOVEFIELD	DALLAS	TX	8/11/1967
APPAREL MART BUILDING	DALLAS	TX	7/13/1964
CARPENTER PLASTERING COMPANY	DALLAS	TX	3/28/1968
DALLAS POWER & LIGHT COMPANY	DALLAS	TX	5/21/1964
FAIR DEPARTMENT STORE, LOCHWOOD VILLAGE	DALLAS	TX	3/31/1960

HAWES TEAM TRACK, MKT RR	DALLAS	TX	2/20/1968
LONE STAR GAS BUILDING	DALLAS	TX	4/2/1966
NO MALL OFFICE BUILDING, EXCHANGE PARK	DALLAS	TX	5/13/1965
PITTMAN TEAM TRACK	DALLAS	TX	2/23/1967
RAY TAYLOR, INCORPORATED	DALLAS	TX	7/22/1966
STORBECK GREGORY & DILLARD	DALLAS	TX	10/6/1964
UNITED TILE COMPANY, DALLAS	DALLAS	TX	4/1/1959
CARTER LATHING & PLASTERING	EL PASO	TX	4/14/1959
EL PASO NATIONAL BANK BUILDING	EL PASO	TX	3/28/1962
UNIVERSITY TOWERS	EL PASO	TX	6/24/1959
MEEKS SERVICE STATION	FARWELL	TX	5/11/1959
ARTC BUILDING, AMON CARTER FIELD	FORT WORTH	TX	10/20/1960
BUILDERS SERVICE COMPANY	FORT WORTH		
		TX	8/9/1961
CAMP BOWIE TEAM TRACK	FORT WORTH	TX	10/14/1966
NORTHEAST CAMPUS JUNIOR COLLEGE	FORT WORTH	TX	4/15/1969
UNIVERSITY CHRISTIAN CHURCH	FORT WORTH	TX	12/2/1966
T. & P. TEAM TRACK	GRAND PRAIRIE	TX	7/25/1967
COUNTY BULL BARN	HEREFORD	TX	12/10/1957
ACOUSTICAL PRODUCTS COMPANY, 6417 EPPS ST.	HOUSTON	TX	4/7/1959
DUNNE-WHITE & ASSOCIATES	HOUSTON	TX	8/22/1969
FIRST CITY NATIONAL BANK BUILDING	HOUSTON	TX	11/22/1960
JOHNS-MANVILLE SALES CORPORATION	HOUSTON	TX	7/7/1967
MUNDET CORK COMPANY, 6601 SUPPLY ROW	HOUSTON	TX	12/17/1962
OWENS-CORNING SUPPLY CORPORATION	HOUSTON	TX	2/23/1965
PERFORMING ARTS BUILDING, BRAZOS AND TEXAS AVE.	HOUSTON	TX	7/7/1965
TOBIN & ROONEY, INCORPORATED	HOUSTON	TX	9/2/1965
UNIVERSITY OF HOUSTON	HOUSTON	TX	7/30/1969
LEO S. JOERIS	SAN ANTONIO	TX	4/15/1958
TEXARKANA HIGH SCHOOL	TEXARKANA	TX	1/1/1966
KATY TEAM TRACK	WICHITA FALLS	TX	4/11/1967
SEARS ROEBUCK SHOPPING CENTER	WICHITA FALLS	TX	2/23/1967
ARTC BUILDING	SALT LAKE CITY	UT	1/1/1961
BULLOUGH ASBESTOS SUPPLY COMPANY, 50 SOUTH 7TH WEST ST.	SALT LAKE CITY	UT	10/22/1963
HILL AIR FORCE BASE, HANGAR MAINTENANCE	SALT LAKE CITY	UT	
,			
KRAFT-MURPHY COMPANY	ALEXANDRIA	VA	
S. & W. CAFETERIA	ALEXANDRIA		8/11/1965
3846 SOUTH FOUR MILE RD.	ARLINGTON	VA	, ,
ARLINGTON TOWERS	ARLINGTON	VA	6/16/1960
BILTON INSULATION & SUPPLY, S. NELSON ST.	ARLINGTON	VA	
BASSETT INDUSTRIES JOB	BASSETT	VA	10/3/1960
LIBERTY HIGH SCHOOL, RT. 122	BEDFORD COUNTY	VA	
STAUNTON RIVER HIGH SCHOOL/BEDFORD COURT HOUSE	BEDFORD COUNTY	VA	11/20/1963
V.P.I. DINING HALL	BLACKSBURG	VA	4/17/1962
V.P.I. HOME ECONOMICS BUILDING	BLACKSBURG	VA	7/14/1967
NORTH SITE JUNIOR HIGH SCHOOL	CHARLOTTESVILLE	VA	3/31/1966
O'FERRAL, INCORPORATED, STREBLING AVE.	CHARLOTTESVILLE	VA	2/17/1964
UNIVERSITY OF VIRGINIA	CHARLOTTESVILLE	VA	6/21/1965

LININ/EDGITY OF MECINIA FIELD HOUSE	CHADLOTTESVILLE	٧/٨	F /C /10CF
UNIVERSITY OF VIRGINIA, MEDICAL EDUCATION BUILDING	CHARLOTTESVILLE CHARLOTTESVILLE	VA VA	5/6/1965
UNIVERSITY OF VIRGINIA, MEDICAL EDUCATION BUILDING VIRGINIA ELECTRIC, PORTSMOUTH POWER STATION	CHESAPEAKE	VA	3/17/1970 3/7/1966
MONTGOMERY COUNTY HOSPITAL	CHRISTIANBURY	VA	5/11/1970
LYNCHBURG TRAINING SCHOOL	COLONY	VA	6/28/1963
FERRUM JUNIOR COLLEGE, BASSETT HALL WOODROW WILSON REHABILITATION CENTER, DINING HALL	FERRUM	VA	1/22/1968
UNION BAG CAMP COMPANY	FISHERVILLE FRANKLIN	VA VA	1/1/1966 4/11/1966
TAYLOR PACKING COMPANY	HALLWOOD	VA	• •
	HAMTON		6/14/1962
NASA, LANGLIE RESEARCH CENTER		VA	6/23/1967 4/22/1971
MADISON COLLEGE, PHYSICAL EDUCATION BUILDING ASSOCIATED TERMINAL	HARRISONBURG LYNCHBURG	VA VA	• •
GRAVES MILL RD.	LYNCHBURG	VA	11/14/1969
MONROE CONSTRUCTION	LYNCHBURG	VA	7/19/1971
PAUL E. STYLES PLASTERING COMPANY, GRAVES MILL RD.	LYNCHBURG	VA	1/1/1969
PAUL E. STYLES PLASTERING COMPANY, RT. 1	LYNCHBURG	VA	6/28/1963
MARTINSVILLE GENERAL HOSPITAL	MARTINSVILLE	VA	6/2/1969
	NARROWS	VA	8/18/1966
FOSTER WHEELER COMPANY, CELANESE CORPORATION ALGONQUIN APARTMENT	NORFOLK	VA	9/14/1964
ATLANTIC FLEET INTELLIGENCE CENTER, NAVY	NORFOLK	VA	9/14/1964
CULTURE & CONVENTION CENTER	NORFOLK	VA	3/19/1971
DEPAUL HOSPITAL, GRANBY ST.	NORFOLK	VA	1/31/1971
MEMORIAL LIBRARY, CITY HALL AVE.	NORFOLK	VA	4/7/1961
MILITARY CIRCLE SHOPPING CENTER, J.C. PENNEY	NORFOLK	VA	4/7/1901 4/23/1969
MILITARY CIRCLE SHOPPING CENTER, MALL STORES	NORFOLK	VA	6/19/1969
MILITARY CIRCLE SHOPPING CENTER, MALE STOKES MILITARY CIRCLE, SMITH & WELTON STORES	NORFOLK	VA	6/23/1969
MILITARY CIRCLE, VIRGINIA NATIONAL BANK	NORFOLK	VA	10/17/1969
NAVAL STAFF COLLEGE, ACADEMIC BUILDING	NORFOLK	VA	5/2/1961
NORFOLK CIVIC CENTER	NORFOLK	VA	2/5/1964
SCOPE, MONTICELLO AND BRAMBLETON AVE.	NORFOLK	VA	6/2/1970
SCOPE, THEATRE BUILDING	NORFOLK	VA	7/15/1970
VIRGINIA STATE COLLEGE, R.O.T.C. BUILDING	NORFOLK	VA	11/10/1966
MERCHANTS & FARMERS BANK, CRAWFORD ST.	PORTSMOUTH	VA	12/15/1967
ADMINISTRATION BUILDING, COURT HOUSE	PRINCESS ANN	VA	
ADMINISTRATION BUILDING, PRINCESS ANN STATION	PRINCESS ANN	VA	6/12/1968
201 PLAZA OFFICE BUILDING	RICHMOND	VA	6/30/1965
301 PLAZA OFFICE BUILDING	RICHMOND	VA	8/10/1965
ATLANTIC LIFE INSURANCE, 6TH AND GRACE STS.	RICHMOND	VA	
C.E. THURSTON & SONS, INCORPORATED, 700 DINNEIDDIE AVE.	RICHMOND	VA	6/14/1971
O'FERRAL, INCORPORATED	RICHMOND	VA	7/30/1962
RICHMOND GREYHOUND TERMINAL	RICHMOND	VA	10/29/1963
SAFETY HEALTH & WELFARE BUILDING, CIVIC CENTER	RICHMOND	VA	10/24/1962
STATE OFFICE BUILDING, BANK AND GOVERNOR STS.	RICHMOND	VA	7/28/1966
STATE PLANTERS BANK, 9TH AND MAIN ST.	RICHMOND	VA	8/15/1967
COCA-COLA PLANT	ROANOKE	VA	3/21/1967
JOHNSON-CARPER FURNITURE COMPANY, HOLLINS RD.	ROANOKE	VA	9/13/1966
ROANOKE CIVIC CENTER	ROANOKE	VA	5/11/1970
		٧, ١	3, 11, 13, 0

FRANKLIN COUNTY HANGO HIGH SCHOOL	DOCUMAN AGUNTAIN		44 /7 /4060
FRANKLIN COUNTY JUNIOR HIGH SCHOOL	ROCKY MOUNTAIN	VA	11/7/1963
FRANKLIN COUNTY OFFICE BUILDING	ROCKY MOUNTAIN	VA	10/21/1968
WESTINGHOUSE ELECTRIC, SURRY POWER STATION	SURRY	VA	5/21/1971
ADMINISTRATION BUILDING, PRINCESS ANN STATION	VIRGINIA BEACH	VA	1/1/1968
FAUQUIER COUNTY HIGH SCHOOL	WARRENTON	VA	4/19/1963
MONTICELLO SHOPPING CENTER	WILLIAMSBURG	VA	
WILLIAM & MARY COLLEGE, LIFE SCIENCE BUILDING	WILLIAMSBURG	VA	•
WILLIAM & MARY COLLEGE, WOMENS GYM	WILLIAMSBURG	VA	3/14/1963
WILLIAMSBURG CONFERENCE CENTER	WILLIAMSBURG	VA	2/22/1963
WILLIAMSBURG LODGE	WILLIAMSBURG	VA	2/5/1963
WOODLAWN HIGH SCHOOL	WOODLAWN	VA	4/17/1962
VIRGINIA ELECTRIC, YORKTOWN POWER PLANT	YORKTOWN	VA	
STUDENT RESIDENT HALL COMPLEX	BURLINGTON BURLINGTON	VT	1/17/1971
UNIVERSITY OF VERMONT, COLLEGE OF MEDICINE VERMONT SAVINGS BANK	BURLINGTON	VT VT	12/15/1966
	CAVENDISH	VT	3/16/1970
MACK MOULDING COMPANY, RT. 131	LYNDONVILLE		10/1/1963
THE COLLEGE COPLEY HOSPITAL		VT	1/20/1969
GREEN MOUNTAIN INSULATION, SAN REMO DR.	MORRISVILLE SOUTH BURLINGTON	VT VT	2/19/1968 7/9/1964
SIMMONS PRECISION PRODUCTS	VERGENNES	VT	10/13/1964
BOEING COMPANY	AUBURN		5/19/1966
BANGOR POLARIS PROJECT	BANGOR		3/19/1966
J.C. PENNEY BUILDING, C/O HOWARD LEASE COMPANY	BREMERTON		12/14/1968
EASTERN WASHINGTON STATE COLLEGE	CHENEY		4/17/1970
WESTERN PNEUMATIC TUBE COMPANY	KIRKLAND		6/20/1963
LONGVIEW FIBER COMPANY	LONGVIEW		8/28/1964
E.J. BARTELLS COMPANY	RENTON		12/31/1969
E.T. SAMPSON & COMPANY / E.J. BARTELLS COMPANY	SEATTLE		6/20/1963
FIRST NATIONAL BANK BUILDING	SEATTLE		1/1/1967
NOISE CONTROL OF SEATTLE	SEATTLE		6/21/1962
AIRPORT, 1203 W. COLLEGE	SPOKANE		4/25/1964
AID ASSOCIATION FOR LUTHERANS	APPLETON	WI	9/21/1964
BELOIT IRON WORKS	BELOIT	WI	
ST. JOANS MILITARY ACADEMY	DELAFIELD	WI	
J.C. PENNEY	EASTOWN MADISON	WI	1/6/1970
FAIRGROUNDS, NEW BENLEN BUILDING	FOND DU LAC	WI	3/4/1966
ESSENTIAL CHEMICALS COMPANY	FREDONIA	WI	7/9/1962
GLENDALE MIDDLE SCHOOL	GLENDALE	WI	12/18/1969
BAY INSULATION, 1207 N. BUCHANAN	GREEN BAY	WI	10/17/1966
H.C. PRANGE COMPANY	GREEN BAY	WI	6/14/1965
PUBLIC SERVICE BUILDING	GREEN BAY	WI	1/1/1969
PUBLIC SERVICE LIBRARY	GREEN BAY	WI	10/10/1969
10712 W. GRANGE	HALES CORNERS	WI	8/15/1966
2290 W. EDGERTON AVE.	HALES CORNERS	WI	
9920 W. EDGERTON AVE.	HALES CORNERS	WI	12/19/1966
A&H, INCORPORATED TRUCKERS	HALES CORNERS	WI	5/17/1968
SHORELINE SUPPLY COMPANY	HALES CORNERS	WI	9/22/1962
			• •

WISCONSIN ACOUSTIC & FIREPROOF, 10712 W. GRANGE	HALES CORNERS	WI	4/21/1965
WISCONSIN ACOUSTICS, INCORPORATED	HALES CORNERS	WI	8/30/1966
POELLINGER PLASTERING SERVICES	LA CROSSE	WI	7/27/1966
600 PARK ST. BUILDING	MADISON	WI	7/27/1300
TRUAX AIR FORCE BASE, BUILDING #102	MADISON	WI	1/13/1964
WEST TOWN SHOPPING CENTER, J.C. PENNEY	MADISON	WI	7/16/1970
WISCONSIN ACOUSTIC & FIREPROOF, 4821 SHEBOYGAN AVE.	MADISON	WI	10/7/1964
RURAL INSURANCE COMPANY, WISCONSIN ACOUSTIC & FIREPROOF	MADISON / WAUSAU / H		1/1/1970
COURT HOUSE	MILWAUKEE	WI	5/7/1969
DALON BUILDING, 1833 N. FARWELL AVE.	MILWAUKEE	WI	7/25/1963
GENERAL MERCHANDISE COMPANY, W. BURLEIGH	MILWAUKEE	WI	6/17/1963
GREYHOUND BUS DEPOT, 7TH AND WISCONSIN	MILWAUKEE	WI	9/15/1965
INDUSTRIAL HEAT CONTROL	MILWAUKEE	WI	2/13/1959
SHORELINE SUPPLY COMPANY	MILWAUKEE	WI	8/19/1960
	MILWAUKEE	WI	
SOUTHERN BUILDINGS, 6TH AND WISCONSIN AVE. ST. RITA'S PARISH, S. 61ST ST.	MILWAUKEE	WI	7/20/1965
UNIVERSITY OF WISCONSIN, SCIENCE COMPLEX	MILWAUKEE		7/2/1963
WISCONSIN ACOUSTIC & FIREPROOF		WI	11/28/1969
	MILWAUKEE	WI	8/7/1964
FIRST NATIONAL BANK BUILDING	NEENAH	WI	7/29/1963
NEENAH FOUNDRY	NEENAH	WI	11/10/1967
TESMER CONSTR COMPANY	NEILLSVILLE	WI	7/2/1959
METAL SPUN PRODUCTS, BARCZAK BROS.	OAKCREEK / RACINE	WI	3/9/1962
GILSON BROTHERS COMPANY	PLYMOUTH	WI	7/6/1962
ALLEN MERCHANDISING COMPANY	RACINE	WI	8/2/1961
JACOBSON THOMAS TIRE COMPANY	RACINE	WI	11/3/1961
NEW BUILDING, CAPE BROTHERS CONSTRUCTION COMPANY, HWY. 41	RACINE	WI	1/1/1961
NEW CLEANING BUILDING, J.I. CASE COMPANY	RACINE	WI	7/29/1967
WISCONSIN NATURAL GAS COMPANY, NEW BUILDING SITE	RACINE	WI	11/25/1960
MEMORIAL MALL SHOPPING CENTER, J.C. PENNEY	SHEBOYGAN	WI	10/16/1968
THE VOLLRATH COMPANY, 1236 N. 18TH ST.	SHEBOYGAN	WI	
MAMMOTH SPRINGS CANNING COMPANY	SUSSEX		9/3/1963
MID-CITIES SHOPPING CENTER	TWO RIVERS	WI	12/20/1967
WATERFORD HIGH SCHOOL	WATERFORD	WI	3/26/1964
WAUSAU IRON WORKS	WAUSAU	WI	3/15/1963
STATE FAIR PARK, BUILDING #23	WEST ALLIS	WI	5/28/1963
WEST BEND COMPANY	WEST BEND	WI	7/9/1962
D.C. EVEREST HIGH SCHOOL	WESTON	WI	1/30/1970
RIVERVIEW HOSPITAL	WISCONSIN RAPIDS	WI	11/15/1965
WISCONSIN ACOUSTIC & FIREPROOF	WISCONSIN RAPIDS	WI	12/19/1966
CHARLESTON GENERAL HOSPITAL	CHARLESTON		7/2/1966
DANCO, INCORPORATED, 520 22ND ST.	HUNTINGTON		3/29/1962
SPENCER STATE HOSPITAL, LAUNDRY BUILDING	SPENCER		6/22/1960
WEIRTON STEEL COMPANY	WEIRTON		8/31/1964
WEIRTON STEEL COMPANY, NEW OFFICE BUILDING	WEIRTON		1/1/1961
BENNIE SNYDER	WHITE SULPHUR SPRING		• •
FMC	GREEN RIVER	WY	10/13/1971